AGREEMENT

between

NEW YORK STATE NURSES ASSOCIATION

and

ERIE COUNTY MEDICAL CENTER CORPORATION

January 1, 2023 - December 31, 2027

TABLE OF CONTENTS

	TIVE REVIEW	
STATEM	IENT OF PURPOSE	1
1. REC	COGNITION	1
2. DEI	FINITIONS	2
	SOCIATION STATUS	
	ASSOCIATION MEMBERSHIP	Δ.
3.01	ASSOCIATION DUESDEDUCTION	
3.02	DEDUCTION OF AGENCY SHOP FEE	
3.03	INSURANCE	
3.04 3.05	DEDUCTIONERROR	
3.06	ASSOCIATION BUSINESS: LOCALREPRESENTATIVE	
3.07	ASSOCIATION BUSINESS: GENERALREPRESENTATIVE	
3.08	ASSOCIATION BUSINESS: ACCESS TO MEMBERSHIPDURING WORKING HOURS	
3.09	UNION RELEASE TIME	
3.10	MEETING SPACE	
3.11	ASSOCIATION BUSINESS: BULLETIN BOARDS	
3.12	RELEASE TIME REPRESENTATIVE	
	IPLOYEE STATUS	
•	CLASSIFICATION	
4.01	REGULARFULI-TIMEEMPLOYEES	
4.02	REGULAR PART-TIMEEMPLOYEES	
4.03 4.04	PART-TIMEEMPLOYEES	
4.04	PER DIEMEMPLOYEES	
4.05	PROBATIONARY PERIOD	
4.07	JOB POSTING: ERIE COUNTY MEDICAL CENTER CORPORATION	
4.08	LAYOFF AND RECALL	
4.09	(HOLD)	
4.10	COMPETITIVE POSITIONS	
4.11	TRANSFERS	
4.12	SENIORITY LISTS	
4.13	TIME OFF FOREXAMS	13
4.14	TEMPORARYRELOCATION	14
5. PR	OFESSIONALSTATUS	
5.01	NURSINGCOMMITTEES	
5.02	STAFF DEVELOPMENTPROGRAM	
5.03	STAFFING	
5.04	Non-Nursing Duties	
	DURS OF WORK	
7. SA	LARY AND INCREMENT RULES	
7.01	PROMOTIONS	
7.02	DEMOTIONS	
7.03	REINSTATEMENT	
7.04	REALLOCATION	
7.05	RECLASSIFICATION	
7.06	TEMPORARY ASSIGNMENTS	
7.07	LEAVE OF ABSENCE - MILITARY	

	7.08	INCREMENTS	
	7.09	PAYPERIOD	23
	7.10	NEW APPOINTMENTS	23
	7.11	CERTIFICATION REIMBURSEMENT	24
0	MACI	NETARY BENEFITS: COMPENSATION FOR TIME WORKED	24
	8.01	OVERTIMECOMPENSATION	24
	8.02	WAGESCHEDULE	26
9.	MO	NETARY BENEFITS: HEALTH AND RETIREMENT	27
	9.01	HEALTH INSURANCE	27
	9.02	RETIREMENT PLAN	
	9.03	NYSNA TAX-SHELTEREDANNUITY	
	9.04	SICK LEAVECONVERSION	33
10). IV	ONETARY BENEFITS: MISCELLANEOUS	33
	10.01	SHIFT DIFFERENTIAL PAY	33
	10.02	TRANSPORTATION ALLOWANCE	
	10.03	PRECEPTOR DIFFERENTIAL	
	10.04	NURSE PRACTITIONER ANNUAL EXPENSE REIMBURSEMENT	
	10.05	Two Medication Cart Discretionary Bonus (Terrace View)	
		IONETARY BENEFITS: COMPENSATION FOR TIME NOT WORKED	
13		VACATIONS	
	11.01	HOLIDAYS	
	11.02	PERSONAL LEAVE	
	11.03		
	11.04	BEREAVEMENT PAY	oc
	11.05	JURY DUTY LEAVE/COURT LEAVE	
	11.06	EMERGENCY AND DISASTERPLAN	39
	11.07	SICK LEAVE	
	11.08	Assault Pay	
	11.09	TUITION REIMBURSEMENT	
12	2. LI	EAVE OF ABSENCE WITHOUT PAY	
	12.01	ELIGIBILITY	
	12.02	APPLICATION FOR LEAVE WITHOUTPAY	
	12.03	CHILD CARE LEAVE	
	12.04	LEAVE BECAUSE OF EXTENDED ILLNESS	
	12.05	LEAVE FOR WARWORK	4 <u>4</u>
	12.06	LEAVE FOR EDUCATION PURPOSES	44
	12.07	EDUCATION LEAVE FOR VETERANS	44
	12.08	LEAVE OF ABSENCE TO SERVICE IN ANOTHER POSITION INTHE COUNTYSERVICE	
	12.09	LEAVE OF ABSENCE TO ACCEPT EMPLOYMENT OUTSIDETHE COUNTY SERVICE	
	12.10	ADOPTIONLEAVE	
	12.11	LEAVE OF ABSENCE FOR CIVIC PURPOSES	
	12.12	LEAVE FOR OTHERREASONS	
1:		RIEVANCES AND ARBITRATION	
1.	13.01	GENERAL	
	13.02	DEFINITIONS	
	13.02	RIGHTS OF THE PARTIES.	
	13.04	GRIEVANCEPROCEDURE	
	13.04	ARBITRATION PROCEDURE	
14		SISCIPLINE AND UNSATISFACTORY WORKPERFORMANCE	
	14.01	DISCHARGE AND DISCIPLINE	
	14.02	PROCEDURE	

14.	O3 DISPUTES AS TO DISCIPLINE	48
14.0	- -	
14.0		
15.	MANAGEMENT RIGHTS	
16.	NO STRIKE CLAUSE	
	PLEDGE AGAINST DISCRIMINATION AND COERCION	
17.		
18.	EQUAL OPPORTUNITIES	
19.	GENERAL PROVISIONS	49
19.		
19.		
19.		
19.		
19.		
19.		
19. 19.		
19. 19.		· ·
19.		
19.		
19.		
20.	SAVINGS CLAUSE	52
21.	TERMINATION AND MODIFICATION	
	PRINTING OF THE COLLECTIVE BARGAINING AGREEMENT	
22.		
23.	EXECUTION	
	NDICES AGREEMENT SIGNATURE PAGE	
APPE	NDIX "A" NYSNA TITLE LISTINGS	55
APPE	NDIX "B" DUES ASSIGNMENT AND DEDUCTION AUTHORIZATION FORM	60
APPE	NDIX "C" 1 NYSNA SALARY SCALE-ECMCC: PAY PERIOD FOLLOWING JUNE 19, 2022	61
APPE	ENDIX "C" 2 NYSNA SALARY SCALE-ECMCC: PAY PERIOD FOLLOWING JULY 30, 2023	62
	ENDIX "C" 3 NYSNA SALARY SCALE-ECMCC: PAY PERIOD FOLLOWING JANUERY 1, 2024	
	ENDIX "C" 4 NYSNA SALARY SCALE-ECMCC: PAY PERIOD FOLLOWING SEPTEMBER 1,2024	
	ENDIX "C" 5 NYSNA SALARY SCALE-ECMCC: PAY PERIOD FOLLOWING SEPTEMBER 1, 2025	
	ENDIX "C" 6 NYSNA SALARY SCALE-ECMCC: PAY PERIOD FOLLOWING SEPTEMBER 1, 2026	
	ENDIX "C" 7 NYSNA SALARY SCALE-ECMCC: PAY PERIOD FOLLOWING SEPTEMBER 1,2027	
	NDIX "D" MOARE: DEFERRED ANNUITY PLAN	
	NDIX "E" MOA RE: TEN, TWELVE AND THIRTEEN HOUR SHIFTS	
	NDIX "F" MOA RE: REGISTERED NURSE ON-CALL COVERAGE: OR, PACU, CAPD, HEMODIALYSIS/PLASMAPHERESIS	
	ITERVENTIONAL SERVICES	72
	NDIX "G" MOARE: INDIVIDUAL OVERTIME AGREEMENT	
	NDIX "H" MOA RE: PER DIEM CRNA AGREEMENT	
APPE	NDIX "I" MOARE: NURSE ANESTHETIST SHIFT COVERAGE	//

OR SATURDAY TRAUMA CALL	78
ENEFIT ADJUSTMENT	80
	87
D NURSE AGREEMENT	88
D NURSE PROGRAM	90
ON-CALL COVERAGE - SANE	91
R PAY	93
ACCRUALS	95
PROGRAM	96
ELIGIBLE FOR EMPLOYER 30 YEAR CONTRIBUTION	
DMMITTEE	99
······	102
ESHIP AND COMMUNITY BASED HIRING PROGRAM	105
DEPENDENCY LEAVE OF ABSENCE	107
	108
NDMEIC PREPARATION AND SAFETY	109
OKRPLACE VIOLENCE PREVENTION	112

LEGISLATIVE REVIEW

IT IS AGREED BY AND BETWEEN THE PARTIES THAT ANY PROVISION OF THIS AGREEMENT REQUIRING LEGISLATIVE ACTION TO PERMIT ITS IMPLEMENTATION BY AMENDMENT OF LAW OR BY PROVIDING THE ADDITIONAL FUNDS THEREFORE, SHALL NOT BECOME EFFECTIVE UNTIL THE APPROPRIATE LEGISLATIVE BODY HAS GIVEN APPROVAL.

STATEMENT OF PURPOSE

It shall be the policy of the County of Erie and the Erie County Medical Center Corporation and the purpose of this Agreement to promote harmonious and cooperative relationships between the County of Erie and the Erie County Medical Center Corporation its employees, and to protect the public by issuing, at all times, the orderly and uninterrupted operations and functions of government.

This Agreement made between the County of Erie and Erie County Medical Center Corporation, hereinafter referred to as the "Employer" or "ECMCC" and the New York State Nurses Association, hereinafter referred to as "Association."

The Employer and the Association recognize their common interests beyond their collective bargaining relationship, thus they pledge to strive together to insure the highest quality of service by the Employer and the highest standards of professional nursing care and practice by the Association.

1. RECOGNITION

- A. The Employer recognizes the Association as the exclusive collective bargaining representative of the employees covered by the Agreement and covers each full-time, each regular part-time (regularly scheduled for forty (40) hours or more per pay period) and each part-time (regularly scheduled for less than forty (40) hours per pay period) and per diem employee licensed or otherwise lawfully authorized to practice as a registered professional nurse (hereafter called "employee") employed by the Employer to perform registered professional nursing in a position specified in Appendix A as defined by the Public Employment Relations Board in its decision of September 26, 1986, as contained in the decision affecting certification of representative status in Case C-2830. This exclusive recognition will extend to the maximum permitted bylaw.
- B. The Association agrees that it will not interfere with, coerce or intimidate any employee into joining the Association. The Association recognizes that no employee is required to join, but that every employee has a right to choose of his/her own free will as to whether or not he/she will or will not join the Association. The Association further agrees that there will be no interference with the free right of any employee of the Employer to enter and leave its facilities and properties unmolested.
- C. The Employer agrees that there shall be no discrimination, interference, restraint or coercion by the Employer on behalf of or against any of its employees because of membership in the Association or for engaging in legal Association activity.
- D. The parties agree to continue the ongoing practice that the wages, benefits, and terms and conditions of employment negotiated in this Agreement shall apply to Nurse Practitioners in the bargaining unit.

2. DEFINITIONS

The following terms as used in this Agreement shall have the following meanings only:

- 2.1 "Employer" means ECMCC.
- 2.2 "Association" or "NYSNA" means New York State Nurses Association.
- "Employee(s)" means those individuals who are covered within the bargaining unit as defined in Section 2.04 of this Collective Bargaining Agreement.
- 2.4 "Bargaining Unit" means the certified collective bargaining unit commonly known as the "Nurse Unit" which is composed of only those employees as defined above (see recognition clause).
- 2.5 "Position" means one of the positions included under one class title in the Plan of Class Titles and Salary Ranges.
- 2.6 "Class" means a group of similar positions included under the same title in the Plan of Class Titles and Salary Ranges.
- 2.7 "Salary Range" means the range of compensation, from base to Step 10, as appearing in the Plan of Class Titles and Salary Ranges.
- 2.8 "Job Group" means a group of classes of positions allocated to the same salary range in the Plan of Class Titles and Salary Ranges.
- 2.9 "Increment" means the annual increment as provided in the table of increments for Job Groups in the Plan of Class Titles and Salary Ranges.
- 2.10 "Increment Step" means the point in the increment scale reached through successive periods of actual service, as designated in Steps 1, 2, 3, 4, 5, 6, 7, 8, 9, 10 and 11 in the Plan of Class Titles and Salary Ranges.
- 2.11 "Open Competitive List" means an eligible list resulting from an open competitive civil service examination.
- 2.12 "Promotional List" means an eligible list resulting from a promotional civil service examination.
- 2.13 "Appointing Authority" means the person or body ultimately responsible for employee selection and appointment within the department (usually the department head) subject to Civil Service Law and any rules promulgated thereunder.
- 2.14 "Actual Service" means total time spent actually working in a position including temporary positions and probationary service and also including any paid time off and/or leaves with pay.
- 2.15 "Service" means "Actual Service" as defined above.
- "Continuous Service" means an uninterrupted period of actual service which includes any authorized leave of absence without pay computed from the date an employee assumes a position with the County of Erie and/or ECMCC. Effective July 1, 2002 a period of actual service in a regular part-time position (forty [40] hours or more per pay period) shall be credited on a prorated basis based on the number of hours an employee works during such continuous service period. However, a period of actual service in a part-time position (less than forty {40} hours per pay period) shall be credited as fifty percent (50%) of such period for

purpose of computing continuous service. An employee's continuous service is interrupted by voluntary resignation, discharge, retirement or layoff. If an employee is rehired within one (1) year or is recalled within two (2) years of layoff or during his/her period of eligibility on a preferred list pursuant to Civil Service Law or reassumes a permanent full-time permanent position after holding a non-regular part-time position without a break in actual service, the interruption in continuous service shall be removed. The period of interruption, however, shall not be included in computing the total period of continuous service. Except as so provided, an interruption shall result in a permanent loss of all continuous service credits. In addition, should an employee assume a "Managerial/Confidential" position with the County of Erie and/or ECMCC and without a break in actual service reassumes a bargaining unit position, time spent outside of the bargaining unit shall be included in computing the total period of continuous service but shall not be included in computing the employee's time in a bargaining unit position.

- 2.17 "Seniority" except as defined and determined by Civil Service Law for purposes of the provisions therein, shall mean the length of an employee's continuous service as defined above with the County of Erie and/or ECMCC, in the bargaining unit and shall cover all areas of the Collective Bargaining Agreement unless otherwise specifically stated in the Agreement.
- 2.18 "Department" shall mean a defined work unit classified by ECMCC as a department.
- 2.19 "Department Head" shall mean the ECMCC employee designated as the head of a respective department.
- 2.20 "Immediate Supervisor" shall mean the employee who normally assigns and supervises another employee's work and approves his/her time record or evaluates his/her work performance. The Employer will designate by name the immediate supervisor in a particular department.
- 2.21 "Permanent Vacancy" means an unencumbered budget position covered by the bargaining unit to be filled on a permanent basis, which may or may not be filled within the sole discretion of the employee's Department Head. All requirements and rights granted under Civil Service Law and regulations must be abided by and upheld in filling any permanent positions.
- 2.22 "Section" means a sub provision of anarticle.
- 2.23 "Nursing Function" is defined as diagnosing and treating human responses to actual or potential health problems through such services as case finding, health teaching, health counseling, and provision of care supportive to or restorative of life and well-being, and executing medical regimens prescribed by a licensed or otherwise legally authorized physician or dentist, as reflected in Article 139 of the State Education Law, Section 6902.
- 2.24 "Ties in Bargaining Unit Seniority"
 Should any ties in bargaining unit seniority arise under this Section or any other section of the contract wherein bargaining unit seniority decides the allocation or selection of a benefit, such tie will be broken by a comparison of total County seniority. The employee with more County seniority will be treated as though he/she has more bargaining unit seniority than the other employee involved in the tie.

If the comparison of county seniority fails to break the tie the employee with the lowest last three numbers of his/her social security number will be treated as though he/she has more bargaining unit seniority than the other employee involved in the tie.

3. ASSOCIATION STATUS

3.01 Association Membership

Membership in the Association will not be a condition of employment, nor will the Employer discriminate in hiring or promoting employees or otherwise because of an employee's membership or participation in the Association or his/her refusal to join such organization.

3.02 Association Dues Deduction

- A. An employee desiring to become a member of the Association may execute a written authorization in the form annexed hereto as Appendix B. Upon receipt of the authorization from an employee, the Employer shall, pursuant to the authorization, deduct from the wages due the employee each pay period.
- B. The Employer following each pay period from which those deductions are made will transmit the amount so deducted to the Association within thirty (30) days. All transmittals shall be sent out by a listing of the members from whom the deductions have been made and the amount deducted from each to:

New York State Nurses Association 155 Washington Avenue Albany, NY 12210

- C. The Association shall certify to the VP of Human Resources for ECMCC in writing the current rate of membership dues and shall give the Employer thirty (30) days' notice prior to the effective date of any changes.
- D. A deduction authorized by any employee shall continue as long as so authorized unless and until such employee notifies the VP of Human Resources for ECMCC of his/her desire to discontinue or to change such authorization in writing and by registered mail and the Employer shall forward a copy of the employee's notification to the Association.

3.03 Deduction of Agency Shop Fee

The Employer will, for each employee covered by this agreement who by written and signed direction voluntarily authorizes the Employer to deduct from the wages due such employee in any month an agency shop fee equal to the regular dues fixed by Association for such month. The deduction shall be made on a biweekly basis and shall be transmitted at the same time and to the same office as set forth in Section 3.02 above. Each such agency shop fee deduction will continue in force and effect until revoked by (a) an employee's written and signed direction on a form prescribed by the Association under Section 3.02 to deduct Association dues from the employee's wages; or (b) termination of such employee's employment in a bargaining unit position. The Association agrees to hold the Employer safe and harmless because of said deduction.

3.04 Insurance

The ECMCC further agrees to grant to the Association an exclusive payroll deduction of premiums for an employee organization-sponsored insurance program. The NYSNA shall have the right to designate an insurance representative to visit an individual employee covered by this Agreement, on the job, for the purpose of explaining the insurance protection of a NYSNA- sponsored insurance program.

Only one such visit per employee per contract year shall be permitted and only after notifying the department head or his/her designee in writing given five (5) days in advance, and in no event shall such visit exceed one-half (½) hour in duration.

The designation of an insurance representative shall be in writing and sent to the V.P. of Human Resources at the time such designation is made, setting forth the specific names and addresses of such insurance representatives. In no event shall there be more than two (2) representatives designated at any one time. Failure on the part of the Association and/or its insurance representatives to comply with the provisions of this Section shall release the ECMCC from any obligation provided in this Section for the remaining term of this Agreement.

3.05 Deduction Error

If, through inadvertence or error, the Employer fails or neglects to make a deduction which is properly due and owing from an employee's paycheck, such deduction shall be made from the next paycheck of the employee and submitted to the collective bargaining representative. The Employer shall not be liable to the collective bargaining representative, employee or any party by reason of the requirements of this Section of the Agreement for the remittance or payment of any sum other than that constituting actual deductions made from employee wages earned.

3.06 Association Business: Local Representative

The Association will notify the Administrators of the local representatives who are authorized to deal with Administrators about employment conditions and adjustments of any problems arising under this Agreement. The Association will notify ECMCC in writing of said representatives' designation and authority and any change in either. Local representative or designee will be allowed reasonable time off with pay to handle grievances.

3.07 Association Business: General Representative

A duly authorized general representative of the Association may visit the Employer's premises to discharge Association's duties as the employees' collective bargaining representative so long as the representative does not interfere with the work of the employees.

3.08 Association Business: Access to Membership During Working Hours

Association local chairpersons and representatives shall, on an exclusive basis have access to employees during working hours to explain Association membership services and programs under mutually developed arrangements with Department heads. Any such arrangements shall insure that such access shall not interfere with work duties or work performance and shall be limited to no more than fifteen (15) minutes per employee per year.

3.09 Union Release Time

- A. The employer agrees to grant time off without charge to accumulated leave credits, with pay, for Association delegates to conventions of the Association, or other bodies with which the Association is affiliated, but not to exceed sixty (60) working days in any one (1) calendar year.
- B. Witnesses at arbitration hearings: witnesses will be released for time missed from work as a result of time spent actually testifying in an arbitration hearing. The grievant will be released for any time missed from work which is spent in his/her arbitration hearing. The association will designate up to two (2) members of a class action grievance for release time during the arbitration hearing.

Release time for participation in arbitrations as noted in (B) above shall be deducted from the time allotted in (A) above.

- C. Requests to utilize Union Release Time provided for in paragraph "A" above that are made at least five (5) days prior to the time off requested shall be approved, provided that the sixty (60) working days in anyone (1) calendar year has not been exhausted. In the event that the five (5) day notice is not possible, the Corporation will grant a request made with less than five (5) days' notice, provided that the sixty (60) working days have not been exhausted.
- D. The parties will meet six (6) months prior to the expiration of the current contract and set a schedule for negotiations. A bargaining committee will be released for time missed from work spent in negotiations for a successor agreement up to a maximum of three hundred twenty (320) hours annually (three hundred twenty {320} hours will be granted each calendar year and unused hours will be rolled over into the next calendar year up to a maximum of eight hundred {800} hours). Upon ratification of a successor agreement the balance of this time will be deleted and accrual will restart.

Employees working the shift during negotiations will be released for the actual hours spent in negotiations and where hours remain in the shift after the negotiation session is concluded or before it begins will be allowed to use benefit time or work (employees discretion). Employees who are not working the shift during negotiations will be permitted to use release time for a full scheduled shift within twenty-four (24) hours prior to the commencement of negotiations or twenty-four (24) hours after negotiations conclude. The union will provide the name of the employees that require this release. If an employee is scheduled for more than one shift within 24 hours of the commencement or scheduled conclusion of negotiations, that employee will notify their immediate supervisor of which shift they choose to utilize union release for. This time will not be deducted from the time allotted above in (A) above.

The association will furnish letters to the employer designating the persons to be released within seven (7) working days of the scheduling of dates, and will keep the employer updated as to any change.

Employees to be released are required to submit the appropriate forms for the time off to their manager or designee.

3.10 Meeting Space

NYSNA is accorded the privilege of use of meeting space on the Employer's

property. Attendance at meetings shall be limited to bargaining unit employees who are on meal break or are off-duty, and Association general representatives.

The employer will make a good faith effort to provide on-site office space for the Associations general and local representative's utilization. The availability of office space will be a labor management agenda item until such time as space is provided.

3.11 Association Business: Bulletin Boards

Employer will provide Association with, a minimum of four (4) bulletin boards on which to post official Association notices, suitably located on the ground floor and second floor of the Medical Center (main hospital), the Outpatient Dialysis Center, and Terrace View Long Term Care. These bulletin boards will at all times carry a label, device or notice clearly identifying them as Association's space (for use). The signature (or facsimile signature) of a duly authorized Association representative will be affixed to every Association notice posted in this space. The bulletin boards shall be locked and a key held by the Employer and the local representative.

3.12 Release Time Representative

Employees designated by the Association will be released to discharge Association duties as the collective bargaining representative for a cumulative total of sixty-four (64) hours per week with all the pay and benefits as if they had remained in their department, on their shift. The Association will designate the employees to be released and the duration of their release two (2) weeks in advance of the release.

This will exclude political work and require the released employees to be on-site.

4. EMPLOYEE STATUS

4.01 Classification

Employees covered by the Agreement will be classified as either (a) full-time; (b) regular part-time; (c) part-time; or (d) per diem.

4.02 Regular Full-Time Employees

A regular full-time employee is an employee covered by this Agreement who is employed on a regular basis to work a normal workweek. A regular full-time employee will receive full fringe benefits.

4.03 Regular Part-Time Employees

Regular part-time employees regularly scheduled forty (40) hours or more per pay period (excluding per diem, seasonal, temporary, substitute, per session on a non-regularly scheduled basis) shall earn credits and be entitled to receive, upon meeting the requirements of the applicable provisions of this Agreement, the following benefits only and only to the extent provided:

- A. Vacation 50% of full-time employee;
- B. Sick Leave 50% offull-time employee;
- C. Holiday 50% of full-time employee;
- D. TransportationAllowance same as full-time employee;
- E. Jury Duty Pay same asfull-time employee;
- F. Shift Differential same asfull-time employee;

- G. Health Insurance same asfull-time employee;
- H. Coverage under Retirement System as provided in this Agreement;
- I. A one-half (1/2) hour paid lunch break for regular part-time employees;
- J. All regular part-time employees shall be granted an increment provided they have worked at least one (1) year for the Employer. Regular part-time employees shall receive increments the same as regular full-time employees;
- K. Bereavement Leave same as full-time employee;
- L. Seniority in accordance with Section 2.16.;
- M. Personal Leave 50% of the full-time employee.

4.04 Part-Time Employees

Part-time employees regularly scheduled less than forty (40) hours shall be entitled to receive, upon meeting the requirements of the applicable provisions of this Agreement, the following benefits only and only to the extent provided:

- A. Transportation Allowance same as full-time employees;
- B. Jury Duty Pay same as full-time employees;
- C. Shift Differential same asfull-time employees;
- D. Coverage under Retirement System as provided by New York State Law;
- E. Increments as provided for part-time employees in this Agreement in Section 7.08 (F);
- F. A one-half (1/2) hour paid lunch break for part-time employees;
- G. All items in the contract shall apply to part-time employees except those listed below:
 - 1. Section 4.14, Temporary Relocation
 - 2. Section 8.01(C), Compensatory Time
 - 3. Section 8.01(H), Emergency Duty
 - 4. Section 9.01, Health Insurance
 - 5. Section 11.01, Vacations
 - 6. Section 11.02, Holidays
 - 7. Section 11.03, Personal Leave
 - 8. Section 11.07, Sick Leave
 - 9. Section 12.04, Leave Because of Extended Illness
 - 10. Section 12.08, Leave to Serve in Another County Position
 - 11. Section 12.09, Leave to Accept Employment Outside County

4.05 Per Diem Employee

Per Diem employees are employees hired and scheduled in accordance with the Memorandum of Agreement in Appendix L. Per Diem employees shall receive benefits only as described elsewhere in this Agreement.

4.06 Probationary Period

The rules governing probationary terms are set forth in Rule XIII of the Rules for

the Classified Civil Service of the County of Erie as amended from time to time.

The probationary term for every permanent appointment to a permanent position from an open competitive or promotional civil service list and every original appointment and promotional appointment to a position in the non-competitive or labor class, shall be for a probationary term of not less than eight (8) nor more than twenty-six (26) weeks.

The probationer shall be advised by his/her supervisor at least at the end of the eight (8) weeks and then at least every nine (9) weeks in writing as to status and progress. If found to be unsatisfactory, the appointing authority shall give the probationer at least one (1) week's written notice that his/her service in the position will terminate at the end of the probationary term. An employee who is terminated under this provision who holds a permanent appointment to a lower level position shall be immediately reinstated to such lower level position without loss of benefits or seniority.

The decision to retain or terminate the probationer in that position will be made solely by the appointing authority and shall not be subject to appeal through the grievance procedure in this contract, unless a violation of Articles 17 or 18 is *involved*. The employee shall be entitled to Association representation upon request.

4.07 Job Posting: Erie County Medical Center Corporation

Whenever a temporary (as defined in B below) or permanent vacancy at the Medical Center occurs, after ECMCC exercises its right to reassign employees throughout ECMCC (ECMC Employees can only be reassigned within ECMC and Terrace View employees can only be reassigned within Terrace View), the remaining vacancy, if any, shall be filled in the following order and manner:

A. Recall rights.

B. Posting

All such vacancies shall be posted online and on the ECMCC job board for a period of seven (7) continuous calendar days. All temporary vacancies that are encumbered due to the incumbent serving a probationary period or a provisional appointment in another position shall be posted as if they are permanent vacancies. This will alleviate the need to re-post the position after the position becomes unencumbered. The opportunity to fill the position shall be given to all employees who meet the minimum qualifications and who have not transferred within one (1) year of the respective posting in accordance with Section 4.12. During the posting period, all interested and eligible employees may apply online for the position in question. In filling vacancies, the Department Head or his or her designee shall review an applicant's qualifications including training, experience, skills, abilities, and past work record. However, in the case of promotion, education is an additional factor to be considered. Where it is determined that the qualifications of two (2) or more applicants are substantively equal, the vacancy shall be given to the person with the most bargaining unit seniority. If no one applies for the vacancy in question or if the department head determines that no qualified applicant is present, ECMCC shall have the right to fill the vacancy from another source. If the initial posting is filled from within, its subsequent resulting vacancy shall also be posted. However, if a resulting third vacancy due to internal movement occurs, it shall be filled at management's discretion. The names of all persons applying for a particular position and the selected employee shall be sent to the unit local representative chairperson(s). The Employer will make their best efforts to place their employee in the position for which he or she is chosen within thirty (30) calendar days. The Employer shall provide the applicable promotional rate of pay thirty (30) days after ECMCC receives the employee's acceptance of the offer of the promotional position. All applications shall be submitted online through the ECMCC website.

If the Employer chooses to post a vacant position, all vacant positions will be posted in accordance with this Section (4.07, Job Posting: Erie County Medical Center Corporation) no more than two weeks from the date the Employer is notified of the impending vacancy, or the day the position becomes vacant, whichever is first. If the date of the permanent vacancy is on a holiday or weekend, the position will be posted on the next successive work day.

In the case of internal transfers, a release date will be given to the vacating nurse, no more than thirty (30) days from the date the position is awarded. All offer letters or awarded internal positions must be time stamped and in writing. Verbal offers must be memorialized and dated the same day that the offer is accepted. The vacancy created by the internal transfer shall be posted in accordance with this Section (4.07, Job Posting: Erie County Medical Center Corporation), the day of the permanent vacancy, or if the date of the permanent vacancy is on a holiday or weekend, the next successive work day, whichever is first.

First contact with all qualified external applicants will be made within forty-eight (48) hours of receipt of the application to schedule for an interview.

4.08 Layoff and Recall

A. General

- In the event of a layoff, the Employer shall identify the positions to be vacated by layoff. Once positions are identified for layoff, the Employer will utilize the procedure detailed throughout this Section.
- 2. Prior to any permanent incumbent in the respective job title being laid off in any department or institution, temporary, provisional or probationary employees in the job title in that department or work area shall be laid off first in that order.
- 3. Any member of the bargaining unit who is designated for layoff under the procedures outlined in either 4.08(B) or 4.08(C) may elect to apply for a vacant General Duty Nurse or Registered Nurse class if, in the opinion of the respective Director of Nursing, the displaced employee possesses the expertise to independently perform the duties of the particular opening. All open vacancies that have not yet been awarded on the date the Employer notifies the Association of it's intent to layoff shall be preserved as the initial option for affected employees. All preserved vacant positions shall be offered to affected employees in order of bargaining unit seniority. An employee shall have twenty-four (24) hours from the time the employer provides a list of preserved vacant positions to identify the position of their choice. Employees affected by a layoff/displacement will not be subject to the twelve (12) month restriction on bidding for a new position.
- 4. Employees shall not displace from one department or institution to another,

- i.e., ECMC to Terrace View LTC, or vice versa.
- 5. Any nurse reassigned or re-employed due to the implementation of this clause will receive an orientation to their new work assignment.
- 6. Should any ties in bargaining unit seniority arise under this Section or any other section of the contract wherein bargaining unit seniority decides the allocation or order of a benefit or choice, such tie will, be broken by a comparison of total County seniority. The employee with more County seniority will be treated as though he/she has more bargaining unit seniority than the other employee involved in thetie.

If the comparison of county seniority fails to break the tie the employee with the lowest last three numbers of his/her social security number will be treated as though he/she has more bargaining unit seniority than the other employee involved in the tie.

B. Competitive Class Employees

- 1. The retrenchment and recall of all competitive class employees in the bargaining unit shall be pursuant to Sections 80 and 81 of the New York State Civil Service Law as amended from time to time.
- 2. In addition, subsequent to the Section 80 and 81 procedure, a competitive class employee may displace a non-competitive class employee if they meet the following criteria:
 - a. The employee has greater bargaining unit seniority; and
 - b. The employee meets the minimum qualifications of the respective title including the relevant experience most of which occurred within the past seven (7) years; and
 - c. The employee previously held the title and in the opinion of the respective Director of Nursing the employee possesses the expertise to independently perform the duties of the particular position. (Such determination by the Director of Nursing shall not be arbitrary and capricious.)
- 3. The review of displacement opportunities to lower titles shall occur in the following order:
 - a. Team Leader/Head Nurse
 - b. Asst. Head Nurse (ECMC)/Charge Nurse If new classes are created and designated as non-competitive, the Employer will meet with the Association to determine placement within this Section.
- 4. If a bargaining unit employee is unable to displace anyone in the titles noted in B.3. of this article, he/she will have the right to displace a General Duty Nurse/Registered Nurse bargaining unit seniority permitting.
- 5. An employee may exercise their right to accept layoff in lieu of displacing another employee.
 - It is agreed and understood that should a competitive class employee not wish to exercise any bump permitted under Civil Service Law or fail to so

exercise within five (5) working days from the date of the Notice of Layoff, such employee will be placed on a preferred list pursuant to Civil Service Law.

C. Non-Competitive Employees

- 1. Where there is a layoff in a specific non-competitive job title and no temporary or probationary employees are involved, the employee with the lowest seniority in the job title shall displace an incumbent with the lowest bargaining unit seniority in the same job title in the respective department or work area. Additionally, full-time employees may displace full-time, regular part-time employees or part-time employees. Regular part-time employees may displace regular part-time employees and part-time employees only.
- 2. If the employee who has been displaced as a result of paragraph "1" above who previously held a lower level bargaining unit job title on a permanent basis, he/she may displace the employee with the lowest bargaining unit seniority in his/her department or work area who holds such lower level job title.
- 3. The employee who was displaced under paragraph "2" who previously held a lower level bargaining unit job title on a permanent basis, may displace the least senior employee based on bargaining unit seniority who holds such lower level job title in his/her department orwork area.
- 4. The employee who was displaced under paragraph "3" who previously held a lower level bargaining unit job title on a permanent basis will be laid off. Such laid off employee will be placed on a recall list by job title in descending order of bargaining unitseniority.
- 5. Layoff, in lieu of displacement. In the event of a layoff, if an employee does not wish to displace into a lower level job title held by a junior employee or fails to exercise his/her displacement rights within five (5) working days from the date of Notice of Layoff, he/she will be placed on a recall list.
- 6. Layoff. Part-time employees not scheduled for work for a period of sixty (60) consecutive calendar days shall be considered laid off and bumping rights shall be effective on the sixty-first (61st) calendar day.
- 7. Recall. Whenever a vacancy occurs in a job title with the Employer, employees who are on layoff in that job title shall be recalled in accordance with their job title seniority in the reverse order in which they were laid off. If a vacancy occurs in a job title where no employee in that job title with the Employer has recall rights, then the laid off employee with the most seniority will be recalled if he/she has the ability to do the work and if not, the next senior employee will be recalled. Recall rights for employees on layoff will expire two (2) years from date of last layoff.

Notice of Layoff

The Employer will give twenty-one (21) calendar days' notice of layoff. This applies to employees who are initially laid off because their position has been abolished and not to any employees who are retrenched as a result of any applicable bumping procedure, whether under this Agreement or New York State Civil Service Law.

D. Notice of Recall

Notification of recall shall be sent to the employee by certified or registered

mail, return receipt requested, to the employee's last known address. An employee will have four (4) workdays from the date of delivery or attempted delivery of the recall notice in which to inform the Employer of his/her intentions to accept the recall or not. If the employee refuses such offer of reemployment or if such offer remains unanswered at the end of the three (3) workday period, such offer of reemployment and the employee's recall rights and privileges shall be terminated. The employee, if he/she chooses to accept the recall, will have up to eleven (11) workdays from the date he/she received the recall notice to return to work.

- 1. The Employer will be liable for any error on a separation or layoff only from the date of the filing of a written grievance bringing the specific error to the Employer's attention.
- 2. Concerning any errors in the recall of an employee, the Employer shall be liable only from the date of filing a written grievance bringing said error to its attention to the date the Employer notifies the employee to return to work.
- 3. A laid off non-competitive employee who is recalled within two (2) years in any position will have any accumulated sick leave, personal leave, and/or compensatory time credits restored. A laid off competitive employee who is recalled within four (4) years in any position will have any accumulated sick leave, personal leave, and/or compensatory time credits restored.

4.09 (Hold)

4.10 Competitive Positions

It is agreed and understood that competitive positions must and shall be governed by the Civil Service Law and Rules and Regulations which shall supersede the provisions of this Agreement.

4.11 Transfers

Employees who voluntarily transfer or accept a promotion must remain on the new assignment and in the same position for a period of one (1) year, unless a shorter period is mutually agreed to by the Employer and the Association. However, employees will not be restricted from transferring to a different shift or status (FT, RPT, PT) within the same nursing unit. The restriction prohibiting employees who have transferred within the last one (1) year period is not applicable to promotional opportunities.

4.12 Seniority Lists

Seniority lists shall be posted by February 1st of each calendar year. A copy shall be simultaneously sent to the Association. The Personnel Department/Human Resources is responsible for maintaining seniority lists (as defined in Article 2) and each department head or his/her designee will be responsible for the posting of such lists.

Upon any layoff or elimination of bargaining unit position, an updated seniority list will be generated and a copy provided to the Association.

4.13 Time Off for Exams

A. Permanent employees will be released from work to compete in promotional and open competitive County Civil Service examinations in their occupational field only. It is understood that in any instance, wherein the examination is over prior to the end of an employee's shift, he or she must return to work. Furthermore, such time off will only be granted if the examination time conflicts with the employee's regularly scheduled work hours.

- B. It is agreed that prior to the posting of work schedules, the Employer may schedule an employee off on the day of a Civil Service exam or State Board exam in order to avoid a conflict with the examination.
- C. Members of the bargaining unit who are required to take the State Board examinations, in whole or part, will be allowed time off with pay.

4.14 Temporary Relocation

When it becomes necessary to move an employee from one geographic, clinical or program assignment to another assignment for a temporary period, unless an employee volunteers, any such temporary assignment in excess of ten (10) consecutive workdays shall be assigned to the least senior qualified nurse who shall be so assigned until such temporary assignment ends.

Any such employee who is assigned temporarily for more than ten (10) consecutive days shall not be temporarily assigned again for a twelve (12) month period following completion of the temporary assignment, unless all other qualified employees in the same unit or work location have also been temporarily assigned during the same twelve (12) month period.

The following shall be exempted from this time limit: ICU, OR, and Hemodialysis. The selection of an employee for temporary assignment shall be based upon the employee's skills and expressed interests. The names of all employees temporarily reassigned for periods greater than thirty (30) days shall be sent to the unit chairperson(s).

The Employer shall not utilize temporary relocation of employees so as to avoid or delay filling a permanent vacancy.

5. PROFESSIONAL STATUS

5.01 Nursing Committees

There shall be an Advisory Committee on Nursing Practice at the Erie County Medical Center (ECMC) and Terrace View LTC. Each Advisory Committee on Nursing Practice may have a Subcommittee on Staffing. The membership of such committees, which shall be limited to employees covered by this Agreement, shall be determined by the Association provided that the Association will advise the Employer, in writing, of the membership of any committees.

The meetings and activities of such committees shall be during non-working hours except in those calendar months when the Association elects to conduct Nursing Practice Committee meetings on duty time. In such event, the Labor Management Committee meetings under Section 19.02 shall be held on off duty time, and management representatives shall make themselves available for these meetings, provided the request complies with Section 19.02.

In the interest of encouraging second and third shift nurses to attend Labor Management and/or Nursing Practice monthly meetings, it is agreed that should nurses from the second or third shift request and actually attend such meetings outside of their regularly scheduled shift, a maximum of two (2) such nurses shall

be paid for actual attendance time at their straight time base rate and such time shall not be counted toward the payment of the overtime premium.

At Terrace View LTC, the Association will submit a written agenda to the Director of Nursing at least five (5) calendar days prior to the meeting date.

A. Purpose

The purpose of the committees shall be to make recommendations with respect to the philosophy of nursing and practice specific to the Division and methods to foster adherence to that philosophy.

B. Functions

- 1. Review and make recommendations regarding standards of nursing practice specific to the Division and consistent with those enunciated by the profession.
- 2. Analyze the factors which facilitate or impede the practice of nursing and make recommendations with regard to the same, for example:
 - a. Nursing involvement in non-nursing responsibilities;
 - b. Staffing patterns and ratios;
 - c. The adequacy of resources and supportive services essential to the practice of nursing;
 - d. The relationships between nursing and other disciplines of the Division.
- C. Recommendations of the Committees on Nursing Practice shall be referred to the Director of Nursing in the Erie County Medical Center and the Director of Nursing at Terrace View LTC. The appropriate nursing management representatives will meet with the committee on duty time and will reply to the committees' recommendations within thirty (30) days in writing.
- D. The recommendations of these committees are advisory in nature and may not be the subject of a grievance. The decision of the appropriate nursing management representatives is final and binding.

5.02 Staff Development Program

The Employer shall continue the past practice with regard to the following:

- A. A planned employee orientation program supervised by the Director of Nursing at the Erie County Medical Center and the Director of Nursing at Terrace View LTC. Assignments to specialty units shall not be made without prior orientation to said unit unless a bona fide emergency exists or unless the nurse is qualified, by experience, to cover such specialty units.
- B. An organized program of in-service training, supervised by the Director of Nursing at the Erie County Medical Center and the Director of Nursing at Terrace View LTC
- C. The employee orientation and in-service training program described in (A) and (B) above will be conducted during the employee's regular work schedule where possible.
- D. Members of the Association will attend nursing and related seminars within budgetary limits and within the discretion of the Department Head, work

5.03 Staffing

- A. In each nursing division a committee will be established which shall meet at least monthly, when necessary. It shall be composed of a mutually agreed upon number of Division members and shall study and make recommendations on staffing to Nursing Administration. It is the intent of this provision to more effectively communicate staffing problems and recommendations between the Division and Nursing Administration and to establish a patient classification system.
- B. In the event any Registered Nurse determines that in his/her professional opinion he/she has been given an assignment that is unsafe and/or endangers patient care, he/she shall immediately notify his/her supervisor and begin the assignment. An employee shall not in any way or manner be disciplined for questioning the assignment.
- C. The supervisor or designee shall in a reasonable time, but no later than the end of the shift, contact the employee to review the problem and explain what steps, if any, were taken to remedy the situation. If the supervisor does not feel a problem exists he/she shall notify the employee.
- D. If the employee is dissatisfied with the supervisor's response, he/she shall continue the assignment and may do so under protest. The employee shall have the right to file an official protest of assignment, within twenty-four (24) hours, on a form agreed to by the parties, with the subcommittee on staffing included in Section 5.03 of this Agreement.
- E. The committee will present their findings and the basis for their rationale to the Director of Nursing for his/her consideration and recommendation.
- F. In the event that the number of registered nurses on a unit falls below the ratios on a given shift (e.g. because of a sick call-out, leave of absence, vacancies etc.), the employer may make a reasonable effort to re-establish the number of registered nurses utilizing available contractual means at the employer's discretion.
- G. "Reasonable effort" shall be defined as utilizing any available contractual means, which may include, but shall not be limited to reassigning registered nurses who can, at the point of being reassigned, perform all essential nursing duties on the unit, utilizing per diems, offering additional hours to part-time or regular part-time registered nurses, utilizing traveler/agency nurses, and/or offering overtime without the extra shift differential.
- H. If after the schedule has been posted, the employer is not able to utilize available contractual means to establish the staffing ratios in Appendix T of the Collective Bargaining Agreement and New York Public Health Law Section 2805-T, within seventy two (72) hours prior to the start of the shift, the employer must offer overtime with the extra shift differential. The staffing office will send out a mass text to all registered nurses on the unit clearly indicating the offering of overtime with extra shift differential. If after the staffing office sends out a mass text to all registered nurses on the unit, the staffing ratios are still not being met, the staffing office will send out a mass text to all registered nurses in the cohort no less than forty eight (48) hours prior to the start of the shift.

- 1. If there is a change in staffing and the need for staff occurs less than seventy two (72) hours prior to the start of the shift, the employer may make a reasonable effort to re-establish the number of registered nurses using available contractual means within two (2) hours from being notified of the call-off. If after a reasonable effort has been made and the ratios are still not being met, the employer must offer overtime with the extra shift differential. The staffing office will send out a mass text to all registered nurses in the cohort.
- J. If there is a change in staffing and the need for staff occurs during the shift, the employer may make a reasonable effort to re-establish the number of registered nurses using available contractual means within two (2) hours from being notified of the call-off. If after a reasonable effort has been made within two (2) hours from being notified of the call-off the ratios are still not being met, the employer must offer overtime with the extra shift differential. The staffing office will send out a mass text to all registered nurses in the appropriate cohort. Priority will be given to registered nurses willing to work an extension of shift.
- K. Once the employer offers the extra shift differential to a registered nurse working an extension of shift or an additional shift as set forth in this Section, on that unit/shift, then all registered nurses working an extension of shift or additional shift on that unit/shift shall receive the extra shift differential for such extension or additional shift. Registered nurses who work a shift as part of their regular schedule shall not be eligible for the extra shift differential for their regular scheduled hours.

The following shall be established as the cohort groupings:

Critical Care
Medical Surgical
Surgery
Transplant
Behavioral Health
Emergency Department
Terrace View
Dialvsis

- L. The provisions (D) and (E) of this Section 5.03 are not grievable.
- M. The Employer agrees to pay up to a maximum of three (3) members of this committee, designated by NYSNA, for attendance at committee meetings. Such pay shall be at straight time rates and shall not exceed two (2) hours pay for any given committee meeting. Such paid time shall not be counted for calculation of eligibility for overtime pay.
- N. The Extra Shift Differential shall not be authorized or paid for attending community events, recruiting activities, job fairs, or other, similar, non-patient care activity during the week or on weekends, unless otherwise approved in writing by the employer.
- O. Per Diem Registered Nurses shall be eligible to receive the Extra Shift Differential for extension of shift only when the extension of shift is a minimum of three (3) hours for thirteen (13) hour shifts or a minimum of four (4) hours for all other shifts, unless otherwise approved by the Employer.

- P. The Extra Shift Differential of \$27.00 per hour for all shifts that qualify under Section 5.03, F-K, Staffing will be paid under the following conditions:
 - a. When a bargaining unit member works a minimum of four (4) hours as an extension (before or after their scheduled hours) of their regularly scheduled shift, or a minimum of three (3) hours as an extension of their regularly scheduled shift for thirteen (13) hour employees.
 - b. When a bargaining unit member works a minimum of a four (4) hour shift in addition to their scheduled shifts.
- Q. This Extra Shift Differential is in addition to any overtime or differential that the employee would normally receive under the CBA for working within this time period.
- R. For Employees working a Forty Hour Agreement, that have a sixteen (16) hour a month clinical time obligation, clinical hours worked beyond their sixteen (16) hour monthly obligation, will not be counted towards their regular forty hours.

5.04 Non-Nursing Duties

The Employer and the Association agree that employees covered by this Agreement are employed to perform nursing and nursing related duties. It is in the best interest of both parties to agree that every effort shall be made to avoid assigning bargaining unit members to duties which are not related to nursing.

6. HOURS OF WORK

- A. Except as provided in Appendix E herein, a normal workday shall not exceed eight (8) consecutive hours in any twenty-four (24) hour period. The normal workweek shall not exceed forty (40) hours.
- B. All employees at ECMCC shall receive a thirty (30) minute lunch period.
- C. All Employees' work schedules shall provide for a fifteen (15) minute rest period during each one-half (1/2) shift.
- D. Employees will have at least sixteen (16) hours off between scheduled shift assignments; however, employees who work a ten (10), twelve (12) or thirteen (13) hour schedule will only have fourteen (14), twelve (12) or eleven (11) hours off between scheduled shifts when assigned consecutive shifts. Employees will work a maximum of one (1) different shift in a normal workweek, except if it is mutually agreed upon, free of any harassment or coercion by the employee and Supervisor. Part-time employees shall receive forty-eight (48) hours' notification prior to a change in the work schedule or work shift.
- E. The Employer will post a schedule of each employee's work seven (7) calendar days in advance of the work period. Once posted, the schedule shall be maintained, except in unanticipated emergency situations until it is superseded by a new schedule or changed by mutual agreement between the Employer and the employee concerned. Each schedule will be provided to the Association general representative(s) electronically within six (6) calendar days of posting.

In the event the Employer deems necessary any changes in the workweek or shift assignment, the employee and the Local Bargaining Unit Chairperson will be notified seven (7) calendar days in advance of the proposed change, except in emergency situations. When there are two (2) or more qualified employees with less than five (5)

years of bargaining unit seniority they will be rotated for no more than eighty (80) hours based on inverse order of seniority, unless otherwise agreed to by the parties. If only one (1) qualified employee has less than five (5) years seniority, that employee will be rotated first. If no qualified employee has less than five (5) years seniority or after exhaustion of all staff with less than five (5) years seniority, all eligible employees on the unit will be rotated as needed for no more than eighty (80) hours based on inverse order of seniority, unless otherwise agreed to by the parties. A new rotation list will be used the first day of each year.

F. Weekends, for the purposes of this clause, are defined as beginning at eleven o'clock post meridian (11:00 PM) Friday and ending at eleven o'clock post meridian (11:00 PM) Sunday. Employees working on a night variable shift shall have the weekend defined as beginning Friday of a regularly scheduled shift through the completion of the end of their regularly scheduled weekend assignment. Terrace View LTC will make every effort to grant employees every other weekend off.

The employer will grant employees twenty-six (26) weekends off per year. Should any employee not receive twenty-six (26) weekends off in a calendar year, they shall receive a bonus equal to four dollars (\$4.00) per hour for all such excess weekend hours. The Hospital shall have the right to schedule an employee to work up to a maximum of three (3) weekends in a row.

An employee who has scheduled his/her vacation shall receive off the weekend before the beginning of his/her vacation, the weekend after and any weekend that falls during the scheduled vacation.

The employer shall not schedule an employee to work a split weekend unless mutually agreeable between the employee and the employer. If mutually agreed, the day worked will be counted as one-half (1/2) of a weekend worked.

If an employee is absent for all or part of the employee's scheduled weekend, the employer may schedule the employee to make up such weekend work. If the employee works part of the scheduled weekend, he/she will be credited with the time worked for purposes of fulfilling the weekend obligation.

The employer and the employee who is absent on a scheduled weekend shall mutually agree on when the employee is to make up the unworked weekend. If the Employer and the employee cannot mutually agree within three (3) weeks, the employer will then have the right to reschedule the employee at its discretion provided that the employer gives at least two (2) weeks' notice of the scheduled weekend makeup.

- G. In the fall of the year when the clocks are set back one (1) hour, night shift employees work an additional hour. All employees who work this extra hour should be paid at that time for the additional hour worked. In the spring of the year when the clocks is set ahead one (1) hour, night shift employees work one hour less. All employees on the shift should be docked an hour at that time. Employees may choose to use benefit time to receive a full paycheck.
- H. The parties to this Agreement recognize that the concept of flextime may offer a relief to scheduling and/or recruiting problems. The parties also recognize that certain flextime plans may be in conflict with the current provisions of this Agreement. Therefore, during the term of this Agreement, if a nurse feels that such flextime would be desirable or if a management representative feels that such flextime would be desirable, a request to meet and confer shall be presented to either party.

Weekend Employees

Section 1. A weekend employee is defined as one who is hired to work and is regularly scheduled to work a full shift every Saturday and Sunday or every Friday beginning at 11:00 p.m. and Saturday beginning at 11:00 p.m. Weekend employees will be required to work the same shift duration as the unit/department to which they are assigned. Weekend Employee positions will be RPT. Employees' can elect to work additional hours during the week. When working additional hours during the week the differential in Section 2 would not be applicable.

Section 2. Employees hired to work in this capacity shall be paid a \$10.00 per hour differential. Employees hired in this capacity are eligible for shift differential in accordance with Article 10.01. The \$10.00 hourly differential shall not apply to time that is not worked, which includes holiday, vacation, sick, personal, and compensatory time.

Section 3. The differential paid to employees under Section 6(F) for working more than twenty-six (26) weekends shall not apply to Employees hired to work in this capacity.

Section 4. Employees hired to work in this capacity shall be required to work every weekend. No more than two (2) weekend shifts may be scheduled off per year, during the period starting with the week of Memorial Day and including the week of Labor Day and between the week including December 15th through the first week in January and any other weeks that contain a recognized holiday under Section 11.02. No more than eight (8) weekend shifts may be scheduled off per year outside of the period of time previously stated. When there are requests from more than one (1) employee for the same time off, the employee with the highest seniority, regardless of his/her category of employment, shall be given preference.

Section 5. Employees in this category of employment will be required to work holidays that fall on a weekend and will be compensated at the holiday pay rate in accordance with Section 8.01(D).

Section 6. All the provisions contained within the collective bargaining agreement shall apply to weekend employees unless otherwise specified in this Section.

7. SALARY AND INCREMENT RULES

7.01 Promotions

Any employee promoted to a position in a higher job group shall receive a salary at the same increment step reached in his/her former position.

7.02 Demotions

A permanent employee who for any reason is appointed to a position that is in a job group lower than the job group of the position in which he/she is serving shall upon appointment to the lower position receive a salary at the increment step in the salary range in the lower job corresponding to the increment step reached in his/her former position.

7.03 Reinstatement

A. A permanent competitive class employee who has been laid off and subsequently reinstated from a preferred list pursuant to Civil Service rules shall be reinstated at the same salary step as received at the time of layoff.

- B. A permanent employee who has resigned and is subsequently reinstated pursuant to Civil Service rules shall be reinstated at the same salary step received at the time of resignation if reinstated to the same position held at the time of resignation.
- C. An employee who is promoted temporarily or provisionally to a higher position and who is returned to his/her position in a lower grade, shall upon return to the lower position receive a salary at the increment level he/she would have reached had he/she continued to serve continuously in that position.

7.04 Reallocation

Upon the reallocation of a class of positions to a higher job group, the employee or employees serving in the reallocated positions shall receive a salary at the increment step in the higher job group that corresponds with the increment step in which they were serving in the lower group.

7.05 Reclassification

When an employee's class title is reclassified to a higher title and job group, it shall be considered as a new position and a promotion. The salary will then be determined in accordance with the salary rule on promotions.

7.06 Temporary Assignments

An employee assigned by the Employer to perform the duties of a position with higher level responsibilities shall initially perform such duties at his/her permanent rate of pay but on the third (3rd) day of such assignment shall begin to receive the applicable salary increase. However, if the assignment is to an encumbered position from which the incumbent is on authorized leave without pay, such employee will be eligible for the new rate immediately upon actually assuming the assigned position.

Should any General Duty Nurse replace a Team Leader or Charge Nurse for a minimum of four (4) hours, such General Duty Nurse shall receive two dollars and fifty cents (\$2.50) per hour for each hour of said shift.

7.07 Leave of Absence - Military

Any employee who is required to render ordered military duty or to attend a training program or perform other duties under United States or New York State supervision, shall be granted military leave of absence with no loss of time or pay not to exceed thirty (30) days per calendar year pursuant to the Military Law. All employees who are active in the military reserves or National Guard must submit a monthly drill schedule to their immediate supervisor within seven-two (72) hours of its publication.

7.08 Increments

- A. Increments are recommended by the appointing authority on the basis of merit and recognition for measuring up to the department's work performance and attendance standards.
- B. 1. Increments, if granted, shall be effective either January 1 or July 1. For employees on ECMCC wage scale appointed or promoted to a position shall be eligible for their first increment upon approval by the appointing authority after six (6) months of actual service on January 1 or July 1 as the case may be. Employees will be eligible for subsequent increments upon approval by the

appointing authority on a yearly basis for Steps 1-8 of the wage scale provided they have a minimum of nine (9) months of actual service since receiving their last increment. Employees will be eligible for Steps 9-10 upon the approval of the appointing authority as outlined in "G" below.

- 2. Effective January 1, 2009, increments, if granted, shall be effective either January 1 or July 1. Employees on ECMCC wage scale appointed or promoted to a position shall be eligible for their first increment upon approval by the appointing authority after six (6) months of actual service on January 1 or July 1 as the case may be. Employees will be eligible for subsequent increments upon approval by the appointing authority on a yearly basis, for Steps 1-6 of the wage scale, provided they have a minimum of nine (9) months of actual service since receiving their last increment. Employees will be eligible for Steps 7-10 upon approval by the appointing authority every other year as outlined in "G" below.
- C. If an employee who holds a permanent position is serving in another position on a temporary or provisional basis, he/she shall be eligible for increments in that other position provided he/she has met the actual service requirements in that position. However, if such employee resigns, is laid off, terminated or otherwise leaves that other position and is at some future time appointed to that other position, the actual service requirements shall be based upon the date of reappointment with no credit for any prior time spent in such position.
- D. In computing increment eligibility, when appointments are made on January 1 or July 1, and the day falls on a holiday or non-scheduled workday, the increment period will include these days.
- E. Because of payroll procedures that enable the Employer to have a regular payday throughout the year, the increment eligibility period and pay periods may not at all times coincide. In such case the increment credit date is the first day of respective pay period during which January 1 or July 1 falls.
- F. Part-Time Increment Movement. Part-time employees shall be moved as follows:
 - 1. Increments are recommended by the appointing authority on the basis of merit and recognition for measuring up to the department's work performance and attendance standards.
 - 2. Increments, if granted, shall be effective either January 1 or July 1. Employees appointed to a position shall be eligible for their first increment upon approval by the appointing authority after twenty-four (24) months of part-time service on January 1 or July 1 as the case may be. Employees will be eligible for subsequent increments upon approval of the appointing authority on the next January 1 or July 1 after completing twenty-four (24) months of part-time service since their last increment.
 - 3. In computing increment eligibility, when appointments are made on January 1 or July 1, and the day falls on a holiday or non-scheduled workday, the increment period will include these days.
 - 4. Because of payroll procedures that enable the Employer to have a regular payday throughout the year, the increment eligibility period and pay periods may not at all times coincide. In such cases the increment credit date is the first day of respective pay period during which January 1 or July 1 falls.

- 5. Credit for years of service for any nurse shall be limited to January 1, 1981, for purposes of movement throughout the salary range (step movement).
- 6. Twenty-four (24) months of part-time service will equal one (1) year of full-time service.
- 7. One (1) hour worked in a month shall be defined as one (1) month of service for movement through the salary range.
- 8. A two (2) year period, as defined in #6 above, during which no part-time service is performed and no leave of absence has been granted, shall result in the part-time employee being returned to Step One (1) of the salary schedule.

G. Steps Nine (9) and Ten (10)

Step Nine (9) will be defined as eight (8) or more years in the New York State Nurses Association bargaining unit, or two (2) years at Step Eight (8).

Step Ten (10) will be reached after four (4) years at Step Nine (9).

Effective January 1, 2009, ECMCC Employees who had attained Step Ten (10) in the former wage scale and had been at this Step for at least two (2) years shall move to Step Seven (7) of the new wage scale implemented July 1, 2008 and shall move through the remainder of the wage scale as outlined below.

All employees shall move through Steps 7-10 as follows:

- Step Seven (7) will be defined as six (6) or more years in the New York State Nurses Association bargaining unit, or two (2) years at Step Six (6).
- Step Eight (8) will be reached after two (2) years at Step Seven (7).
- Step Nine (9) will be reached after two (2) years at Step Eight (8).
- Step Ten (10) will be reached after two (2) years at Step Nine (9).

7.09 Pay Period

- A. The salaries and wages of employees shall be paid biweekly. In the event this day is a holiday, the preceding day shall be the payday. Every effort shall be made to pay the evening and night shifts a day previous to day shift employees. Every effort shall be made to pay an employee who is on rotation at the same time the other employees on that shift are paid. If an employee is scheduled for a day off on his/her regular payday, every effort will be made to pay the employee before he/she leaves for the day off; however, no check will be released before 3:00 p.m. on the day prior to the scheduled payday.
- B. During the term of this Agreement, the Employer will attempt to create and institute a system which identifies on an employee's pay voucher the following:
 - 1. The current status of his/her benefits; i.e. sick leave, vacation time, personal leave days and compensatory time.
 - 2. An accurate interpretation of the distribution of the employee's gross income stating all premiums and/or differentials.

7.10 New Appointments

If during the life of this Agreement the Employer encounters difficulties in recruiting candidates to fill vacancies in a job classification, the CEO of ECMC may authorize an appointment at a step higher than Step II for the job. Such authorization shall not become effective unless recommended by the Director of Nursing and approved by

the ECMC Board of Directors. No such authorization shall become effective without five (5) workdays advance notification to the Association and to the Erie County Director of Budget and Management.

Prior to January 1, 2009, new hires may begin work at other than the starting rate (start rate is Step 2).

Starting rate will be Step Two (2):

Two (2) years of verifiable experience - one (1) step above the normal starting rate in effect at time of hire (Step 3).

Four (4) years of verifiable experience - two (2) steps above the normal starting rate in effect at the time of hire (Step 4).

Six (6) years of verifiable experience - three (3) steps above the normal starting rate in effect at the time of hire (Step 5).

Eight (8) years of verifiable experience - four (4) steps above the normal starting rate in effect at the time of hire (Step 6).

Ten (10) years of verifiable experience - five (5) steps above the normal starting rate in effect at the time of hire (Step 7).

Employees hired since July 1, 1997 will be granted one (1) additional step effective January 1, 2001. (Steps on wage scale - see Appendix C.)

Effective January 1, 2009, new hires may begin work at other than the starting rate (starting rate is Step One (1)). A registered nurse with experience in the field for which he/she is hired may be started at a step higher than the normal starting step as follows:

Two (2) years of verifiable experience - one (1) step above the normal starting rate in effect at time of hire (Step 2).

Four (4) years of verifiable experience - two (2) steps above the normal starting rate in effect at the time of hire (Step 3).

Six (6) years of verifiable experience - three (3) steps above the normal starting rate in effect at the time of hire (Step 4).

Eight (8) years of verifiable experience - four (4) steps above the normal starting rate in effect at the time of hire(Step 5).

Ten (10) years of verifiable experience - five (5) steps above the normal starting rate in effect at the time of hire (Step 6).

7.11 Certification Reimbursement

The Employer shall pay for the cost of application, registration, and test for required certification courses which the employee successfully completes.

8. MONETARY BENEFITS: COMPENSATION FOR TIME WORKED

8.01 Overtime Compensation

A. All employees who actually work over forty (40) hours per week or eight (8) hours per day performing direct patient care and related activities shall be paid two (2) times such employee's straight time hourly rate plus shift differential, for all such hours worked in excess of forty (40) hours per week or eight (8) hours

per day. All paid leave time except sick leave, including paid lunch hours and compensatory time shall be counted as time worked.

Exceptions to the overtime premium of double time include:

- 1) Time spent in meetings and educational programs;
- 2) The first hour worked beyond the end of a shift. However, once an employee works more than one (1) hour of overtime beyond the end of a shift, the individual will be paid at double time for all overtime hours worked back to the moment overtime began.

For any overtime hours worked not qualifying for double time, the employee's straight time hourly rate plus shift differential will be paid at time and one-half the employee's straight time hourly rate plus shift differential per 8.01(B).

- B. All employees who actually work any hours over forty (40) hours per week or eight (8) hours per day, and such hours do not qualify for double time pursuant to 8.01(A), shall be paid time and one-half (1½) times such employee's straight time hourly rate plus shift differential, for all hours worked in excess of forty (40) hours per week or eight (8) hours per day. Excluded from computation of time worked for purposes of the overtime premium of one and one-half (1½) is all sick leave. All other paid leave time, including paid lunch hours and compensatory time shall be counted as time worked. Holiday time paid, whether the holiday was worked or not, will be considered hours worked for purposes of overtime calculation, for any unscheduled time worked the week of the holiday.
- C. Each employee covered under this contract may request, in writing, compensatory time off in lieu of cash payment for overtime worked in accordance with the limitations provided in this Section. Each employee who requests compensatory time in writing shall be granted the right to accumulate such time up to a maximum of one hundred (100) hours per calendar year. The written request must be filed with the department head by the third Monday in January of the year the employee elects to take compensatory time in lieu of cash payment for overtime. The election to take compensatory time shall remain in effect until the employee revokes it. Revocation may only occur during the month of December in each year. Any employee who revokes his or her request for compensatory time shall be paid for overtime hours worked as provided in Section 8.01 (A or B). New employees shall be paid cash for overtime until the third Monday of January of the year following their date of hire, at which time they shall be eligible to select compensatory time as described above.

Each employee who desires compensatory time off shall accumulate the aforementioned time at the rate of double time or time and one-half (1 $\frac{1}{2}$) for each hour or part thereof worked depending on whether the compensatory time accrued under 8.01(A) (double time) or 8.01(B) (time and one-half). The maximum number of overtime compensatory hours that may be accumulated by an employee at any one time is one hundred (100) per calendar year. This amount shall exclude all compensatory time formerly earned for lunch hours, as provided in Section 6(C), and any compensatory time formerly earned under Section 8.01(D), <u>Holidays</u>.

Once an employee accrues the maximum amount (one hundred (100) hours per calendar year) of accrued compensatory time at any time during the calendar year, he or she shall be paid in cash for each hour or part thereof worked thereafter.

A compensatory time accrual rate of double (2) time or one and one-half (1½) hours of compensatory time for each hour of overtime work shall apply only to time worked in excess of forty (40) hours per week or eight (8) hours per day per 8.01(A and B).

- D. An employee who is required to work on a day celebrated as a holiday as provided in this Agreement, will be paid his/her regular straight time pay plus one and one-half (1½) times his/her straight time hourly rate for every hour actually worked on such holiday.
 - Part-time employees will receive one and one-half (1½) times their regular hourly rate for all hours worked on a holiday.
- E. Employee requests to use compensatory time will be granted based on the order in which requests were made. Employees will request compensatory time off from their supervisor. Employee requests to use compensatory time will not be unreasonably denied. Denials shall be provided in writing to the employee.
- F. Compensatory Time Buyback: Effective upon ratification [July 17, 2023], all registered nurses will be allowed to sell back up to one hundred (100) hours of banked compensatory time under the following circumstances:
 - This option must be requested prior to November 1 (paid the first full pay period in December) and/or prior to April 1 (paid the first full pay period in May), and will be paid via a separate check.
 - 2. Henceforth, there will be only one (1) compensatory time bank. This bank will be defined as the total number of hours deposited in an employee's overtime and holiday compensatory time banks prior to this transition.
- G. Mandatory overtime will only be done in accordance with the law.
- H. Whenever an employee is requested to report for work outside his/her regularly scheduled shift or for emergency duty, he/she will receive a minimum of three (3) hours straight time pay and restricted to one (1) nursing assignment per day. Any additional nursing assignment to the same employee during the same day must be approved by the Director of Nursing.

8.02 Wage Schedule

- A. ECMCC Salary Scale Pay Period after first pay period following ratification [July 17, 2023]: see Appendix C. 2.
- B. ECMCC Salary Scale Pay Period after January 1, 2024: see Appendix C. 3.
- C. ECMCC Salary Scale Pay Period after September 1, 2024: see Appendix C. 4.
- D. ECMCC Salary Scale Pay Period after September 1, 2025: see Appendix C. 5.
- E. ECMCC Salary Scale Pay Period after September 1, 2026: see Appendix C. 6.
- F. ECMCC Salary Scale Pay Period after September 1, 2027: see Appendix C. 7

Effective first pay period following date of ratification [July 17, 2023] the wage scale will be increased by 4.75% across the board.

Effective the first pay period following date of ratification [July 17, 2023] a new wage

scale will be created for the position of Unit Manager, which shall be set at four percent (4.0%) above wage scale N4. Current Unit Managers will be placed at the same step of N4A as they were on N4.

Effective pay period following September 1, 2024 the wage scale will be increased by 4.25% across the board.

Effective pay period following September 1, 2025 the wage scale will be increased by 4.25% across the board.

Effective pay period following September 1, 2026 the wage scale will be increased by 4% across the board.

Effective pay period following September 1, 2027 the wage scale will be increased by 4% across the board.

All bargaining unit members shall be paid a one-time Recognition Payment equivalent to two hundred fifty dollars (\$250) to be paid the first full pay period following date of ratification.

Reaction to Market Changes

In the event benefits and wages are below the regional market, the Employer may opt to present a proposal to NYSNA to increase wages or benefits, using the following procedure:

- a. the Employer will present the proposal to NYSNA in writing;
- b. NYSNA will respond to the proposal within ten (10) calendar days.

Effective January 1, 2024 there will be a step eleven (11) added to the wage scale. The new steps will be calculated after the across the board wage increase has been applied.

Step eleven (11) will be reached after two (2) years at step ten (10) and will be calculated at three (3) percent above step ten (10).

All years of service worked prior to ratification [July 17, 2023] shall be counted in calculating placement on step eleven (11).

9. MONETARY BENEFITS: HEALTH AND RETIREMENT

9.01 Health Insurance

- Effective November 1, 2003, current employees and employees who retired on or after January 1, 2003 shall have the Labor/Management Health Care Coalition Trust Fund as their health insurance provider. Employees shall have a choice among three (3) insurance products: the Enhanced Plan, the Core Plan, or the Value Plan. Retirees shall have the Core Plan, except as indicated in paragraph 8 herein.
- The Employer shall contribute one hundred percent (100%) of the cost of the Core Plan for all eligible employees, and eligible retirees who retired on or after January 1,2003.

Effective March 1, 2009, the Employer shall contribute eighty five percent (85%) of the cost of the Core Plan for all eligible full-time and regular part- time

employees.

Employees hired prior to January 1, 2009 shall receive a monthly stipend to reduce or eliminate their contribution toward their health insurance. This amount shall be equal to fifteen percent (15%) of the monthly premium cost for 2009 for either single or family coverage the employee selects. Employees may opt for either family or single coverage based on their individual needs during open enrollment or upon a qualifying event. The stipend amount(s) are as follows and shall be applied in future years for either option an employee chooses. For example, an employee does not need to select the family plan in 2009 to receive the family stipend in 2010. Likewise, if an employee selects a family plan in 2009 and a single plan in 2010, he/she shall receive the single stipend when the single plan is selected.

Health Plan Elected	Dependent CoverageLevel	Bi-Weekly Stipend
Value	Single	\$25.58
Value	Family	\$74.37
Core	Single	\$28.62
Core	Family	\$80.69
Enhanced	Single	\$28.62
Enhanced	Family	\$80.69

- 3. In addition, employees who choose the Enhanced Plan shall pay the difference in the cost between the Core Plan and the Enhanced Plan. The employee will bear the expense, through bi-weekly payroll deductions, of any amount in excess of the Employer contribution.
- 4. For employees who choose the Value Plan, the Employer shall deposit fifty percent (50%) of the difference in the monthly premium cost between the Core Plan and the Value Plan in an I.RS. Section 105-h account. Monies deposited in such account shall roll over year to year until expended, or until the employee's death, at which time any unexpended funds shall revert to the Employer.
- 5. Open Enrollment: Employees may select from among the insurance plans, annually, during the open enrollment period. The open enrollment period will take place after the annual rates are received from the insurance provider.
- 6. The negotiated provisions of the Collective Bargaining Agreement regarding dental coverage will remain in effect.
- 7. The negotiated provisions of the Collective Bargaining Agreement regarding health insurance waivers shall remain in effect.
- 8. (a) Pre-65 Retirees: The Employer shall pay one hundred percent (100%) of the monthly premium single rate for the Core plan for eligible employees who retire from the Employer until age 65. The Employer shall pay one hundred percent (100%) of the monthly premium family rate for the Core Plan for families of eligible employees who retire from ECMCC service until age 65, upon written proof of family status. In addition, a pre-65 retiree may choose the pre-65 option D plan. Pre 65 employees who choose Option D shall pay the difference in the cost between the core plan and the pre-65 Option D premium.

Employees hired prior to January 1, 2009 and who are or will be eligible to retire

on or before December 31, 2013 shall continue to have their retiree health insurance fully paid by the Employer regardless of their individual date of retirement.

Staff hired prior to 1/1/2009 who obtained thirty (30) years of service with ECMCC on or before 12/31/13 will receive:

- a. Retire after 12/31/2014 receive 60% employer contribution.
- b. Retire after 12/31/2015 receive 70% employer contribution.
- c. Retire after 12/31/2016 receive 75% employer contribution.

Actual names of those eligible are included in Appendix "R". Employees hired prior to January 1, 2009 who are eligible to retire on or after January 1, 2014 shall receive fifty percent (50%) payment toward their retiree health insurance paid by the Employer.

Employees hired after January 1, 2009 shall not receive retiree health insurance paid by the Employer.

(b) Post-65 Retirees: Employees who retire under this Agreement, and their eligible spouses, shall be required to select the designated Medicare Wraparound product that includes prescription drug coverage and basic out-of-network benefits, at age 65. Any employee who retires under this Agreement, and his or her eligible spouse who is under age 65, will be provided with a single Core Plan for the non-age 65 member. A post-65 retiree, and his or her eligible spouse aged 65, may choose from Option A, B, or C as referenced on the attached matrix. Both members must select the same option, and the Employer will pay one hundred percent (100%) of the monthly premium for the single or double rate for Options A, B, or C. In addition, a post-65 retiree who chooses Option D shall pay the difference in the cost between the highest premium of Option A, B, or C, and the Option D premium. Health care coverage will be provided for the lifetime of the retiree. Family coverage will be provided upon written documentation. All other Employer contributions shall be eliminated.

Employees hired prior to January 1, 2009 and who are or will be eligible to retire on or before December 31, 2013 shall continue to have their retiree health insurance fully paid by the Employer regardless of their individual date of retirement.

Staff hired prior to 1/1/2009 who obtained thirty (30) years of service with ECMCC on or before 12/31/13 will receive:

- a. Retire after 12/31/2014 receive 60% employer contribution.
- b. Retire after 12/31/2015 receive 70% employer contribution.
- c. Retire after 12/31/2016 receive 75% employer contribution.

Actual names of those eligible are included in Appendix "R". Employees hired prior to January 1, 2009 who are eligible to retire on or after January 1, 2014 shall receive fifty percent (50%) payment toward their retiree health insurance paid by the Employer.

Employees hired after January 1, 2009 shall not receive retiree health insurance paid by the Employer.

9. The negotiated provisions of the Collective Bargaining Agreement regarding

continued health insurance coverage in certain circumstances (for example: injury or illness) shall remain in effect.

10. The Employer shall provide the GHI Spectrum Plus Full Basic (Diagnostic and Preventative) Dental Plan for each employee covered under this contract in accordance with the type of coverage (single or family) desired by the employee.

The Employer shall provide for one hundred percent (100%) orthodontics and one hundred percent (100%) prosthetics coverage, in addition to the full basic plan indicated above. The Employer shall pay the full cost per month for single coverage and forty three dollars and eight cents (\$43.08) per month for family coverage. Any premium costs above those costs set forth in this Section shall be paid by the employee on a biweekly payroll deduction.

If the Employer offers an improved dental plan to another bargaining unit during the term of this Agreement, the Employer and Association will engage in negotiations on that issue.

11. In the event an employee is disabled from work by accident or illness, the Employer agrees to continue his/her health insurance for the length of his/her accumulated sick leave, plus one hundred twenty (120) days thereafter. Unless qualified otherwise under FMLA, employees must have actually worked two hundred forty (240) regular hours prior to this episode of leave to be covered by this provision.

Employees currently receiving health insurance coverage from the Employer who are unable to work due to an assault in the course of their employment as defined by penal law and in accordance with 11.08 Criminal Assault shall have their current health insurance plan at the current contribution rate continued for the duration of their absence or up to eighteen (18) months, whichever is shorter. An employee who terminates employment for any other reason prior to recovery from this injury or the eighteen (18) month time period will also terminate insurance coverage at the time of the separation.

- 12. Should a permanent ECMCC employee, for whom the ECMCC is providing family health insurance coverage die, the employee's health insurance shall be continued for the employee's survivors during the month the death occurs and for two (2) calendar months thereafter.
- 13. Employees desiring to withdraw from any health insurance coverage shall be permitted to do so upon signing a waiver provided by Human Resources. The Employer shall provide a date/time stamped copy of the signed waiver to the employee. Upon the effective date of such withdrawal, the employee shall receive each month, in lieu of coverage, a payment of sixty-seven dollars (\$67.00) for each calendar month if waiving single coverage, or one hundred dollars (\$100.00) for each calendar month if waiving family coverage. Such payment shall be paid in the last paycheck of each calendar month.
- 14. Consistent with Section 4.03 Regular Part-Time Employees and paragraph 8 of this Section, current regular part-time employees and those retired on or after 1/1/03 will have single coverage paid in full by the Employer with option of family coverage at the employee's expense (the employee will pay the difference between single and family coverage).

Effective March 1, 2009, consistent with Section 4.03 Regular Part-Time Employees, the regular part-time employee shall receive a health benefit equal to that of the full-time employee.

- 15. Part-time employees shall be entitled to purchase health insurance (any option) through payroll deduction, at the group rate.
- 16. Retiree's Sick Leave Conversion (This Subsection 16 shall not be interpreted to diminish anybenefits described in Subsections 1-15 above.)
 - Employees who retire with ten (10) years or more County service shall be eligible for the following:
 - A. Employees with a minimum of eight hundred (800) hours of accumulated sick leave on their record on the day of retirement shall receive on the payday following the day of retirement a cash bonus payment of five thousand dollars (\$5,000).
 - B. Employees who have a minimum of one thousand two hundred (1,200) hours of accumulated sick leave on their record on the day of retirement shall receive on the payday following the day of retirement a cash bonus payment of seven thousand dollars (\$7,000).
 - C. Employees who have a minimum of one thousand eight hundred (1,800) hours of accumulated sick leave on their record on the day of retirement shall receive on the payday following the day of retirement a cash bonus payment of fifteen thousand dollars (\$15,000).
 - D. Employees who have a minimum of two thousand four hundred (2,400) hours of accumulated sick leave as of the date of retirement shall receive thirty thousand dollars (\$30,000) on the payday following the day of retirement as a cash bonus.
 - * Employees who will utilize these funds for the payment of retiree health insurance can elect to have the payment placed into an account to be used for such purpose. The Employer and the Union will upon mutually agreement determine the best account to use for this purpose that is in the best interest of the employer and the employee.
 - Employees who retire with thirty (30) years or more County service and were hired after January 1, 2009 shall be eligible for the following:
 - E. Employees with a minimum of six hundred (600) hours of accumulated sick leave on their record on the day of retirement shall receive on the payday following the day of retirement a cash bonus payment of fifteen thousand dollars (\$15,000).
 - F. Employees who have a minimum of one thousand (1000) hours of accumulated sick leave on their record on the day of retirement shall receive on the payday following the day of retirement a cash bonus payment of twenty thousand dollars (\$20,000).
 - G. Employees who have a minimum of one thousand four hundred (1,400) hours of accumulated sick leave on their record on the day of retirement shall receive on the payday following the day of retirement a cash bonus payment of thirty thousand dollars (\$30,000).
- 17. Upon notification, the Association will appoint representatives to attend County meetings designed to contain rising health insurance costs.

HEALTH INSURANCE WAIVER NEW YORK STATE NURSES ASSOCIATION

NO EMPLOYER PROVIDED HEALTH INSURANCE FOR YOU OR YOUR FAMILY MEMBERS WILL BE CONTINUED UNDER THE EFFECTIVE TERMS OF THIS WAIVER!

I hereby for myself, my heirs, executors and administrators, waive my rights to Employer- provided health insurance coverage pursuant to the Collective Bargaining Agreement between the Erie County Medical Center and the New York State Nurses Association.

I understand the RISK inherent to electing the Health Insurance Waiver Option and assume any and all responsibility for said RISK to myself, my heirs, executors and administrators.

I release any and all rights and claims I may have against the Erie County Medical Center and/or NYSNA and their respective representatives as a result of my waiver of health insurance coverage to which I was previously entitled.

I understand that once this withdrawal of health insurance coverage is in effect, I may re- enter any Employer-provided insurance plan with reasonable notice of my intention to the Employer.

I have read the above waiver and, upon my reading, fully understand its content.

Employee:	Date:	_
NYSNA Representative:	Date:	-
Erie County Commissioner	Date:	

9.02 Retirement Plan

The County shall provide retirement plans known as "The New Career Retirement Plan" - 75-1 with riders 60-B and 41-J, or the "Coordinated-Escalator Plan" whichever is applicable, to all employees in the bargaining unit who are eligible under law.

9.03 NYSNA Tax-Sheltered Annuity

The County will make available a Tax-Sheltered Annuity Plan when legally permitted by the State of New York contingent upon the following conditions being met:

- A. The Association will present to the Department of Labor Relations documented legal proof that members of the Association are entitled to participate in such a plan under appropriate Federal and State laws and regulations.
- B. The provider of such a plan be selected by the Association and the Association shall provide certification indicating approval by the Internal Revenue Service and the State of New York for the carrier so selected by the Association.
- C. All contributions under such plan are the direct obligation, through payroll deduction, of participating employees.
- D. The implementation and continuation of such plan is contingent upon there being no additional costs, direct or indirect to the Employer, over and above that normally attributable to other payroll deductions currently provided to the bargaining unit herein.
- E. Such program is effective only if, and so long as, the percentage of employee participation required under such program, if any, is maintained.

9.04 Sick Leave Conversion

For employees hired after January 1, 2009 (who elect not to participate in sick time conversion 9.01 #16), the number of accumulated sick days in the retirement bank on the date of retirement shall be divided by thirty (30) and the resulting whole number (excluding fractions) shall indicate the number of months for which the Employer will continue the retiree's insurance at no cost to the employee.

10. MONETARY BENEFITS: MISCELLANEOUS

10.01 Shift Differential Pay

A three dollar (\$3.00) per hour shift differential will be granted to all employees on the 11p.m. - 7a.m. (night) and 3p.m. - 11p.m. (evening) shifts. Upon ratification the 11:00 p.m. -7:00 a.m. shift differential will be increased to three dollars and twenty-five cents (\$3.25) per hour.

Effective June 28, 2022, shift differential will be increased to six dollars (\$6.00) per hour for all employees.

Shift differential will be paid to employees for actual hours worked in accordance with the above provisions.

All cash payments, for employees who are entitled to shift differential and who work overtime, shall be made not later than in the next regular payroll check.

10.02 Transportation Allowance

- A. The mileage allowance will be as set forth in the IRS schedule or a minimum of four dollars (\$4.00) per day for all employees. The mileage allowance shall be adjusted to the then current IRS rate when next announced after ratification.
- B. Toll charges reimbursed if supported by appropriate receipts.
- C. Those nurses who park in paid parking facilities while on official business at hospitals in the County are reimbursed if appropriate receipts are submitted with mileage voucher.
- D. The Employer will make every effort to make payment to employees, who have submitted mileage vouchers, no later than thirty (30) days from submission.

10.03 Preceptor Differential

An employee who is assigned by management to be a preceptor shall receive an additional three dollars (\$3.00) per hour for all time spent performing such duties.

Effective the first full pay period following September 1, 2023, this differential will be increased to three dollars and fifty cents (\$3.50) per hour for all time spent performing such duties.

10.04 Nurse Practitioner Annual Expense Reimbursement

The Employer agrees to reimburse Nurse Practitioners in the bargaining unit for up to three hundred dollars (\$300.00) each calendar year for costs associated with recertification, including continuing education units, and/or medical/dental staff dues.

10.05 Two Medication Cart Discretionary Bonus (Terrace View)

ECMCC shall offer all General Duty Nurses, Charge Nurses, Nursing Supervisors and Team Leaders employed at Terrace View Long-Term Care Facility an additional two hundred dollar (\$200.00) discretionary bonus when responsible for two medication carts. General Duty Nurses, Charge Nurses, Nursing Supervisors and Team Leaders must work a minimum of four (4) hours to qualify. Only one (1) discretionary bonus will be paid per employee, per day. This will only be applicable when Licensed Practical Nurse is not also responsible for the same two medication carts.

11. MONETARY BENEFITS: COMPENSATION FOR TIME NOT WORKED

11.01 Vacations

Employees may use their vacation in one (1) hour blocks. Every effort will be made to grant employees their vacation at the requested time subject to their department's responsibility to maintain work coverage and efficient operations for service to the public.

Credits

Vacation credits will accrue and be available for use on a pay period basis for full-time employees after the first pay period of employment, providing they are on a compensable pay status for at least fifty percent (50%) of the pay period. A full-time employee must have worked no less than forty (40) hours (of an eighty {80} hour pay period) and have accruals under the maximum balance as describes within to be eligible to receive vacation hours credit for the pay period. Vacation

credits will be granted per pay period in accordance with the following schedule:

Service	Hours Per Pay Period
From date of employment through completion of two (2) years of service	3.08
From second (2nd) year anniversary date through completion of ten (10) years of service	4.62
From the tenth (10th) year anniversary date through completion of seventeen (17) years of service	6.16
From the seventeenth (17th) year anniversary date through completion of twenty-five (25) years of service	7.70
From the twenty-fifth (25th) year anniversary date through successive years of service	9.23

Department Heads will establish vacation periods, schedules and vacation units based on their work requirements.

Maximum vacation credit balance is to be defined as the calculation of: (a) twenty six (26) pay periods of vacation credit accruals, as determined in accordance with the member's placement on the chart in Section 11.01; plus, (b), the number of hours in the unit member's vacation bank, which cannot exceed one hundred sixty (160) hours. It will be the responsibility of each member to manage his or her vacation accruals and bank, and to keep the combined total under the maximum vacation credit balance. Bargaining unit members at or above the maximum vacation credit balance will not be eligible to accrue additional vacation time.

<u>Pay</u>

Vacation pay shall be the regular straight-time rate of pay in effect for the employee's regular position at the time he/she takes his/her vacation.

Holidays

If a holiday occurs during an employee's vacation, the holiday will not be charged against vacation credits. An employee on paid sick leave, jury duty, paid military duty, paid vacation time or full pay status will be considered as time worked in determining vacation credits.

Transfer to Other Department

If an employee is promoted or transferred to another department, vacation credits will be transferred.

Vacation sell back

Commencing January 1, 2014, and continuing annually thereafter, employees shall be permitted to sell back forty (40) hours of accrued and unused vacation leave at the employee's regular rate of pay. Employees must have been on the Employer's payroll for the entirety of the year of sale and must have a minimum of eighty (80)

hours accrued and unused vacation leave at the time of sale.

Employees must notify their supervisor by September 1st of each year if they wish to sell vacation hours. Payment shall be made by Payroll Period No. 24 of the same year.

Resignation

An employee who resigns, retires or is laid off prior to taking his/her vacation shall be compensated for the accumulated vacation credits. The employee's estate will receive compensation for an employee's unused vacation in case of the death of an employee.

Leave of Absence

A leave of absence without pay or a resignation followed by reinstatement or rehire in the County services within one (1) year shall not constitute an interruption of service for the purposes of this provision; provided, however, that the period of leave without pay between resignation and termination, shall not be counted in determining vacation credits per year or rate per pay period.

Scheduling

The initial vacation sign-up period will be from January 1 through the last day of February each year. Vacation requests submitted during the initial sign-up period will be granted based on bargaining unit seniority. After that date, available dates will be granted on a first come, first serve basis. If a week becomes open which had previously been selected during the sign-up period, that week shall be made available on a rotating seniority basis. Approval of vacation requests submitted after the initial posting period will be given in writing within three (3) weeks from the submission date; if such approval is not presented in writing, the vacation shall be considered denied. A copy of the vacation schedule will be posted on each unit no later than March 30th. If in the pay period preceding the scheduled vacation week or at the commencement of vacation, an employee does not have the vacation accruals sufficient to cover the entire week of vacation they have selected, the scheduled vacation will be cancelled, and the employee will be required to work. If a week of vacation is cancelled less than two (2) weeks prior to the scheduled vacation reposting will be at the discretion of the employer.

An employee who voluntarily transfers into a different vacation unit will have prior approved vacation granted, if possible and practical in the new unit. If it is not possible to grant said vacation, the employee accepting the transfer will forfeit his/her rights to those vacation days regardless of seniority. An involuntarily transferred employee will retain previously granted vacation time.

Forty-five (45) days prior to the initial sign-up period, management will submit the Vacation Guidelines, including all modifications to NYSNA representatives. Any modifications from the previous year's Vacation Guidelines must be submitted electronically and clearly indicated by "bolding" and "underlining" new language and utilizing "strike-through" for language to be deleted. Thirty (30) days prior to the initial sign-up period, management will post the Vacation Guidelines providing copies to New York State Nurses Association representatives. The Director of Nursing will meet, upon request, with NYSNA representatives in order to discuss the groupings and protocol.

At the Erie County Medical Center, during the period starting with the week of Memorial Day and including the week of Labor Day, and between the week

including December 15 through the first week in January, vacations will be initially limited to a maximum length of two (2) weeks at a time (in order to give all employees an opportunity to have time off).

When these entitlements have been honored, any remaining additional "prime time" which remains open, will be granted on a rotating seniority basis to a maximum of one (1) week per employee, per rotation, until all remaining "prime time" are filled.

Part-Time and Per Diem Service

Effective upon ratification a full-time or regular part-time employee who reverts to part-time or per diem service for more than one (1) year and then returns to full-time or regular part-time status shall, upon returning to full-time service, begin to earn vacation credits at the rate of accrual enjoyed just prior to becoming part-time or per diem.

11.02 Holidays

The following holidays shall be observed by all employees in the bargaining unit as paid holidays:

New Year's Day

Independence Day

Martin Luther King, Jr. Day

Labor Day

President's Day Easter Sunday Veterans' Day Thanksgiving Day Christmas Day

Memorial Day Juneteenth

If any of the aforementioned holidays falls on a Saturday, the Employer will observe the holiday on the prior Friday. If a holiday falls on a Sunday, the following Monday will be observed as the holiday. However, for all employees who are not seven (7) days per week, the Easter holiday shall be recognized on Good Friday.

If one (1) of the aforementioned holidays falls on an employee's regularly scheduled day off, such employee will receive eight (8) hours pay at the employee's regular compensation rate for said holiday.

Payment for hours worked on a holiday set forth herein shall be pursuant to Section 8.01(C) of this Agreement. Every effort will be made to distribute the taking of holidays on an equivalent basis.

An employee who is on sick leave the day before or after a holiday will be paid for the holiday.

Notwithstanding the conditions set forth in other portions of this Section, employees in any facility covered by this contract who work a regular schedule of seven (7) days per week shall celebrate holidays on the actual calendar date on which the holiday traditionally falls:

11.03 Personal Leave

Full-time employees including temporary and provisional personnel will become eligible for and receive four (4) days (32 hours) personal leave after one (1) year of continuous service and also become eligible for and receive the same allowance for each succeeding year of employment providing they are on a compensable salary and wage basis for at least six (6) months of continuous service in the

preceding anniversary year and otherwise meet all eligibility requirements.

Personal leave is not cumulative from year to year. Unused personal leave credit shall be added to an employee's accumulated sick leave bank at the end of the employee's anniversary year. This addition does not extend the permissible accumulation of sick leave beyond the maximum provided by this Agreement. Once an employee has reached the maximum sick leave accumulation allowed under Section 11.07 Sick Leave, unused personal leave shall go into an employee's vacation bank.

In order for the department head and/or the immediate supervisor to arrange for adequate work coverage, applications for personal leave must be filed by an employee on a prescribed form with the head of the department or immediate supervisor at least five (5) working days in advance when the requested time is four (4) days, and three (3) working days in advance when the requested time is for three (3) days or less. There shall be no restrictions as to when this leave is to be taken unless stated in this Section. In cases of emergency, the five (5) or three (3) days of advance notice may be waived by the department head. All the requests must receive the approval of the employee's immediate supervisor or department head and shall not be granted in less than one (1) hour units. Personal leave may be used in one (1) hour units, if staffing permits. In cases of personal emergency, when the above time limits have expired, the Employer will make an effort to find a replacement before invoking staffing needs as basis for denial.

In cases of reinstatement or transfers, unused personal leave credits shall be restored or transferred.

11.04 Bereavement Pay

An employee who has a death in the immediate family (parent, spouse, brother, sister, children, grandparent, grandchild, parent-in-law; brother-in-law, sister-in-law, son-in-law, daughter-in-law, foster child, stepparent, domestic partner (as defined by Public Health Law §2961) or other relative who is an actual member of the employee's household) shall be given time off without loss of pay up to a maximum of five (5) consecutive calendar days inclusive of either the date of death or the date of the funeral. Bereavement days taken over the course of the weekend do not have to be made up for purposes of satisfying Article 6(F) Hours of Work.

11.05 Jury Duty Leave/Court Leave

On proof of the necessity of jury service or attending court for other than personal matters, a leave of absence with pay shall be granted to the employee.

Employees will not be required to report for work prior to or subsequent to the performance of their jury duty. Employees required to report for jury service or court will be assigned a Monday through Friday schedule on the employee's regular unit. The employee's work assignment may be switched to his/her regular assignment beginning the Monday following the conclusion of the employee's jury service or court appearance.

When an employee is assigned to the evening or night shift on the day he/she performs jury duty, he/she is to be excused with pay for evening or night shift on that day.

When an employee is required to attend court for other than personal matters, he/she will be permitted a leave of absence with pay, along with reasonable travel

time to and from the court appearance. Staff would be expected to return to work if time remains in their scheduled shift. If a court appearance is necessary for ECMCC matters, and the employee is not working that day, he/she will be compensated at the current overtime rate for their time in court.

Any portions of this Section inconsistent with New York State Law shall be superseded by statute. The balance of this Section will remain in full force and effect.

11.06 Emergency and Disaster Plan

In the event of an emergency, Administration (business hours) or the NCC (off hours) will contact the administrator on call to make the decision to initiate the Disaster Plan (EOC 800). The Incident Command system will be implemented.

In addition to the guidelines contained in the general policy, there are a number of concerns specific to a weather emergency. Among them are: transportation, adequate staffing, food for clients, visitors and staff, sleeping, bathing and laundry facilities for staff, snow removal, and revision of staffing schedules. ECMCC staff are considered essential personnel and are expected to report to work. In the event that staff are unable to report and documentation to that fact can be authenticated by the staff member, the resulting time off from work shall be treated as follows:

- A. Chargeable time off pursuant to the above shall be to one of the following at the option of the employee, if available: compensatory time, personal leave, or vacation.
- B. School closings and lack of adequate childcare would not be a reason for using time under this section.

11.07 Sick Leave

All full-time permanent employees in the bargaining unit shall earn sick leave credits immediately upon entering the service of the Employer at the rate of 4.62 hours per pay period.

An employee may accumulate sick leave up to a maximum of two thousand four hundred (2400) hours.

Employees who use one (1) scheduled shift or less of sick leave in an anniversary year will receive eight (8) hours of additional pay on their anniversary date. The additional pay will be paid at straight time and not counted for the purposes of calculating overtime.

Temporary employees and provisional employees without permanent status will not earn sick leave credits until after the completion of six (6) months of continuous service.

Reasons for Granting Sick Leave

Sick leave with pay shall be granted by the Employer to an employee, when incapacitated or unable to perform the duties of his/her position by reason of:

- A. Sickness or injury;
- B. Serious illness in the employee's immediate family, requiring care and attendance of employee. Immediate family shall include parent, spouse, brother, sister, children or grandparents; or other blood relative who is an actual member of the employee's household. Certification or affidavit, issued by the

attending physician certified to the necessity for the attendance of the employee may be required by the department head if there is reason to suspect abuse of this Section. All requests for sick leave for serious family illness must be requested in writing and such request, including a doctor's affidavit if required, shall be filed with the employee's department head or designee and forwarded to payroll; sick leave for this purpose shall be granted only with the approval of the appointing authority;

- C. Quarantine regulations;
- D. Emergency medical or dental visits;
- E. Maternity;
- F. Doctor's appointments Time off will be allowed to employees for doctor's appointments; however, such time off is limited to a total of twenty-four (24) hours per calendar year and shall require verification of the time and date of the visit(s). Notice of an employee's need to leave to attend a medical appointment shall be provided to his/her department head prior to the posting of the employee's work schedule, except in the case of emergency appointments.

Sick Leave Credits and Charges

Credit for sick leave under this provision shall be allowed at the rate of 4.62 hours per pay period for each pay period during which the employee shall have been on a full pay status for at least fifty percent (50%) of the working days of that pay period.

Charges against sick leave may not be made in units of less than a half (1/2) hour. Requests for use of sick leave shall be submitted on the prescribed form.

Extended Sick Leave

- A. An employee who has completed fifteen (15) years of continuous service may receive such additional sick leave with pay as may be recommended by the employee's department head and approved by the Vice President of Human Resources, but no such additional sick leave shall be approved by the Vice President of Human Resources in excess of a period of five (5) months in addition to sick leave accumulated by such employee.
- B. Employees shall be eligible only once for the additional periods of sick leave granted in accordance with this provision.

Reporting Time

- A. In case of absences, the time for reporting absences shall not be less than two (2) hours prior to the actual start time of the shift. If canceling a voluntary overtime shift, an employee must call in a minimum of two (2) hours prior to the start of the voluntary shift. Failure to call-in two (2) hours prior to the start of the voluntary shift will be counted as an unapproved occurrence. In case of failure to report within the stated time limits, unless for reasons satisfactory to the employee's department head or his/her designee, the absence shall not be deductible from sick leave and shall be considered as time off without pay.
- B. Daily call-in is required each and every day except as outlined in "The Clarification of Policy and Procedure for the Reporting of Absence Under the Sick Leave Provisions" issued by the Department of Personnel as amended by the Commissioner from time to time.

- C. A certificate or affidavit, showing incapacity and inability of the employee to perform his/her duties issued by the attending physician, shall be filed with the department head or designee and forwarded to payroll in case of absence of more than three (3) consecutive workdays. The Employer may check further on any illness regardless of certificate or affidavit.
- D. If an employee fails to submit proof of illness to his/her department head or designee when required to do so, the absence shall not be deductible from sick leave and shall be considered as time off without pay.
- E. If the proof submitted, in the judgment of the Employer, does not justify the employee's absence, such absence shall not be deducted from sick leave and shall be considered time off withoutpay.

Abuse of Sick Leave Benefits

Abuse of sick leave privileges shall be grounds for disciplinary action. In addition, where an employee's absences are such that the Employer has reasonable grounds to believe that an abuse of sick leave may exist, such employee, where practicable, will be notified of such suspected abuse and thereafter may be required, regardless of the duration of the absence, to submit a satisfactory doctor's certificate or affidavit indicating the specific nature of the disability and its duration to his/her department head or designee and forwarded to payroll before such abuse may be charged against the employee's accumulated sick leave balance. The Association will work cooperatively with the Employer to reduce and prevent abuses of sick leave.

Reinstatement of Sick Leave

When an employee is reinstated into the same position or reemployed in the County bargaining unit within one (1) year following resignation or layoff, sick leave credits accumulated at time of resignation or layoff shall be restored.

Medical or Dental Visits

In the case of an emergency which requires the employee to make visitations during his/her working hours, as shall be determined by the employee's department head or his/her designee upon sufficient proof by the employee, time off for medical or dental visits may be granted by the employee's department head or his/her designee. Such absence may be deducted from accumulated sick leave in units of not less than a half (1/2) hour.

Resignation or Retirement

Sick time taken after notice of resignation or retirement shall require a doctor's certificate or an equivalent amount of vacation time will be deducted.

11.08 Assault Pay

In the event that an employee is necessarily absent from duty as a result of an assault, as defined below, upon his/her person during the course and in the discharge of his/her job responsibilities and duties for the Employer resulting in an injury for which the New York State Workers' Compensation Board has allowed benefits as an occupational injury, such employee shall receive up to the first five (5) months of such absence from the date of injury with full pay and benefits, without use of any sick leave credits. Thereafter, he/she shall have all accrued sick leave benefits necessarily used during such absence restored upon his/her return to duty.

The Employer shall provide full health insurance coverage, and/ or supplemental

coverage for any uncovered claims directly related to the Assault for the time period that the Employee is on Assault Pay.

For the purposes of this Article and Section, assault shall be defined as any incident involving forcible and unauthorized physical contact taken against an employee in the workplace.

In order to be eligible to receive assault pay, one of the following must be available or provided to the Leaves Department:

- Police report
- A report from a medical examination
- A witness statement
- An Incident Report/ Confidential Employee Occurrence Report

A Claim must be made within ten (10) working days from the date of the assault.

In no instance shall the Employer discourage or otherwise coerce an employee from exercising their full rights under the law, up to and including pursuing (or "pressing") charges, activating a police report against the individual responsible for the assault, or filing an incident report.

11.09 Tuition Reimbursement

Effective fiscal year following ratification, the tuition reimbursement will be increased and may be subject to employee reimbursement as follows:

ECMC \$250,000 Terrace View LTC \$30,000

If the employee receiving tuition reimbursement resigns from his/her position at ECMCC, the employee will be required to reimburse the Employer based on the following repayment schedule:

An exception to the above would be for the employee to accept a position that the employee qualifies for that is not available at ECMCC within three (3) months of graduation. An employee would not be subject to repayment if they maintain employment within ECMCC regardless if it is a bargaining unit position.

Length of Employment After Course Completion	Required Employee Reimbursement
6 months or less	100%
Over 6 months and less than 12 months	75%
Over 12 months and less than 24 months	50%

Payment will be made within thirty (30) days of the respective semester's due date for grades as per the Tuition Reimbursement Policy. Procedures for reimbursement have been established by the parties and are incorporated herein by reference.

12. LEAVE OF ABSENCE WITHOUT PAY

12.01 Eligibility

Only permanent employees shall be eligible for leaves of absence without pay after satisfactory completion of their probation period.

12.02 Application for Leave Without Pay

Application for leave of absence without pay, for any of the reasons cited in this provision, shall be filed by the employee, on the prescribed form with the head of the employee's department or his/her designee. Such application shall state the reasons for the requested leave and the duration thereof. If approved by the head of the department or his/her designee, the application shall be submitted to the Commissioner of Personnel, and leave of absence shall be granted only when finally approved by the Commissioner of Personnel. It is understood that if the leave is granted, such employee will be permitted to return to the same class title within the same department.

12.03 Child Care Leave

A. Leave

The Commissioner of Personnel or his/her designee shall grant pregnant employees a leave of absence without pay, upon competent medical proof that such employee is unable to perform her regular duties for the period of such disability, not to exceed one (1) year in duration. For these purposes, the Commissioner of Personnel or his/her designee may require suitable medical evidence from the employee's physician at such employee's expense and/or may require that the employee be examined by a physician chosen by the Employer at the Employer's expense.

B. Sick Leave and Vacation Leave

Employees granted maternity leave pursuant to this Section shall be permitted to reduce the amount of leave without pay by the use of any and all accumulated sick leave and vacation leave credits as may be available to them.

C. Extension

At the request of the employee, and after receiving the recommendation of her department head, the Commissioner of Personnel or his/her designee may grant extensions of such leave of absence in accordance with this Section.

D. Following her maternity disability period, an employee shall be granted a three (3) month family leave of absence. Such leave must be requested from the department head, in writing, at least two (2) weeks prior to the end of the disability period. The two (2) week notification requirement shall be waived in cases of emergency.

E. Child Care Leave

A leave of absence without pay for the purposes of child rearing, necessitated by children residing with the employee, shall be granted for a period of up to six (6) months. Once granted, such leave may not be extended and will be granted only twice during an individual's term of employment, regardless of the time used during such leave of absence. The cumulative total time off for one (1) or both leaves shall not exceed six (6) months. Such leaves must be presented in writing to the department head at least thirty (30) calendar days in advance of the commencement of the child-rearing leave.

An employee on a child-rearing leave will notify the department head of his/her intention to return to work at least thirty (30) calendar days prior to the expiration of the leave of absence.

An employee returning to work after a child-rearing leave shall return to the same position in the same department the employee left, if available, and shall be returned to the same step the employee occupied when the leave commenced.

It is further agreed that any employee to be eligible for child-rearing leave must exhaust all accumulated leave time except sick leave prior to commencing such leave of absence. Upon return to work from the leave, all sick leave credits, as well as other contractual rights which were accumulated or effective on the commencement date of the leave, shall be restored on the date the employee returns to work.

12.04 Leave Because of Extended Illness

When an employee has exhausted all of his/her sick leave credits and is still incapacitated and unable to perform the duties of his/her position, the department head or his/her designee may in his/her sole discretion grant leave of absence without pay for a period not to exceed one (1) year, subject to extension pursuant to County Civil Service Rules as amended from time to time. Such decision shall not be arbitrary or capricious.

12.05 Leave for War Work

A permanent employee may, in the sole discretion of the department head, be granted a leave of absence without pay for a period of time, not to exceed one (1) year, to enter the service of the Federal Government or its associated powers in time of war or to engage in war industries for the United States or its associated powers. Such leave of absence, in the sole discretion of the department head or his/her designee, and with the approval of the Commissioner of Personnel, may be renewed for additional periods, not exceeding one (1) year in each instance without requiring such person to return to his/her position in the Civil Service between successive leaves; provided, however, that no such renewal of a leave of absence without pay shall extend beyond six (6) months after the termination of thewar.

12.06 Leave for Education Purposes

- A. On the approval of the department head or his/her designee, permanent employees may be granted leave of absence without pay for a period of one and one-half (1½) years for the purpose of acquiring additional education and training that will increase the usefulness and efficiency of the employee in his/her position.
- B. An additional leave of absence without pay of not more than one (1) year may be granted by the department head, work permitting.

12.07 Education Leave for Veterans

Any veteran who is qualified to receive education or training or vocational rehabilitation under the provisions of any Federal or New York State Law, shall be granted leave of absence without pay for the period of such education or training or vocational rehabilitation, provided that the attendance of a veteran is required at times that will preclude employment in his/her position. Such leave of absence shall not extend beyond a period of four (4) years, nor beyond the period for which the veteran shall be eligible to continue the education or training or vocational

rehabilitation. It shall terminate at any time that the veteran ceases actual attendance at the classes or courses required by the education, training or rehabilitation program. A veteran who has been on such leave of absence shall be reinstated in his/her position, provided he/she makes application for such reinstatement within sixty (60) days after the termination of such leave of absence. He/she may be reinstated at any time after such sixty (60) day period and within one (1) year after termination of such leave of absence in the discretion of the department head.

12.08 Leave of Absence to Service in Another Position in the County Service

Leave of absence without pay may be granted by the department head or his/her designee to a permanent employee in the competitive class to enable such employee to serve temporarily or provisionally in another position in the classified service in the County.

12.09 Leave of Absence to Accept Employment Outside the County Service

Leave of absence shall not be granted to an employee to accept employment outside the County service except for leaves of absence granted under the provision of Section 12.11.

12.10 Adoption Leave

If an employee covered under this Agreement adopts a child, the Employer agrees to give a maximum one (1) year leave of absence as required by State Law.

12.11 Leave of Absence for Civic Purposes

Employees may be granted a leave of absence for one (1) year or the complete term of a duly elected political office. Such leave is to begin from the first date in office.

12.12 Leave for OtherReasons

Leaves of absence without pay, for reasons other than those cited in this provision, may be granted by the department head or his/her designee only in unusual circumstances, which in his/her judgment justifies the granting of such leave. Requests for such leave and the recommendation of the department head shall be submitted to the Commissioner of Personnel, and the granting of such leave shall be subject to the approval of the Commissioner of Personnel.

13. GRIEVANCES AND ARBITRATION

13.01 General

- A. It is the intent of this Article to promote and provide a mutually satisfactory procedure for the settlement of grievances of employees arising out of the meaning, application or operation of this Agreement.
- B. The Association representative shall be permitted to participate in the activity and progress of any grievance in each stage through the final decision. All other labor organizations will be excluded from the grievance procedure involving the unit covered by this Agreement.

13.02 Definitions

- A. "Grievance" shall mean any claimed violation, interpretation, or inequitable application of this Agreement, and any County rule or regulation as amended from time to time other than the Rules for the Classified Civil Service of the County of Erie. This term shall not include any matter involving any employee's rate of compensation except merit increment increases, retirement benefits, disciplinary proceedings, Rules for the Classified Civil Service of the County of Erie, or any other matter which is otherwise reviewable pursuant to law, or any rule or regulation having the force and effect of law as it applies to employees covered by Section 75 of the Civil Service Law.
- B. "Grievant" is an employee who has a grievance. However, if the grievance involves all (or substantially all) of the employees in the bargaining unit or in a division of that unit, the grievant may be the Association. When the grievant is the Association, the grievance shall be moved directly to Step 2.
- C. "Day" refers to a calendar day and not workday.
- D. "Workday" shall mean all days other than Saturdays, Sundays and legal holidays. Saturdays, Sundays and legal holidays shall be excluded in computing the number of legal days in which action must be taken in any stage of the grievance procedure.

13.03 Rights of the Parties

- A. The parties shall exchange any written statements or records used at the second step meeting provided for in this Article at the time of such meeting. Each party shall have access to written statements or records which are presented as evidence by the other party at an arbitration hearing at least *five* (5) working days in advance of such hearing.
- B. The Local Bargaining Unit Chairperson shall receive a copy of any written grievance and any decision rendered in the grievance procedure.
- C. The Employer, Association and/or grievant shall have the right to submit briefs to support or refute allegations of any party.
- D. The Association may have an observer at the arbitration hearing, even though not a part thereof, and shall be granted reasonable notice of the date, time and place of the hearing to the Association's Albany office.
- E. The time limits set forth in this Article are of the essence. They may, however, be extended by mutual agreement of the parties. The failure of the grievant to proceed within the time limit set forth shall terminate the grievance at that step. The failure of the Employer to answer within the time limit set forth will entitle the grievant to proceed to the next step of the grievance procedure, upon notice to the Employer.
- F. The grievant covered by the terms of this Agreement shall have the right, if he/she so desires, to be represented by an Association representative at any step of the grievance procedure subject to the provisions contained in Section 13.01 above.
- G. The Association's designated representative referred to in Section 13.03(F) shall suffer no loss of compensation while attending any meeting in the steps of the Grievance or Arbitration Procedure.

13.04 Grievance Procedure

Step 1

The employee aggrieved shall present his/her grievance in writing, in a form to be provided, setting forth the date, time and place of the alleged grievance, facts of the grievance, the particular section of the Collective Bargaining Agreement and the relief sought to Human Resources within fifteen (15) working days from the occurrence of the grievance or when the employee knew or should have known of the fact situation giving rise to the grievance. At the written request of the employee, the Human Resources Department designee and/or the relevant ECMC Department Head, if such request is made simultaneously with the filing of the grievance, shall hold an informal hearing within ten (10) days after receiving such written request. The Human Resources Department designee or the relevant ECMC Department Head shall render a decision in writing within ten (10) working days of the receipt of the grievance or date of hearing, whichever is later.

Step 2

If the employee is not satisfied with the disposition of the grievance at the preceding step, it is agreed that the employee may appeal the grievance within ten (10) working days of the written step 1 decision. Step 2 grievances shall be heard by the ECMC Vice President of Human Resources and/or designee. A written disposition shall be given by the ECMC Vice President of Human Resources and/or designee within ten (10) days of such meeting.

13.05 Arbitration Procedure

- A. If the Association Committee in Step 2 of the grievance procedure is not satisfied with the disposition of the grievance rendered by the County Labor Relations Committee or VP of Human Resources at such second step meeting, such decision may be appealed to arbitration within fifteen (15) working days of the disposition.
- B. The arbitrator may be selected by mutual agreement between the parties.
- C. In the event the parties fail to mutually agree upon an arbitrator, either party will have the right to request a list of the names from the Public Employment Relations Board. The parties shall follow the procedures established by PERS for selection of an arbitrator.
- D. The arbitrator's decision shall be rendered within thirty (30) days of the hearing or within thirty (30) days of the receipt of the written position of both parties. As timely arbitration decisions are in the best interest of both parties, it is agreed that if a decision is not rendered within the *above* stated time limits, both parties will mutually contact the arbitrator to expedite the award.
- E. The cost of any arbitration hearing will be borne equally by the parties to this Agreement.
- F. The decision of the arbitrator shall be final and binding on both parties.

14. DISCIPLINE AND UNSATISFACTORY WORKPERFORMANCE

14.01 Discharge and Discipline

A. The Employer shall not exercise its right to discharge or otherwise discipline an employee without just cause. Other penalties including written warnings

shall not be issued without just cause.

B. The Employer shall have the right to discharge or otherwise discipline an employee for unsatisfactory work performance when for just cause. Disciplinary action or measure shall include only Oral Reprimand, Written Reprimand, Suspension or Discharge. All records of oral or written reprimands will be removed from any employee's personnel file after two (2) years. Suspensions shall remain in the file, however shall not be considered for appointments, promotions, transfers and progressive discipline after five (5) years.

14.02 Procedure

- A. The Employer will notify in writing to the General Representative of the Association at 155 Washington Ave, Albany, NY 12210, within three (3) working days after it discharges or suspends an employee under this Article, and the reasons thereof, by registered mail.
- B. If the Employer has any reason to reprimand an employee, it shall be done in a manner that will not unduly embarrass the employee before other employees or the public.
- C. If the employee is to be sent home because of a disciplinary suspension or discharge, the Local Representative or his/her designee shall be called in.
- D. The employee shall be entitled to Association representation upon request.

14.03 Disputes as to Discipline

- A. The Association shall have the right to take up a suspension and/or discharge as a grievance at Step 1 of the grievance procedure, and the matter shall be handled through the arbitration step, if deemed necessary by either party.
- B. The parties agree that the sole and exclusive procedure for the settlement of disputes as to discipline shall be through the grievance procedure as outlined in Section 14.03(A).

14.04 Private Hearing

Upon application by the employee involved and/or the Association, an arbitrator in a discipline case shall have the authority to direct that the arbitration shall be held in private.

14.05 Reinstatement and/or Back Pay

An employee found to be unjustly suspended or discharged, or that his/her penalty was too severe, shall be reinstated and compensated for all, part or no compensation for lost time, as may be determined by the arbitrator.

15. MANAGEMENT RIGHTS

Except as expressly limited by other provisions of this Agreement, all of the authority, rights and responsibilities possessed by the Employer are retained by it, including, but not limited to, the right to determine the mission, purposes, objectives and policies of the Employer; to determine the facilities, methods, means and number of personnel for the conduct of the Employer programs; to administer the merit system, including the examination, selection, recruitment, hiring, appraisal, training, retention, promotion, assignment or transfer of employees pursuant to law; to direct, deploy and utilize the work force; to establish specifications for each class of positions, and to classify orre-classify, and to allocate or re-allocate new or existing positions in accordance with law; and to

discipline or discharge employees in accordance with law and the provisions of this Agreement.

16. NO STRIKE CLAUSE

The Association recognizes the status of the Employer employees as "public employees," and the provisions of law applicable thereto.

The Association shall not engage in a strike, nor cause, instigate, encourage or condone one. In the event a strike or work stoppage occurs, the Association shall exert its best efforts to prevent and terminate the same.

No lockout of employees shall be instituted by the Employer during the term of this Agreement.

17. PLEDGE AGAINST DISCRIMINATION AND COERCION

- A. The provisions of this Agreement shall be applied equally to all employees in the bargaining unit without discrimination as to age, sex, marital status, race, color, creed, national origin, political affiliation, veterans' status or handicap. The Association shall share equally with the Employer the responsibility for applying this provision of the Agreement.
- B. All references to employees in this Agreement designate both sexes, and wherever the male gender is used, it shall be construed to include male and female employees.
- C. The Association recognizes its responsibility as bargaining agent and agrees to represent all employees in the bargaining unit without discrimination, interference, restraintor coercion.

18. EQUAL OPPORTUNITIES

The Employer and the Association realize a responsibility to promote and provide equal opportunities for employment. It shall be the positive and continuing policy of the Employer and the Association to assure an equal opportunity in employment regardless of race, color, religion, sex, national origin or age.

19. GENERAL PROVISIONS

19.01 Employee Health Examinations

- A. The Employer agrees to provide at no expense to the employee, the required medical examinations, immunizations and tests as required by Title 10, Chapter V of the State Hospital Code. The employee may elect to have a physician or facility other than the Employer perform such examinations, but at no expense to the County of Erie.
- B. An employee returning to work after an absence of thirty (30) consecutive calendar days will be required to have a health examination, provided by Employer.
- C. Examinations may be required by the employee's department head for an employee returning from an illness of less than thirty (30) days.

19.02 Labor Management Committee

The Administration of the Erie County Medical Center and Terrace View LTC will meet with representatives designated by the Association on a monthly basis to discuss matters of mutual interest.

The Association representatives referred to above shall give sufficient advance notice to their immediate supervisor they are leaving their work area to attend the Department's Labor Management meeting and will sign out and signin.

Agenda items to be discussed will be submitted by the Association and the Administration to the Labor Management Committee seven (7) calendar days before such meeting.

19.03 Parking Facilities

The Employer will provide parking space for employees using their personal cars, while conducting essential Employer business, where physically possible on Employer property.

19.04 Notification of Employees

Every three (3) months during the duration of this Agreement, the Employer will furnish the New York State Nurses Association, Albany office, a list of new employees in the bargaining unit in addition to any change of address of current employees in the unit. Such list of new employees shall contain the name, address, department, classification title and social security number.

It is understood that it is the obligation of an employee to notify the Employer immediately of any change of address, phone number, name and marital status. Failure to do so may result in disciplinary action taken against the employee.

Each month the Employer shall provide a list electronically to the Association's general representative(s) containing the name, department, and title of each bargaining unit member employed by the Employer.

19.05 Travel Policies

The policies and procedures governing travel expenses for employees conducting official Employer business are reflected in the Rules and Regulations issued by and on file in the Finance Department as amended from time to time with a copy and any amendments to be sent to the Association's Albany office.

19.06 Employee Safety and Health

The Employer will observe all applicable health and safety laws and regulations.

The Employer will take all steps practical to protect employee safety and health.

19.07 Uniforms

In work areas where the employees are provided scrub uniforms, the scrub uniforms shall be returned in a timely fashion from the laundry. Each employee shall be provided a minimum of one (1) clean uniform each workday.

19.08 Gym Maintenance and Employee Utilization

The Employer will reopen the gym on April 18, 2022 and maintain access for Employees in accordance with Department of Health regulations.

The parties agree that this provision is subject to change as a result of New York State law and/or Department of Health regulations.

The Employer will establish an electronic system accessible to Employees for the purpose of signing up for appointments.

19.09 On-The-Job Disability

The Employer will comply with the New York State Workers' Compensation Statutes. An employee who is certified by a physician to be not capable of performing the regular duties he or she was performing when the disability began, because of an on-the-job injury, will be required provided the assignment is within the employees' restrictions to return to lighter assignment on the same shift as the employee's regular shift and department, if available, at no reduction in pay until one hundred twenty (120) days or complete recovery (whichever is sooner) when he/she will be allowed to return to his or her previous job in the same shift and unit/program. If a light duty assignment is not available on the employee's regular shift, the employee will be eligible to work light duty on a different shift in a different department, if available.

19.09 On-The-Job Injuries

Employees who incur an injury on the job during their regular shift and must report to the Emergency Department or other treatment area because of this injury, will be paid while in the Emergency Department or other treatment area - only up to the end of their regular shift. Time spent in the Emergency Department or other treatment area after their regular working hours will remain unpaid.

19.10 Disaster Relief

NYSNA and ECMCC agree that nurses can make a significant contribution to disaster victims who require skilled medical care. Because of this belief, NYSNA and ECMCC will cooperate to facilitate leaves of up to two weeks for nurses who volunteer to help in disaster relief for NYSNA or hospital recognized medical missions. Nurses must provide as much advance notice as possible of their request for such leave. Nurses may elect to use paid time off during these leave periods.

Nurses who wish to volunteer for disaster relief/medical missions shall be permitted to take up to two weeks of leave without interruption to medical benefits or loss of seniority under the following circumstances:

- a. Leave is approved by ECMCC in consideration of operational needs.
- b. The term "disaster" refers to a situation that has been designated as a disaster by FEMA (or its state-based analog) or the United Nations.
- c. Nurses must provide documentary proof that the Nurse has engaged in the disaster relief/medical mission for which the Nurse has been granted leave.

Denial of leaves under this section shall be subject to the grievance process, but not subject to arbitration.

19.11 Release for Participation in Hospital Committees

The employer will pay employees at their regular compensation rate for Committee meeting participation or related work authorized by the employer.

For employees who participate in a committee meeting or perform related authorized work during their regular scheduled work time, such time will be considered time worked for the purpose of this agreement.

For employees who participate in a committee meeting or perform related authorized work during their non-regularly scheduled work time, such time will be counted as time worked on the purposes of this agreement as it relates only to premium compensation and overtime.

20. SAVINGS CLAUSE

This Agreement and its component provisions are subordinate to any present or future laws and regulations. If any federal or New York law or regulation, or the final decision of any federal or New York court or administrative agency, affects any provision of this Agreement, each such provision will be deemed amended to the extent necessary to comply with such law, regulation or decision, but otherwise this Agreement will not be affected.

21. TERMINATION AND MODIFICATION

Unless otherwise specified, the provisions of this Agreement shall be effective as of the date of its execution and shall remain in full force and effect through the 31st day of December, 2027. It shall be automatically renewed from year to year thereafter unless either party shall notify the other in writing ninety (90) days prior to this Agreement's termination date that it desires to modify, alter or amend this Agreement.

22. PRINTING OF THE COLLECTIVE BARGAINING AGREEMENT

The Employer shall pay the Association the current cost of producing the Collective Bargaining Agreement for each copy of the Agreement ordered by the Employer. The Association shall present a per copy cost calculation to the Employer as soon as it is available. The Parties shall exchange modified drafts until a jointly completed Agreement is fully executed. In no case shall a fully executed Agreement be finalized later than six (6) months from the date of Employer ratification.

23. EXECUTION

Signed by Employer and Association.

ERIE COUNTY MEDICAL CENTER CORP	NEW YORK STATE NURSES ASSOCIATION
Ву	By faturo fane
Title: CEO	Title Executive Director
Date 12/5/2024	Date

APPENDICES AGREEMENT SIGNATURE PAGE

Erie County and New York State Nurses Association agree to add the following appendices to the Collective Bargaining Agreement between the parties that expires on December 31, 2027.

- .	<u> </u>
Appendix "A"	NYSNA Title Listings
Appendix "B"	Dues Deduction Form
Appendix "C"	 ECMCC Salary Scale Pay Period June 19, 2022 ECMCC Salary Scale Pay Period after July 30, 2023 ECMCC Salary Scale Pay Period after January 1, 2024 ECMCC Salary Scale Pay Period after September 1, 2024 ECMCC Salary Scale Pay Period after September 1, 2025 ECMCC Salary Scale Pay Period after September 1, 2026 ECMCC Salary Scale Pay Period after September 1, 2027
Appendix "D"	Memorandum of Agreement: Deferred Annuity Plan
Appendix "E"	Memorandum of Agreement: Ten, Twelve and Thirteen Hour Shifts
Appendix "F"	Memorandum of Agreement: Registered Nurse On-Call Coverage: OR, PACU, CAPD, Hemodialysis/Plasmapheresis and Interventional Services (Cardiac Cath and Angio)
Appendix "G"	Memorandum of Agreement: Individual Overtime Agreement
Appendix "H"	Memorandum of Agreement: Per Diem CRNA Agreement
Appendix "l"	Memorandum of Agreement: Nurse Anesthetist Shift Coverage
Appendix "J"	Memorandum of Agreement: CRNA Coverage for Saturday Trauma Call
Appendix "K"	Memorandum of Agreement: CRNA Wage and Benefit Adjustment
Appendix "L"	Memorandum of Agreement: Per Diem Registered Nurse Agreement
Appendix "M"	Memorandum of Agreement: Agency Registered Nurse Program
Appendix "N"	Memorandum of Agreement: Registered Nurse On-Call Coverage - SANE
Appendix "O"	Memorandum of Agreement: Adjunct Educator Pay
Appendix "P"	Memorandum of Agreement: NYSNA Vacation Accruals
Appendix "Q"	Memorandum of Agreement: Clinical Ladder Program
Appendix "R"	Memorandum of Agreement: Employees Eligible for Employer 30 Year Contribution

Appendix "S"	Memorandum of Agreemen	t: Nurse Staffin	g Committee								
Appendix "T"	Staffing Grids										
Appendix "U"	Memorandum of Agreemen Hiring Program	t: Nurse Appre	nticeship and Community Based								
Appendix "V"	Memorandum of Agreement: Substance Abuse/Dependency Leave of Absence										
Appendix "W" Memorandum of Agreement: Referral Bonus											
Memorandum of	Agreement: Pandemic Prepa	ration and Safe	ty								
Memorandum of	Agreement: Workplace Violer	nce Prevention									
Signature below Collective Bargai		orporate the ap	opendices set forth above into the								
Signed by Emplo	yer and Association.										
ERIE COUNTY N	MEDICAL CENTER CORP	NEW YORK S	STATE NURSES ASSOCIATION								
Ву		By Ja	turo Lane								
Title:	CEO	Title	Executive Director								
Date 12	2/5/2024	Date	10/22/2024								

NYSNA TITLE LISTINGS

Title	Classification	Pay Grade
Admissions Coordinator- Long-Term Care	С	N2
Anesthetist	NC	See App K
Anesthetist per-diem	NC	See App H&K
Assistant Head Nurse	NC	N2
Assistant Nurse Case Manager	C	N2
Charge Nurse	NC	N2
Charge Nurse - Ambulatory Care	С	N2
Clinical Documentation Specialist	С	N2
Clinical Nurse Specialist (Behavioral Health)	C	N6
Clinical Nurse Specialist (Critical Care)	C	N6
Clinical Nurse Specialist (Emergency Services)	С	N6
Clinical Nurse Specialist (Rehabilitation)	С	N6
Clinical Patient Care Liaison	С	N5
Clinical Resource Nurse (RPT)	С	N2
Clinical Resource Nurse (Emergency Services)	С	N2
Clinical Teacher	С	N5
Coordinator of Stroke and Sepsis Quality Care	С	N4
Director of Admissions – Medical Rehabilitation Unit	C	N4
Denial Management Nurse	С	N2
Employee Health Nurse	NC	N3
EP Lab Services Coordinator	С	N7
Emergency Department Patient Flow Team Leader	С	N3
General Duty Nurse	NC	N1
Head Nurse	NC	N3
HIV Training Coordinator	С	N3
Hospital Nurse Screener	С	N2
Infection Prevention Practitioner	C	N3
Infection and Wound Care Preventionist (Long-Term Care)	С	N4
Inservice Education Coordinator	С	N4
Lead Clinical Documentation Specialist	C	N3

Manager - Infection Control	С	N4
Minimum Data Set (MDS) Director	С	N4
Minimum Data Set (MDS) Specialist	C	N3
Nurse Case Manager	С	N4
Nurse Case Manager - Acute Geriatrics	C	N4
Nurse Case Manager - AIDS Services	С	N4
Nurse Case Manager - Surgery	С	N4
Nurse Clinician - AIDS Center	C	N4
Nurse Clinician - Psychiatry	C	N4
Nurse Clinician - Renal	C	N4
Nurse Clinician Surgery	С	N4
Nurse Clinician Surgical Intensive Care	С	N4
Nurse Clinician Transplant	С	N4
Nurse Practitioner (ECMC)	NC	N6
Nurse Practitioner Cardiovascular Lab	С	N6
Nurse Practitioner (ECMC) RPT	NC	N5
Nurse Practitioner Emergency Department	С	N5
Nurse Practitioner Neurology	C	N5
Nurse Practitioner Plastic & Reconstructive	С	N6
Nurse Practitioner Rehabilitation Services	C	N6
Nurse Practitioner Transplant	С	N6
Nurse Practitioner Orthopedic Services	С	N6
Nurse Practitioner Wound Care/Ostomv	С	N6
Nurse Navigator (Department of Plastic and Reconstructive Surgery)	С	N4
Nurse Recruiter	C	N4
Nursing Care Coordinator	С	N5
Nursing Care Coordinator – Emergency Department	С	N5

Nursing Care Coordinator - Quality Improvement	С	NS				
Nursing Informatics Manager	С	N5				
Nursing Informatics Manager (Critical Care)	С	N3				
Nursing Inservice Instructor (Critical Care)	C	N3				
Nursing Inservice Instructor (ECMC)	С	N3				
Nursing Inservice Instructor (Psychiatry)	С	N3				
Nursing Inservice Instructor Operating Room	С	N3				
Nursing Inservice Instructor - Psychiatry	С	N3				
Nursing Supervisor- Long-Term Care	С	N4				
Nursing Team Leader	NC	N3				
Nursing Team Leader Dialysis Services	С	N3				
Nursing Team Leader - ElectroPhysics Lab	С	N3				
Nursing Team Leader - Orthopedics	С	N3				
Nursing Team Leader- LTC	С	N3				
Nursing Team Leader – Ostomy/Wound Care	С	N3				
Nursing Team Leader (Radiology)	С	N3				
Nursing Team Leader Vascular Access	С	N3				
Nursing Team Leader- Utilization Review	С	N3				
Patient Safety Clinical Investigation Coordinator	С	N4				
Prospective Payment Coordinator - Rehab	С	N1				
Population Health Clinical Data Analyst	С	N4				
Quality Care Coordinator	C	N4				
Registered Nurse Behavioral Health Data Reviewer	C	N3				
Registered Nurse Long Term Care Data Reviewer	C	N2				
Registered Nurse Medical/Surgical Data Reviewer	C	N2				

Registered Nurse Renal/Hemodialysis Data Reviewer	C	N2
Registered Nurse Trauma Data Reviewer	С	N2
Registered Nurse Trauma Data Registrar	С	N3
Rehabilitation Nurse Screener	C	N2
Renal Services Support Nurse	С	N2
Renal Services Support Nurse-Home Hemodialysis Training	С	N2
Renal Services Support Nurse-Home Peritoneal Training	С	N2
Senior Nurse Practitioner	NC	N4
Senior Nurse Practitioner Neurology	C	N6
Senior Nurse Practitioner – Burn Treatment	С	N7
Staff Development Specialist	C	N5
Staff Development Coordinator - LTC	С	N4
Transplant Coordinator	С	N4
Trauma Injury Prevention & Education Coordinator	С	N3
Trauma Nurse Liaison	С	N4
Trauma Program Manager	С	N6
Unit Manager (Medical/Surgical)	С	N4A
Unit Manager (Operating Room)	С	N4A
Unit Manager Post Anesthesia Care	С	N4A
Unit Manager Ambulatory Care	С	N4A
Unit Manager Behavioral Health	С	N4A
Unit Manager Burn Treatment	C	N4A
Unit Manager Cardiac Cath Laboratory	С	N4A
Unit Manager Critical Care	C	N4A
Unit Manager Rehab Medicine	С	N4A
Unit Manager - Emergency Department	С	N4A
Unit Manager Hemodialvsis	С	N4A
Unit Manager Long term Care	C	N4A

Unit Manager Transitional Care Unit	С	N4A
Unit Manager Transplant	С	N4A
Utilization Coordinator	С	N2
Utilization Management Coordinator	С	N2
Utilization Review Nurse	NC	N1

All title include Full-time, PT, RPT

NC= Non-competitive

C= Competitive

PJC= Pending jurisdiction classification

Denotes classification status at time of the ratification.



NYSNA Membership Application & Authorization for Deduction of Dues

When you join NYSNA you will become part of an organization of nursing professionals, 40,000 members strong, advocating for our patients and our practice in our work place, in our communities and in the legislature. TES, I hereby request and accept membership in NYSNA, or if I am already a member, I hereby reaffirm my membership. YES, I hereby agree to support NYSNA by paying dues and hereby authorize the deduct in each regular payroll from my salary or wages the applicable dues and any duly authorized change in that amount from my salary/wages and to remit that amount to NYSNA. YES, I grant permission to receive email and text notifications from NYSNA of upcoming events, CE classes, Certification Review, and other free or low cost educational/professional events for NYSNA members. Last Four Digits of SSN# Full Name Home Address Zip Code City Phone (Home) Phone (Cell) **Email Address** Work Area Status: Title Full Time Part Time Other Dated: Signed: I understand that my membership is continuous and carries over from year to year. My authorization to deduct union dues is subject to the terms of applicable States law. In order to withdraw from membership and/or revoke my voluntary dues deduction authorization, I must send written notice in the thirty day period prior to the anniversary date of my membership or as otherwise provided by applicable law, by certified mail, return receipt requested, addressed to the NYSNA Membership Department at 131 West 33rd Street, New York, NY 10001 and to the payroll department of my employer. The NYSNA Political Action Committee (NYSNA-PAC) is a nonpartisan committee that assists candidates for state and local office who support nursing's positions on health care. With your permission, each year \$10 of your annual dues is allocated to support NYSNA-PAC. Your \$10 helps increase NYSNA's political clout in promoting quality patient care, protecting RNs' rights and defending our scope of practice. It is allocated automatically from your dues, you will not pay any

additional amount. Under applicable law you must be a U.S. citizen or legal permanent resident (green card holder) to approve this allocation. If you do

not or cannot approve, check the box below. 🔲 I am unable or do not wish to approve the allocation described above.

APPENDIX "C" 1.

ECMCC 7.75% Pay Period following June 19, 2022

,	۷ Z		c	ם מ	:	ć	ηZ		4	> Z	•	ωΖ		:	١	<u> ح</u>	:		Z			
hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	2.5% inc.	
\$60.644	\$4,851.51	\$126,139	\$54.198	\$4,335.80	\$112,731	\$49.515	\$3,961.20	\$102,991	\$45.912	\$3,672.96	\$95,497	\$41.717	\$3,337.35	\$86,771	\$38,736	\$3,098.86	\$80,570	\$35.772	\$2,861.77	\$74,406	Step 1	
\$63.699	\$5,095.94	\$132,494	\$56.920	\$4,553.58	\$118,393	\$51.978	\$4,158.25	\$108,114	\$48.162	\$3,852.97	\$100,177	\$43.797	\$3,503.79	\$91,099	\$40.636	\$3,250.89	\$84,523	\$37.521	\$3,001.65	\$78,043	Step 2	
\$65.220	\$5,217.64	\$135,659	\$58.271	\$4,661.71	\$121,204	\$53.208	\$4,256.68	\$110,674	\$49.271	\$3,941.70	\$102,484	\$44.816	\$3,585.27	\$93,217	\$41.594	\$3,327.56	\$86,517	\$38.401	\$3,072.11	\$79,875	Step 3	
\$66.739	\$5,339.15	\$138,818	\$59.625	\$4,770.03	\$124,021	\$54.457	\$4,356.52	\$113,270	\$50.383	\$4,030.62	\$104,796	\$45.851	\$3,668.07	\$95,370	\$42,553	\$3,404.23	\$88,510	\$39.269	\$3,141.53	\$81,680	Step 4	
\$68.762	\$5,500.97	\$143,025	\$61.420	\$4,913.58	\$127,753	\$56.084	\$4,486.70	\$116,654	\$51.902	\$4,152.12	\$107,955	\$47.219	\$3,777.52	\$98,215	\$43.815	\$3,505.21	\$91,135	\$40.439	\$3,235.16	\$84,114	Step 5	
\$70.814	\$5,665.15	\$147,294	\$63.258	\$5,060.62	\$131,576	\$57,770	\$4,621.58	\$120,161	\$53.459	\$4,276.74	\$111,195	\$48.647	\$3,891.77	\$101,186	\$45.137	\$3,610.99	\$93,886	\$41.670	\$3,333.59	\$86,673	Step 6	
\$72.942	\$5,835.35	\$151,719	\$65.158	\$5,212.64	\$135,529	\$59.503	\$4,760.24	\$123,766	\$55.064	\$4,405.13	\$114,533	\$50.107	\$4,008.57	\$104,223	\$46.489	\$3,719.12	\$96,697	\$42.916	\$3,433.24	\$89,264	Step 7	
\$75.132	\$6,010.55	\$156,274	\$67.105	\$5,368.44	\$139,579	\$61.296	\$4,903.69	\$127,496	\$56.721	\$4,537.66	\$117,979	\$51.613	\$4,129.05	\$107,355	\$47.888	\$3,831.02	\$99,607	\$44.210	\$3,536.76	\$91,956	Step 8	
\$77.381	\$6,190.46	\$160,952	\$69.127	\$5,530.17	\$143,784	\$63.138	\$5,051.01	\$131,326	\$58.424	\$4,673.95	\$121,523	\$53.165	\$4,253.19	\$110,583	\$49.319	\$3,945.56	\$102,585	\$45.533	\$3,642.63	\$94,708	Step 9	
\$/9./06	\$6,376.49	\$165,789	\$/1.193	\$5,695.48	\$148,082	\$65.021	\$5,201.72	\$135,245	\$60.172	\$4,813.74	\$125,157	\$54.744	\$4,3/9.51	\$113,867	\$50.808	\$4,064.62	\$105,680	\$46.900	\$3,751.99	\$97,552	Step 10	

7	Z		c	n Z	!	c	n Z		>	:4	z	4	Z		<u>ر</u>	Z		^	ح د			Z	:		
hourly	hiweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized		
\$63.524	\$5,081.95	\$132,131	\$56.772	\$4,541.75	\$118,086	\$51.867	\$4,149.35	\$107,883	\$49.980	\$3,998.39	\$103,958	\$48.093	\$3,847.43	\$100,033	\$43.698	\$3,495.88	\$90,893	\$40.576	\$3,246.05	\$84,397	\$37.471	\$2,997.71	\$77,940	Step 1	
\$66.725	\$5,338.00	\$138,788	\$59.623	\$4,769.87	\$124,017	\$54,447	\$4,355.76	\$113,250	\$52.448	\$4,195.87	\$109,093	\$50.450	\$4,035.98	\$104,936	\$45.878	\$3,670.22	\$95,426	\$42.566	\$3,405.30	\$88,538	\$39.303	\$3,144.23	\$81,750	Step 2	
\$68.318	\$5,465.47	\$142,102	\$61.039	\$4,883.14	\$126,962	\$55.736	\$4,458.87	\$115,931	\$53.674	\$4,293.90	\$111,641	\$51.612	\$4,128.93	\$107,352	\$46.945	\$3,755.57	\$97,645	\$43.570	\$3,485.62	\$90,626	\$40.225	\$3,218.03	\$83,669	Step 3	ECMCC 4.7
\$69.909	\$5,592.76	\$145,412	\$62.458	\$4,996.61	\$129,912	\$57.043	\$4,563.46	\$118,650	\$54.910	\$4,392.76	\$114,212	\$52.776	\$4,222.07	\$109,774	\$48.029	\$3,842.30	\$99,900	\$44.574	\$3,565.93	\$92,714	\$41.134	\$3,290.75	\$85,559	Step 4	ECMCC 4.75% Pay Period following
\$72.028	\$5,762.27	\$149,819	\$64.337	\$5,146.98	\$133,821	\$58.748	\$4,699.82	\$122,195	\$56,557	\$4,524.58	\$117,639	\$54.367	\$4,349.35	\$113,083	\$49.462	\$3,956.95	\$102,881	\$45.896	\$3,671.70	\$95,464	\$42.360	\$3,388.82	\$88,109	Step 5	riod follow
\$74.178	\$5,934.24	\$154,290	\$66.262	\$5,301.00	\$137,826	\$60.514	\$4,841.11	\$125,869	\$58.256	\$4,660.50	\$121,173	\$55,999	\$4,479.89	\$116,477	\$50.958	\$4,076.63	\$105,992	\$47.281	\$3,782.51	\$98,345	\$43.649	\$3,491.93	\$90,790	Step 6	ing July 30, 2023
\$76.407	\$6,112.53	\$158,926	\$68,253	\$5,460.24	\$141,966	\$62,329	\$4,986.35	\$129,645	\$60.004	\$4,800.36	\$124,809	\$57.680	\$4,614.37	\$119,974	\$52.487	\$4,198.98	\$109,174	\$48.697	\$3,895.78	\$101,290	\$44.954	\$3,596.32	\$93,504	Step 7	, 2023
\$78.701	\$6,296.06	\$163,697	\$70.293	\$5,623.44	\$146,209	\$64.208	\$5,136.62	\$133,552	\$61.811	\$4,944.91	\$128,568	\$59.415	\$4,753.20	\$123,583	\$54.065	\$4,325.18	\$112,455	\$50.162	\$4,012.99	\$104,338	\$46.309	\$3,704.76	\$96,324	Step 8	
\$81.056	\$6,484.51	\$168,597	\$72.411	\$5,792.85	\$150,614	\$66.137	\$5,290.93	\$137,564	\$63.668	\$5,093.45	\$132,430	\$61.200	\$4,895.97	\$127,295	\$55.690	\$4,455.22	\$115,836	\$51.662	\$4,132.97	\$107,457	\$47.595	\$3,815.66	\$99,207	Step 9	
\$83.492	\$6,679.38	\$1/3,004	\$170.001	\$5,960.01	\$100,110	\$00.110	\$5,448.80	\$141,669	\$65.570	\$5,245.59	\$136,385	\$63.030	\$5,042.39	\$131,102	\$57.344	\$4,587.53	\$119,276	\$53.227	\$4,257.69	\$110,700	\$49.128	\$3,930.21	\$102,186	Step 10	

	N 7			N ₆			N ₅			N4A			<u>x</u>			Z3			N2			Z			
hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized		
\$63.524	\$5,081.95	\$132,131	\$56.772	\$4,541.75	\$118,086	\$51.867	\$4,149.35	\$107,883	\$49.980	\$3,998.39	\$103,958	\$48.093	\$3,847.43	\$100,033	\$43,698	\$3,495.88	\$90,893	\$40.576	\$3,246.05	\$84,397	\$37.471	\$2,997.71	\$77,940	Step 1	
\$66.725	\$5,338.00	\$138,788	\$59,623	\$4,769.87	\$124,017	\$54.447	\$4,355.76	\$113,250	\$52.448	\$4,195.87	\$109,093	\$50,450	\$4,035.98	\$104,936	\$45.878	\$3,670.22	\$95,426	\$42.566	\$3,405.30	\$88,538	\$39.303	\$3,144.23	\$81,750	Step 2	
\$68.318	\$5,465.47	\$142,102	\$61.039	\$4,883.14	\$126,962	\$55.736	\$4,458.87	\$115,931	\$53.674	\$4,293.90	\$111,641	\$51.612	\$4,128.93	\$107,352	\$46.945	\$3,755.57	\$97,645	\$43.570	\$3,485.62	\$90,626	\$40.225	\$3,218.03	\$83,669	Step 3	ECMCC
\$69.909	\$5,592.76	\$145,412	\$62,458	\$4,996.61	\$129,912	\$57.043	\$4,563.46	\$118,650	\$54,910	\$4,392.76	\$114,212	\$52.776	\$4,222.07	\$109,774	\$48.029	\$3,842.30	\$99,900	\$44.574	\$3,565.93	\$92,714	\$41.134	\$3,290.75	\$85,559	Step 4	Pay Peri
\$72.028	\$5,762.27	\$149,819	\$64.337	\$5,146.98	\$133,821	\$58.748	\$4,699.82	\$122,195	\$56.557	\$4,524.58	\$117,639	\$54.367	\$4,349.35	\$113,083	\$49.462	\$3,956.95	\$102,881	\$45.896	\$3,671.70	\$95,464	\$42.360	\$3,388.82	\$88,109	Step 5	ECMCC Pay Period following Janua
\$74.178	\$5,934.24	\$154,290	\$66.262	\$5,301.00	\$137,826	\$60.514	\$4,841.11	\$125,869	\$58.256	\$4,660.50	\$121,173	\$55.999	\$4,479.89	\$116,477	\$50.958	\$4,076.63	\$105,992	\$47.281	\$3,782.51	\$98,345	\$43.649	\$3,491.93	\$90,790	Step 6	ing Janua
\$76.407	\$6,112.53	\$158,926	\$68.253	\$5,460.24	\$141,966	\$62.329	\$4,986.35	\$129,645	\$60.004	\$4,800.36	\$124,809	\$57.680	\$4,614.37	\$119,974	\$52.487	\$4,198.98	\$109,174	\$48.697	\$3,895.78	\$101,290	\$44.954	\$3,596.32	\$93,504	Step 7	ry 1, 2024
\$78.707	\$6,296.06	\$163,697	\$70.293	\$5,623.44	\$146,209	\$64.208	\$5,136.62	\$133,552	\$61.811	\$4,944.91	\$128,568	\$59.415	\$4,753.20	\$123,583	\$54.065	\$4,325.18	\$112,455	\$50.162	\$4,012.99	\$104,338	\$46.309	\$3,704.76	\$96,324	Step 8	
\$81.056	\$6,484.51	\$168,597	\$/2.411	\$5,/92.85	\$150,614	\$66.137	\$5,290.93	\$137,564	\$63,668	\$5,093.45	\$132,430	\$61.200	\$4,895.97	\$127,295	\$55.690	\$4,455.22	\$115,836	\$51.662	\$4,132.97	\$107,457	\$47.696	\$3,815.66	\$99,207	Step 9	
\$83.492	\$6,679.38	\$1/3,664	\$/4.5/5	\$5,800,UI	\$100,110	\$68.110	\$5,448.80	\$141,669	\$65.5/0	\$5,245.59	\$136,385	\$03.030	\$5,042.39	\$131,102	\$57.344	\$4,587.53	\$119,276	\$53.221	\$4,257.69	\$110,700	\$49.128	+-	+-	Step 10	
786.CO	\$6,879.76	\$1/8,8/4	\$/50.01Z	\$76.043 \$76.043	\$139,770	\$/O.100	\$5,612.26	\$145,919	\$67.537	\$5,402.96	\$140,477	\$04.92	\$5,193.00	\$135,035	\$59.064	\$4,725.16	\$122,854	\$54.818	\$4,385.42	\$114,021	\$50.601	\$4,048.12	\$105,251	Step 11	

	N7	- William Company		N ₀			N ₅			N4A			<u>X</u>			N S			Z N			Z			
hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized		
\$66.224	\$5,297.94	\$137,746	\$59.185	\$4,734.78	\$123,104	\$54.071	\$4,325.70	\$112,468	\$52.104	\$4,168.32	\$108,376	\$50.137	\$4,010.95	\$104,285	\$45.556	\$3,644.45	\$94,756	\$42.300	\$3,384.01	\$87,984	\$39.064	\$3,125.11	\$81,253	Step 1	
\$69.561	\$5,564.86	\$144,686	\$62.157	\$4,972.59	\$129,287	\$56.761	\$4,540.88	\$118,063	\$54.677	\$4,374.20	\$113,729	\$52.594	\$4,207.51	\$109,395	\$47.828	\$3,826.21	\$99,481	\$44.375	\$3,550.03	\$92,301	\$40.973	\$3,277.86	\$85,224	Step 2	п
\$71.222	\$5,697.76	\$148,142	\$63.633	\$5,090.67	\$132,358	\$58.105	\$4,648.37	\$120,858	\$55.955	\$4,476.39	\$116,386	\$53.805	\$4,304,41	\$111,915	\$48.940	\$3,915.18	\$101,795	\$45.422	\$3,633.76	\$94,478	\$41.935	\$3,354.80	\$87,225	Step 3	CMCC 4.2
\$72.881	\$5,830.45	\$151,592	\$65.112	\$5,208.96	\$135,433	\$59.468	\$4,757.41	\$123,693	\$57.243	\$4,579.46	\$119,066	\$55.019	\$4,401.51	\$114,439	\$50.070	\$4,005.60	\$104,146	\$46.469	\$3,717.48	\$96,655	\$42.883	\$3,430.61	\$89,196	Step 4	5% Pay P
\$75.090	\$6,007.16	\$156,186	\$67.072	\$5,365.72	\$139,509	\$61.244	\$4,899.56	\$127,389	\$58,961	\$4,716.88	\$122,639	\$56.677	\$4,534.20	\$117,889	\$51.564	\$4,125.12	\$107,253	\$47.847	\$3,827.75	\$99,522	\$44.161	\$3,532.85	\$91,854	Step 5	eriod folk
\$77.331	\$6,186.45	\$160,848	\$69.079	\$5,526.29	\$143,683	\$63.086	\$5,046.86	\$131,218	\$60.732	\$4,858.57	\$126,323	\$58.379	\$4,670.28	\$121,427	\$53,124	\$4,249.89	\$110,497	\$49.291	\$3,943.26	\$102,525	\$45.504	\$3,640.34	\$94,649	Step 6	owing Sep
\$79.654	\$6,372.32	\$165,680	\$71.154	\$5,692.31	\$148,000	\$64.978	\$5,198.27	\$135,155	\$62.555	\$5,004.37	\$130,114	\$60.131	\$4,810.48	\$125,073	\$54.718	\$4,377.44	\$113,813	\$50.767	\$4,061.35	\$105,595	\$46.865	\$3,749.17	\$97,478	Step 7	ECMCC 4.25% Pay Period following September 1, 2024
\$82.045	\$6,563.64	\$170,655	\$73.280	\$5,862.44	\$152,423	\$66.937	\$5,354.92	\$139,228	\$64.438	\$5,155.06	\$134,032	\$61.940	\$4,955.21	\$128,835	\$56.362	\$4,509.00	\$117,234	\$52.294	\$4,183.55	\$108,772	\$48.278	\$3,862.21	\$100,417	Step 8	2024
\$84.501	\$6,760.10	\$175,763	\$/5.488	\$6,039.05	\$157,015	\$68.94/	\$5,515.80	\$143,411	\$66.374	\$5,309.92	\$138,058	\$63.801	\$5,104.05	\$132,705	\$58,057	\$4,644.57	\$120,759	\$53.858	\$4,308.62	\$112,024	\$49.723	\$3,977.82	\$103,423	Step 9	
\$87.041	\$6,963.25	\$181,045	\$11.145	\$6,219.57	\$101,709	\$71,000	\$5,680.37	\$147,690	\$68.357	\$5,468.53	\$142,182	\$65.709	\$5,255.69	\$136,674	\$59.781	\$4,782.50	\$124,345	\$55,483	\$4,438.64	\$115,405	\$51.216	\$4,097.25	\$106,528	Step 10	
\$89.652	\$7,172.15	\$186,476	\$40.077	\$6,406.16	9100,000	\$/3.133	\$5,850.78	\$152,120	\$70.407	\$5,632.59	\$146,447	\$67.680	\$5,474.39	\$140,774	\$67.5/5	\$4,925.98	\$128,075	\$57.147	\$4,5/1.80	\$118,867	\$52.752	\$4,220.16	\$109,724	Step 11	

ECMCC 4.25% Pay Period following September 1, 2025

	N7			N6			Z5			N4A			Z 4			N ₃			N2			Z			
hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized														
\$69.039	\$5,523.10	\$143,601	\$61.700	\$4,936.00	\$128,336	\$56,369	\$4,509.54	\$117,248	\$54.318	\$4,345.48	\$112,982	\$52.268	\$4,181.41	\$108,717	\$47.492	\$3,799.34	\$98,783	\$44.098	\$3,527.83	\$91,724	\$40.724	\$3,257.93	\$84,706	Step 1	
\$72.517	\$5,801.37	\$150,836	\$64.799	\$5,183.92	\$134,782	\$59.173	\$4,733.87	\$123,081	\$57.001	\$4,560.10	\$118,563	\$54,829	\$4,386.33	\$114,045	\$49.860	\$3,988.82	\$103,709	\$46.261	\$3,700.90	\$96,224	\$42.715	\$3,417.17	\$88,846	Step 2	
\$74.249	\$5,939.91	\$154,438	\$66.338	\$5,307.03	\$137,983	\$60.574	\$4,845.93	\$125,994	\$58,333	\$4,666.64	\$121,333	\$56.092	\$4,487.34	\$116,671	\$51.020	\$4,081.58	\$106,121	\$47.352	\$3,788.19	\$98,493	\$43.717	\$3,497.38	\$90,932	Step 3	
\$75.978	\$6,078.24	\$158,034	\$67.879	\$5,430.34	\$141,189	\$61.995	\$4,959.60	\$128,949	\$59.676	\$4,774.08	\$124,126	\$57.357	\$4,588.57	\$119,303	\$52.198	\$4,175.84	\$108,572	\$48.443	\$3,875.48	\$100,762	\$44.705	\$3,576.41	\$92,987	Step 4	
\$78.281	\$6,262.47	\$162,824	\$69.922	\$5,593.77	\$145,438	\$63.847	\$5,107.79	\$132,803	\$61.467	\$4,917.35	\$127,851	\$59.086	\$4,726,90	\$122,899	\$53.755	\$4,300.44	\$111,811	\$49.880	\$3,990.43	\$103,751	\$46.037	\$3,683.00	\$95,758	Step 5	
\$80.617	\$6,449.37	\$167,684	\$72.014	\$5,761.16	\$149,790	\$65.767	\$5,261.35	\$136,795	\$63,313	\$5,065.06	\$131,692	\$60.860	\$4,868.77	\$126,588	\$55.381	\$4,430.51	\$115,193	\$51,386	\$4,110.85	\$106,882	\$47 438	\$3,795.05	\$98,671	Step 6	
\$83.039	\$6,643.14	\$172,722	\$74.178	\$5,934.23	\$154,290	\$67.740	\$5,419.19	\$140,899	\$65.213	\$5,217.06	\$135,644	\$62.687	\$5,014.93	\$130,388	\$57.043	\$4,563.48	\$118,650	\$52.924	\$4,233.96	\$110,083	\$48.856	\$3,908.51	\$101,621	Step 7	
\$85.532	\$6,842.59	\$177,907	\$76.395	\$6,111.59	\$158,901	\$69.781	\$5,582.51	\$145,145	\$67.177	\$5,374.15	\$139,728	\$64.573	\$5,165.80	\$134,311	\$58.758	\$4,700.63	\$122,216	\$54.517	\$4,361,35	\$113,395	\$50.329	\$4,026.35	\$104,685	Step 8	
\$88.093	\$7,047.41	\$183,233	\$78.696	\$6,295.71	\$163,688	\$71.878	\$5,750.22	\$149,506	\$69,195	\$5,535.59	\$143,925	\$66.512	\$5,320.97	\$138,345	\$60.525	\$4,841.96	\$125,891	\$56.147	\$4,491.74	\$116,785	\$51.836	\$4,146.88	\$107,819	Step 9	
\$90.740	\$7,259.19	\$188,739	\$81.049	\$6,483.90	\$168,581	\$/4.022	\$5,921./9	\$153,967	\$/1.262	\$5,700.94	\$148,225	\$68.501	\$5,480.10	\$142,483	\$62.322	\$4,985.76	\$129,630	\$57.841	╄		\$53.392	\$4,271.38	\$111,056	Step 10	
\$93.462	\$7,475.96	\$194,401	\$83.480	\$6,678.42	\$1/3,639	\$/6.243	\$6,099.44	\$158,586	\$/3.400	\$5,871.97	\$152,671	\$/0.556	\$5,644.50	\$146,757	\$64.192	\$5,135.33	\$133,519	\$59.576	\$4,766.10	\$123,919	\$54.994	\$4,399.52	\$114,388	Step 11	

П
Ω
\leq
Ξ
4
8
ૅ
37
ECMCC 4.00% Pay Period following September 1, 2026
~
ק
Ō
<u> </u>
ਰੌ
\equiv
ĕ
₹:
Ö
co
O.
졏
<u> </u>
_ <u>⊒</u>
90
ä
Ŋ
0.2
Ö

	N7			N6			N5			N4A			V 4			Nω			N N			Z			
hourly	biweekly	annualized																							
\$71.800	\$5,744.02	\$149,345	\$64.168	\$5,133.44	\$133,470	\$58.624	\$4,689.92	\$121,938	\$56.491	\$4,519.30	\$117,502	\$54.358	\$4,348.67	\$113,065	\$49.391	\$3,951.32	\$102,734	\$45.862	\$3,668.94	\$95,393	\$42.353	\$3,388.24	\$88,094	Step 1	
\$75.418	\$6,033.42	\$156,869	\$67.391	\$5,391.28	\$140,173	\$61.540	\$4,923.23	\$128,004	\$59.281	\$4,742.51	\$123,305	\$57.022	\$4,561.79	\$118,606	\$51.855	\$4,148.38	\$107,858	\$48.112	\$3,848.94	\$100,072	\$44.423	\$3,553.85	\$92,400	Step 2	
\$77.219	\$6,177.51	\$160,615	\$68.991	\$5,519.31	\$143,502	\$62.997	\$5,039.77	\$131,034	\$60.666	\$4,853.30	\$126,186	\$58.335	\$4,666.84	\$121,338	\$53.061	\$4,244.84	\$110,366	\$49.246	\$3,939.72	\$102,433	\$45.466	\$3,637.27	\$94,569	Step 3	
\$79.017	\$6,321.37	\$164,356	\$70.594	\$5,647.56	\$146,837	\$64.475	\$5,157.98	\$134,107	\$62.063	\$4,965.05	\$129,091	\$59.651	\$4,772.11	\$124,075	\$54.286	\$4,342.87	\$112,915	\$50.381	\$4,030.50	\$104,793	\$46.493	\$3,719.46	\$96,706	Step 4	
\$81.412	\$6,512.96	\$169,337	\$72.719	\$5,817.52	\$151,255	\$66.401	\$5,312.10	\$138,115	\$63.925	\$5,114.04	\$132,965	\$61.450	\$4,915.98	\$127,815	\$55.906	\$4,472.46	\$116,284	\$51.876	\$4,150.05	\$107,901	\$47.879	\$3,830.32	\$99,588	Step 5	
\$83.842	\$6,707.35	\$174,391	\$74.895	\$5,991.60	\$155,782	\$68.398	\$5,471.80	\$142,267	\$65.846	\$5,267.66	\$136,959	\$63.294	\$5,063.52	\$131,652	\$57.597	\$4,607.73	\$119,801	\$53.441	\$4,275.29	\$111,157	\$49.336	\$3,946.86	\$102,618	Step 6	
\$86.361	\$6,908.87	\$179,631	\$77.145	\$6,171.60	\$160,462	\$70.449	\$5,635.96	\$146,535	\$67.822	\$5,425.74	\$141,069	\$65.194	\$5,215.52	\$135,604	\$59.325	\$4,746.02	\$123,396	\$55.041	\$4,403.31	\$114,486	\$50.811	\$4,064.85	\$105,686	Step 7	
\$88.954	\$7,116.30	\$185,024	\$79.451	\$6,356.05	\$165,257	\$72.573	\$5,805.81	\$150,951	\$69.864	\$5,589.12	\$145,317	\$67.155	\$5,372.43	\$139,683	\$61.108	\$4,888.65	\$127,105	\$56.698	\$4,535.80	\$117,931	\$52.343	\$4,187.41	\$108,873	Step 8	
\$91.616	\$7,329.30	\$190,562	\$81.844	\$6,547.54	\$170,236	\$74.753	\$5,980.23	\$155,486	\$71.963	\$5,757.02	\$149,682	\$69.173	\$5,533.81	\$143,879	\$62.945	\$5,035.64	\$130,927	\$58,393	\$4,671.41	\$121,457	\$53.909	\$4,312.76	\$112,132	Step 9	
\$94.369	\$7,549.56	\$196,288	\$84.291	\$6,743.26	\$175,325	\$76.983	\$6,158.66	\$160,125	\$74.112	\$5,928.98	\$154,154	\$71.241	\$5,699.30	\$148,182	\$64.815	\$5,185.19	\$134,815	\$60.155	1		\$55.528	\$4,442.23	\$115,498	Step 10	
\$97.201	\$7,776.04	\$202,177	\$86.819	\$6,945.55	\$180,584	\$79.293	\$6,343.42	\$164,929	\$/6.336	\$6,106.85	\$158,778	\$/3.3/9	\$5,870.28	\$152,627	\$66.759	\$5,340.75	\$138,859	\$61.959	\$4,956.75	\$128,875	\$57.194	\$4,5/5.50	\$118,963	Step 11	

_
ECMCC
\leq
1CC
4
4.00%
% T
ay
-
Peri
0
ਰੰ
=
Period following
OC
ğ
September
<u>be</u>
1, 2027
2027
4

	N7			N ₆			N5			N4A			Z 4			몺			N N			Z			
hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized	hourly	biweekly	annualized		
\$74.672	\$5,973.79	\$155,318	\$66.735	\$5,338.78	\$138,808	\$60.969	\$4,877.52	\$126,816	\$58.751	\$4,700.07	\$122,202	\$56,533	\$4,522.61	\$117,588	\$51.367	\$4,109.37	\$106,844	\$47.696	\$3,815.70	\$99,208	\$44.047	\$3,523.77	\$91,618	Step 1	
\$78.434	\$6,274.76	\$163,144	\$70.087	\$5,606.93	\$145,780	\$64.002	\$5,120.16	\$133,124	\$61.653	\$4,932.21	\$128,237	\$59.303	\$4,744.26	\$123,351	\$53.929	\$4,314.31	\$112,172	\$50.036	\$4,002.90	\$104,075	\$46.200	\$3,696.01	\$96,096	Step 2	
\$80.308	\$6,424.61	\$167,040	\$71.751	\$5,740.08	\$149,242	\$65.517	\$5,241.36	\$136,275	\$63.093	\$5,047.43	\$131,233	\$60.669	\$4,853.51	\$126,191	\$55.183	\$4,414.63	\$114,781	\$51.216	\$4,097.31	\$106,530	\$47.285	\$3,782.76	\$98,352	Step 3	
\$82.178	\$6,574.23	\$170,930	\$73.418	\$5,873.46	\$152,710	\$67.054	\$5,364.30	\$139,472	\$64.546	\$5,163.65	\$134,255	\$62.037	\$4,963.00	\$129,038	\$56.457	\$4,516.58	\$117,431	\$52.396	\$4,191.72	\$108,985	\$48.353	\$3,868.24	\$100,574	Step 4	
\$84.669	\$6,773.48	\$176,111	\$75.628	\$6,050.22	\$157,306	\$69.057	\$5,524.59	\$143,639	\$66,483	\$5,318.60	\$138,284	\$63.908	\$5,112.62	\$132,928	\$58.142	\$4,651.35	\$120,935	\$53.951	\$4,316.05	\$112,217	\$49,794	\$3,983.53	\$103,572	Step 5	
\$87,196	\$6,975.64	\$181,367	\$77.891	\$6,231.27	\$162,013	\$71.133	\$5,690.67	\$147,957	\$68.480	\$5,478.37	\$142,438	\$65.826	\$5,266.06	\$136,918	\$59.901	\$4,792.04	\$124,593	\$55.579	\$4,446.30	\$115,604	\$51.309	\$4,104.73	\$106,723	Step 6	
\$89.815	\$7,185.22	\$186,816	\$80.231	\$6,418.46	\$166,880	\$73.267	\$5,861.40	\$152,396	\$70.535	\$5,642.77	\$146,712	\$67.802	\$5,424.14	\$141,028	\$61.698	\$4,935.86	\$128,332	\$57.243	\$4,579.45	\$119,066	\$52.843	\$4,227.44	\$109,913	Step 7	
\$92.512	\$7,400.95	\$192,425	\$82.629	\$6,610.30	\$171,868	\$75.475	\$6,038.04	\$156,989	\$72.659	\$5,812.69	\$151,130	\$69.842	\$5,587.33	\$145,271	\$63.553	\$5,084.20	\$132,189	\$58,965	\$4,717.23	\$122,648	\$54.436	\$4,354.90	\$113,228	Step 8	
\$95.281	\$7,622.47	\$198,184	\$85.118	\$6,809.44	\$177,045	\$77.743	\$6,219.44	\$161,705	\$74.841	\$5,987.30	\$155,670	\$71.939	\$5,755.16	\$149,634	\$65.463	\$5,237.07	\$136,164	\$60.728	\$4,858.27	\$126,315	\$56.066	\$4,485.27	\$116,617	Step 9	
\$98,144	\$7,851.54	\$204,140	\$87.662	\$7,012.99	\$182,338	\$80.063	\$6,405.01	\$166,530	\$//.0//	\$6,166.14	\$160,320	\$74.091	\$5,927.28	\$154,109	\$67.407	\$5,392.60	\$140,208	\$62.561			\$5/./49	\$4,619.92	\$120,118	Step 10	
\$101.089	\$8,087.09	\$210,264	\$90.292	\$7,223.38	\$187,808	\$82.464	\$6,597.16	\$1/1,526	\$79.389	\$6,351.13	\$165,129	\$76.314	\$6,105.09	\$158,732	\$69.430	\$5,554.38	\$144,414	\$64.438	\$5,155.01	\$134,030	\$59.482	\$4,758.52	\$123,722	Step 11	

APPENDIX "D"

MEMORANDUM OF AGREEMENT
BETWEEN THE
NEW YORK STATE NURSES ASSOCIATION AND
COUNTY OF ERIE
And
ERIE COUNTY MEDICAL CENTER CORPORATION

(Deferred Annuity Plan)

This Memorandum of Agreement modifies the collective bargaining agreement between the parties dated January 1, 2023 - December 31, 2027.

It is hereby agreed and understood that the Employer shall provide a payroll deduction column for NYSNA employees who elect to enter a Deferred Annuity Plan.

It is further agreed and understood that the County of Erie shall negotiate with the NYSNA concerning the selection of a provider for such a plan immediately following the date that the final rules and regulations are approved by the Deferred Compensation Board and as soon as localities can legally implement such a Deferred Compensation Plan.

APPENDIX "E"

MEMORANDUM OF AGREEMENT BETWEEN THE NEW YORK STATE NURSES ASSOCIATION AND COUNTY OF ERIE And

(Ten. Twelve and Thirteen Hour Shifts)

This Memorandum of Agreement modifies the collective bargaining agreement between the parties dated January 1, 2023 - December 31, 2027.

It is agreed by and between the Employer and the Association that if the Director of Nursing of a department determines the need for variable work hours, they may be instituted. All current employees who work in the affected areas will be offered an opportunity to participate in the variable (flexible) work schedules being used in their work areas, as long as patient care requirements are met.

The following articles of the current Agreement will be amended as follows:

6. Hours of Work

- A. A normal workday shall not exceed twelve (12) consecutive hours or ten (10) consecutive hours in any twenty-four (24) hour period. The normal workweek shall not exceed forty (40) hours for 10-hour employees. The normal pay period will not exceed eighty (80) hours for 12-hour employees.
- B. Employees will be assigned a thirty (30) minute lunch break.
- C. All employees work schedule shall provide for two (2) fifteen (15) minute rest periods.

8.1 Overtime

Ten (10) Hour Employees: All work over forty (40) hours in a week or ten (10) hours in a day performing direct patient care and related activities shall be paid two (2) times such employees' straight time hourly rate, plus shift differential.

Exceptions to the overtime premium of double time include:

- 1. Time spent in meetings and educational programs;
- 2. The first hour worked beyond the end of a shift. However, once an employee works more than one (1) hour beyond the end of a shift the individual will be paid at double time for all overtime hours worked back to the moment overtime began.

All employees who work over forty (40) hours in a week or ten (10) hours in a day, and such hours do not qualify for double time, shall be paid at time and one-half ($1\frac{1}{2}$) times the employee's straight hourly rate, plus shift differential for all such hours.

Twelve (12) Hour Employees: All work over eighty (80) hours in a pay period or over twelve (12) hours on a 12-hour day or over eight (8) hours on a scheduled 8-hour day performing direct patient care and related activities will be paid at two (2) times such employees' straight time hourly rate, plus shift differential. Sick leave is excluded from the computation of time worked for purposes of overtime premium for the given pay week in which the employee calls in sick.

Exceptions to the overtime premium of double time include:

- 1. Time spent in meetings and educational programs;
- 2. The first hour worked beyond the end of a shift. However, once an employee works more than one (1) hour of overtime beyond the end of a shift the individual will be paid at double time for all overtime hours worked back to the moment overtime began.

All employees who work over eighty (80) hours in a pay period or over twelve (12) hours on a 12-hour day or over eight (8) hours on a scheduled 8-hour day, and such hours do not qualify for double time, shall be paid at time and one-half (1½) times the employee's straight hourly rate, plus shift differential for all such hours.

10.01 Shift Differential Pay

Twelve (12) Hour Employees: Shift differential will be paid at the prevailing rate for all hours worked that fall during the 3p.m. - 11p.m. or 11p.m. - 7a.m. shifts.

Ten (10) Hour Employees: When an employee's shift causes him/her to work four (4) or more hours beyond the traditional shifts in the work area, the employee shall be paid the higher differential rate for the entire shift.

11.01 Vacations

Vacation time will be deducted from accumulated time in twelve (12) hour increments for each day, and six (6) hour increments for a half (1/2) day for twelve (12) hour shift employees; in ten (10) hour increments for each day, and five (5) hour increments for a half (1/2) day for ten (10) hour shift employees.

11.03 Personal Leave

Personal leave will be deducted from the thirty-two (32) hour annual allowance in twelve (12) hour increments for a day and six (6) hour increments for a half (1/2) day for twelve (12) hour shift employees; in ten (10) hour increments for a day and five (5) hour increments for a half (1/2) day for ten (10) hour shift employees. The remaining hours which do not equal a day may be taken at the employee's option.

11.07 Sick Leave

Sick leave will be deducted from accumulated time in twelve (12) hour amounts for a day, and six (6) hour amounts for a half (1/2) day for twelve (12) hour employees and ten (10) hours for full day and five (5) hours for a half (1/2) day for ten (10) hour employees.

Use of sick time may not be in amounts of less than one (1) hour.

THIRTEEN HOUR SHIFT

6. Hours of Work

- A. A normal workday shall not exceed thirteen (13) consecutive hours in any twenty-four (24) hour period. The normal work period shall not exceed seventy-eight (78) hours in a two (2) week pay period.
- B. Employees will be assigned a thirty (30) minute lunch break.
- C. All employees work schedule shall provide for two (2) fifteen (15) minute rest periods.

8.1 Overtime

Thirteen (13) Hour Employees. All work over seventy-eight (78) hours in a pay period or over thirteen (13) hours in a day shall be paid at two (2) times such employees' straight time hourly rate, plus shift differential.

Exceptions to the overtime premium of double time include:

- 1. Time spent in meetings and educational programs;
- 2. The first hour worked beyond the end of a shift. However, once an employee works more than one (1) hour of overtime beyond the end of a shift the individual will be paid at double time for all overtime hours worked back to the moment overtime began.

All employees who work over seventy-eight (78) hours in a pay period or over thirteen (13) hours on a 13-hour day or over eight (8) hours on a scheduled 8-hour day, and such hours do not qualify for double time, shall be paid at time and one-half (1½) times the employee's straight hourly rate, plus shift differential for all such hours.

10.01 Shift Differential Pay

Thirteen /13) Hour Employees: Shift differential will be paid at the prevailing rate for all hours worked that fall during the 3p.m. - 11p.m. or 11p.m. - 7a.m. shifts.

11.01 Vacations

Vacation time will be deducted from accumulated time in thirteen (13) hour increments for each day, and six and one-half (6½) hour increments for a half (1/2) day.

11.03 Personal Leave

Personal leave will be deducted from the thirty-two (32) hour annual allowance in thirteen (13) hour increments for a day and six and one-half (6½) hour increments for a half (1/2) day.

11.7 Sick Leave

Sick leave will be deducted from accumulated time in thirteen (13) hour increments for a day and six and one-half ($6\frac{1}{2}$) hour increments for a half (1/2) day.

The Employer and Association shall meet to discuss the feasibility of a shift pilot within a certain unit and shift within thirty (30) calendar days of receipt by the Vice President (Administrator of Terrace View) of a petition signed by the majority of employees on a unit on the same shift requesting to pilot a change in shift. If the pilot program is approved by the Senior Vice President, all current employees who work in the affected area and shift will be offered an opportunity to participate in the variable (flexible) work schedules being used in their work areas, as long as patient care requirements are met.

APPENDIX "F"

MEMORANDUM OF AGREEMENT BETWEEN THE NEW YORK STATE NURSES ASSOCIATION AND ERIE COUNTY MEDICAL CENTER CORPORATION

Registered Nurse On-Call Coverage
Operating Room, PACU, CAPD, Hemodialysis/Plasmapheresis
and Interventional Services (Cardiac Cath and Angio)

This Memorandum of Agreement replaces the previous MOAs between the parties for the departments above on the subject of on-call coverage. The following constitutes an agreement between the New York State Nurses Association ("NYSNA") and the Erie County Medical Center Corporation ("ECMCC") covering compensation for On-call.

- A. On-call hours will be during the off hours of a department or at time the unit is closed for daily operations including Saturday, Sunday, and Holidays coverage.
- B. These on-call hours shall be compensated at the rate of one (1) hour of straight time pay for every four (4) hours of on-call and for each hour so assigned.
- C. In addition to the on-call pay, if an employee in this rotation schedule is activated, upon clocking in, he/she shall receive a minimum of four (4) hours call in pay and all such hours will be paid double time (2X) regardless of absences prior to the call-in. In the case that an employee's on-call shift is adjacent to their regular worked shift, if the employee is retained for patient care more than fifteen (15) minutes past their regular worked shift, "on-call" shall be considered activated.
- D. The employee shall be paid applicable shift differential for the hours worked as per section 10.01 shift differentials.
- E. After being called in and working a minimum of two (2) hours within eight hours prior to the employee's scheduled shift the employee may receive rest pay and accumulated leave time, by utilizing the option below:
 - i. Receive up to four (4) hours of call-in rest pay and can elect to use up to four (4) hours of accumulated leave time to insure that the employee has eight (8) hours off in between call activation end and the scheduled shift.
 - ii. In times where the call-in period overlaps the scheduled shift, work one half (½) the hours of the scheduled shift after the call in activation work period ends and thereafter be released from work and receive up to one half (½) the duration of their scheduled shift in call-in rest pay.
- F. An employee who is called in within three (3) hours or less prior to the beginning of his/her shift will be paid the actual hours worked prior to his/her regularly scheduled shift.
- G. The on-call schedule will be made out by the Nursing Manager/designee in one (I) month segments. Assignments will be evenly distributed. Any schedule changes between peers can be made subject to the approval of the nursing manager. Any unforeseen occurrence, i.e., sudden illness, will be dealt with via the Nursing management. Except in emergency situations, the Hospital will give employees a minimum of two (2) weeks' notice should it be necessary to

- change the on-call schedule.
- H. Employees will be issued on call beepers, if no response is made to a "beep" within five (5) minutes the employee will be called by the nursing office or department. It is the responsibility of the person on call to insure that their pager number and number is current and in working order. Employees who wish to utilize their personal cell phones as a contact number will be permitted, and therefore not be issued a beeper.
- I. Employees shall be present in their department and prepared to begin duty no later than sixty minutes (60) after being contacted.
- J. On-call rooms will be available. Employees who wish to use an on-call room should notify their manager ahead of time or the off shift nursing supervisor for any last minute needs.
- K. Employees on call shall not be called in to fill in for short staffing in other departments. Employees on call will not be called in for the purpose of relieving other employees for meal periods.

Department specific on-call details:

1. Interventional department (cath lab/angio suite):

- a. On-call hours will be on a weekly basis rather than daily. Weekly call will begin at 5PM Friday and continue until 7AM the following Friday. Holiday coverage will begin at 11PM the night before the day on which the holiday is celebrated, and will extend for a twenty four (24) hour period. Regular weekly assignments will resume after the holiday coverage.
- b. Employees shall be present in their department and prepared to begin duty no later than thirty minutes (30) after being contacted.

2. Surgical Services Operating Room RN - Two call teams, Trauma A and Trauma B

- c. Monday- Friday and Minorholidays,
 Trauma A on call hours will be 3:00PM- 7:00AM or 5:00PM -7:00AM depending on work shift, Trauma B call Monday- Friday and Minor holidays will be I IPM-7AM,
 - Saturday, Day shift 7AM-3PM Trauma A, 3PM-7AM Trauma A and B Sunday, Day shift 7AM-3PM Trauma A, 3PM-7AM Trauma A and B
- d. Major Holidays, Trauma A and Trauma B call 24 Hours 7AM-7AM (New Years Day, Memorial Day, 4th of July, Labor Day, Thanksgiving and Christmas).
 - Call assigned and rotated between Trauma A and B equally
 - It is understood that call hours start after your assigned workshift
- e. Employees shall be present in their department and prepared to begin duty no later than thirty minutes (30) after being contacted.

3. PACU

Employees shall be present in their department and prepared to begin duty no later than thirty minutes (30) after being contacted.

APPENDIX "G"

MEMORANDUM OF AGREEMENT BETWEEN
THE
NEW YORK STATE NURSES ASSOCIATION
AND
COUNTY OF ERIE
And
ERIE COUNTY MEDICAL CENTER CORPORATION

(Individual Overtime Agreement)

This Memorandum of Agreement modifies the collective bargaining agreement between the parties dated January 1, 2023 – December 31, 2027.

It is agreed that individual nurses in Erie County may elect to be exempt from the provisions of Article 8, Section 8.01 by mutual agreement between ECMCC and NYSNA. Such exemption shall become effective only upon the signing of an "Individual Overtime Agreement," as follows:

SUBJECT: INDIVIDUAL OVERTIME AGREEMENT dueto It is agreed that the nature of his/her responsibilities and job requirements as cannot realistically meet his/her duties and responsibilities by being scheduled eight (8) hours a day, five (5) days per week. The nature of his/her work calls for more than eight (8) hours of work on some days and less than eight (8) hours of work on others. Therefore, it is agreed that the above-named nurse shall be exempt from the provisions of Article 8, Section 8.01 and shall be paid time and one half (1 1/2) for all hours worked in excess of (forty) 40 hours in a work week. This Agreement is subject to review every six (6) months and may be terminated by either party at that time. Signature of Nurse Date Signature of ECMCC CEO/CFO Signature of NYSNA Representative

APPENDIX "H"

MEMORANDUM OF AGREEMENT BETWEEN THE NEW YORK STATE NURSES ASSOCIATION AND

ERIE COUNTY MEDICAL CENTER CORPORATION

Per Diem CRNA Agreement

This Memorandum of Agreement modifies the collective bargaining agreement between the parties dated January 1, 2023 — December 31, 2027. Except as modified herein, the collective bargaining agreement remains in full force and effect.

The following constitutes an agreement between the NYSNA and the ECMC concerning per diem CRNAs.

- Persons appointed to Per Diem CRNA positions will be employees of ECMC and members of the NYSNA bargaining unit. They shall be utilized to supplement the CRNA staff to assure an adequate and efficient means to provide appropriate professional services.
- 2. It is understood that these employees will have the right to refuse to work if called. However, per diem CRNAs hired after January 1, 2017 must be available to work at least two (2) shifts in each four (4) week schedule period, one of which must be a twelve (12) hour weekend shift. Per diem CRNAs hired prior to January 1, 2017 must be available to work at least one (1) shift in each four (4) week schedule period, if requested by the employer.
- 3. Per Diem CRNAs will have no guaranteed hours of work and when scheduled will be subject to being cancelled up until one (1) hour before the beginning of their scheduled shift. If a CRNA is cancelled with less than one (1) hour notice he/she shall receive a minimum of three (3) hours pay.
- 4. Each per diem CRNA will be expected to be scheduled to work one (1) shift on one (1) major holiday per year. Major holidays are designated by the ECMC as:

New Year's Day Memorial Day Independence Day Labor Day Thanksgiving Day Christmas Day

- 5. A Per Diem CRNA may be scheduled to work the hours of an employee on a leave of absence for up to three (3) months. This time may be extended by mutual agreement to six (6) months.
- 6. Per Diem CRNAs will not be utilized in such a manner as to avoid filling full-time, regular part-time or part-time positions.
- 7. All requests are to be made in writing on the forms available in the CRNA office. The posted schedule request deadlines will be adhered to.

- 8. Per Diem CRNAs will have no benefits set forth in the Collective Bargaining Agreement between the Erie County Medical Center and New York State Nurses Association except as specifically set forth as follows:
 - i. Seniority -- one-half (1/2) month for each month during which at least one (1) shift is worked.

ii. Job posting.

- Payment at the rate of his/her regular straight time pay plus one and one-half (1 1/2) times his/her straight time hourly rate for every hour actually worked on the holidays of New Year's Day, Thanksgiving, and Christmas.
- iv. Access to the grievance procedure. The just cause standard applies as set forth in Section 14.01 of the Collective Bargaining Agreement.
- v. A one half (1/2) hour paid lunch break.
- 9. If an incumbent FT/RPT/PT employee changes his/her status to Per Diem said employee shall receive payment for any and all accrued benefit time (Vacation, Compensatory, etc.), except personal and sick time at the time of the status change. However, accrued sick time shall remain in the bank in accordance with Section 11.07 Sick Leave. Said employee may not utilize any sick time in his/her bank while the employee remains a Per Diem employee. Such Per Diem employees may utilize accrued sick time if the employee returns to FT/RPT status within ten (10) years of his/her switch to Per Diem status.

This Subsection (9) does not apply to current retirees of Erie County who may be working as a Per Diem CRNA.

Per Diem CRNAs shall be paid an hourly rate of fifteen percent (15%) above base qualifying salary rate as determined based on years of experience.

- 10. Orientation to ECMC policy and procedures shall be provided prior to CRNA performing any clinical services. Each per diem CRNA shall complete a skills assessment to verify competencies prior to being assigned independently. All per diem CRNAs shall be credentialed by Medical/ Dental. Such employee may practice for thirty (30) days while said credentialing is completed.
- 11. Per Diem CRNAs will not be assigned on call responsibilities unless by mutual agreement between the Employer and the Employee. Per diem CRNAs will not be offered on-call until and unless the on-call shift(s) have been first offered to all full-time, regular part-time and part-time CRNAs on the unit.

APPENDIX "I"

MEMORANDUM OF AGREEMENT BETWEEN
THE
NEW YORK STATE NURSES ASSOCIATION AND
COUNTY OF ERIE
AND
ERIE COUNTY MEDICAL CENTER CORPORATION

NURSE ANESTHETIST SHIFT COVERAGE

This Memorandum of Agreement modifies the collective bargaining agreement between the parties dated January 1, 2023 - December 31, 2027. Except as modified herein, the collective bargaining agreement remains in full force and effect.

The following constitutes an agreement between the NYSNA and the ECMC concerning the anesthesia night and weekend coverage by hospital-employed CRNAs:

- 1. Weekly Daytime hours will be designated as: Monday Friday, 7:00 am 7:00 pm, which may be worked in eight (8), ten (10) or twelve (12) hour shifts.
- 2. Night and Weekend hours will be designated as:
 - Monday Friday: 7:00 pm 7:00 am
 - Saturday 7:00 am 7:00 am
 - Sunday: 7:00 am 7:00 am
 - All Major Holidays: 7:00 am 7:00am. (Major Holidays are designated as New Year's Day, Memorial Day, Independence Day, Labor Day, Thanksgiving Day and Christmas Day).
- 3. Night and Weekend assignments may be one of the following:
 - 16 hour SHIFT (3:00 pm 7:00 am)-(upon staff request mutual agreement)
 - 12 hour SHIFT (7:00 am 7:00 pm or 7:00 pm 7:00 am)
 - 4 hour to 10 hour SHIFTS (hours to be scheduled according to departmental needs and to permit mutual agreement by CRNAs to switch and or cover for each other).
 - Overtime is paid for all such hours worked in excess of (40) forty hours per week, regardless of the 8/10/12/16 hour shifts.
- 4. All CRNAs employed by the ECMC are required to work nights and weekends, and will be scheduled any combination of the above assignments to meet the staffing requirements of the Department of Anesthesia, as determined by the VP of Surgical Services, or his/her designee. New CRNAs hired by the Department will be scheduled for night and weekend hours when approved by the VP of Surgical Services. Such assignments shall be equitably distributed. No CRNA will be scheduled to rotate from a day shift assignment to a night shift assignment or vice versa with less than twelve (12) hours of time off between work assignments unless mutually agreed.
- 5. The work schedule will be made out by the Chief CRNA and approved by the VP of Surgical Services. The schedule will be of one month's duration, and will be posted three (3) weeks prior to the start of the schedule. The number of shifts per pay period will be based on staffing needs and staff requests.
- 6. An on-call room will be available for the CRNA to use while assigned the overnight shift.

APPENDIX "J"

MEMORANDUM OF AGREEMENT
BETWEEN THE
NEW YORK STATE NURSES ASSOCIATION
AND
COUNTY OF ERIE
AND
ERIE COUNTY MEDICAL CENTER CORPORATION

CRNA Coverage for Saturday Trauma Call

This Memorandum of Agreement replaces the previous MOAs between the parties on the subject of on-call coverage.

This Memorandum of Agreement modifies the collective bargaining agreement (CBA) between the parties dated January 1, 2023 - December 31, 2027. Except as modified herein, the collective bargaining agreement remains in full force and effect.

- 1. The following constitutes an agreement between the New York State Nurses Association and the Erie County Medical Center covering assignment of on-call and compensation for such "on-call" CRNA's. This shall remain in effect as a pilot until a new collective bargaining agreement is voted on by the membership. If this pilot is to become permanent and if any changes are made to this pilot this will be subject to the further ratification by the CRNAs.
- 2. Coverage for Trauma On Call on Saturdays will be one (1) CRNA from 7:00 a.m. 3:00 p.m.
- 3. Employees assigned to be on-call receive compensation at the rate of one (1) hour of straight time pay for every four (4) hours of on-call and for each hour or so assigned.
- 4. In addition to the on-call pay, if an employee in this rotation schedule is called in, he/she shall receive a minimum of four (4) hours call in pay, and all such hours will be paid at twice (double time) the applicable base rate regardless of absences prior to the call in. An employee who is called in within three (3) hours or less prior to the beginning of his/her shift will be paid for actual hours worked prior to his/her regularly scheduled shift.
- 5. This call will be covered by all CRNAs in a fair and equitable manner.
 - No CRNA will be assigned to trauma call following his/her scheduled 7:00 p.m. 7:00 a.m. shift.
 - RPT CRNAs will be assigned one half (1/2) the number of on-call shifts as a FT employee.
- 6. The trauma on-call schedule will be prepared by the CRNA staff following receipt of the planned monthly work schedule. CRNA staff will complete the on-call assignments at least three (3) days in advance of the final approval and posting of both schedules. Assignments will be evenly distributed amongst staff participating in the call requirement. The final trauma call schedule will be reviewed and approved by the Chief CRNA or surgical service management before posting.
- 7. Any scheduled changes between peers can be made via the Chief CRNA. Any unforeseen

- occurrence, i.e., sudden illness, will be dealt with by the charge Anesthesiologist.
- 8. Employees will be issued beepers. If no response is made to a page within fifteen (15) minutes, the employee will be called at home. A list of current contact numbers will be kept at the switchboard.
- 9. Except in emergency situations, the hospital will give employees a minimum of two (2) weeks' notice if it should be necessary to change the on-call schedule.

APPENDIX "K"

MEMORANDUM OF AGREEMENT BETWEEN

THE NEW YORK STATE NURSES ASSOCIATION AND ERIE COUNTY MEDICAL CENTER CORPORATION

CRNA Wage and Benefit Adjustment

This Memorandum of Agreement modifies the collective bargaining agreement (CBA) between the parties dated January 1, 2023 – December 31, 2027. It addresses specific items. All other benefits shall remain the same as provided in the collective bargaining agreement, as amended.

There is a need to recruit and retain Certified Registered Nurse Anesthetist ("CRNA") staff to maintain staffing and anesthesia coverage for the operating room.

<u>Wages: Schedule A</u> - the CRNA wage scale will no longer be listed in Appendix A; instead, reference shall be made to this MOA for wages and increments.

Current:

Effective January 1, 2021, increase wage rates by 3% as follows:

New employees with 0 calendar months of full-time CRNA experience through the completion of 60 calendar months of full-time CRNA experience will receive \$195,700.00 annually as starting wage.

The hiring and next step for experienced candidates after completion of 60 calendar months of full-time CRNA experience will be \$211,150.00 annually.

All active Full-Time and Regular Part-Time CRNA's will receive a one thousand two hundred dollar (\$1200) lump sum bonus, to be paid within thirty (30) days of ratification.

Effective June 1, 2022, increase wage rates by 8% as follows:

New employees with 0 calendar months of full-time CRNA experience through the completion of 60 calendar months of full-time CRNA experience will receive \$211,356.00 annually as starting wage.

The hiring and next step for experienced candidates after completion of 60 calendar months of full-time CRNA experience will be \$228,042.00 annually.

Effective January 1, 2023, increase wage rates by 3.5% as follows:

New employees with 0 calendar months of full-time CRNA experience through the completion of 60 calendar months of full-time CRNA experience will receive \$218,753.46 annually as starting wage.

The hiring and next step for experienced candidates after completion of 60 calendar months of full-time CRNA experience will be \$236,023.47 annually.

Effective January 1, 2024 increase wage rates by 3.5% as follows:

New employees with 0 calendar months of full-time CRNA experience through the completion of 60 calendar months of full-time CRNA experience will receive \$226,409.83 annually as starting wage.

The hiring and next step for experienced candidates after completion of 60 calendar months of full-time CRNA experience will be \$244,284.29 annually.

Only experience working as a CRNA at ECMCC shall constitute CRNA experience for purposes of progression through the compensation and PTO tiers after hire as a CRNA.

Unless otherwise mutually agreed upon, the parties agree to meet to negotiate successor language to the terms and conditions set forth in Appendix H, Appendix I, Appendix J, and Appendix K commencing no later than nine (9) months after this Agreement's last memorialized wage increase.

Overtime: Article 8

All overtime shall be paid at a rate of one and one half (1.5) times for any hours over forty (40) hours in a work week.

Holiday time paid, whether the holiday was worked or not, will be considered hours worked for purposes of overtime calculation.

Shift Differential: Article 10.01

Effective February 13, 2020, a ten dollar (\$10.00) per hour shift differential will be provided to all full-time, regular part-time and part-time CRNAs that work one (1) or more consecutive hours between 3 p.m. and 7 a.m. Monday through Friday for each hour worked during this period.

Effective June 22, 2022, a twenty dollar (\$20.00) per hour shift differential will be provided to all full-time, regular part-time and part-time CRNAs that work one (1) or more consecutive hours between 3 p.m. and 7 a.m. Monday through Friday for each hour worked during this period.

Effective February 13, 2020, a ten dollar (\$10.00) per hour shift differential will be provided to all full-time, regular part-time, and part-time CRNAs that work four (4) or more consecutive hours on a Saturday, Sunday (includes all hours during the shift starting Sunday at 7:00 p.m. to Monday at 7:00 a.m.), or a Holiday recognized under Appendix K for each hour worked during these days.

Effective June 22, 2022, a twenty dollar (\$20.00) per hour weekend/ holiday differential will be provided to all full-time, regular part-time, and part-time CRNAs that work four (4) or more consecutive hours on a Saturday, Sunday (includes all hours during the shift starting Sunday at 7:00p.m. to Monday at 7:00 a.m.), or a Holiday recognized under Appendix K for each hour worked during these days.

The parties agree that the shift differential under Article 10.01 shall not be applicable to full-time, regular part-time, and part-time CRNAs. The parties agree that the shift differential under Article 10.01 shall be applicable to per diem CRNAs.

Holidays: Article 11.02 - shall be modified as below:

Minor holidays worked will be paid at the employee's straight time rate of pay (5 Holidays: Martin Luther King Day; President's Day; Easter Sunday; Veteran's Day, and Juneteenth).

Major Holidays worked will be paid at regular straight time pay plus one and one half (1.5) times the employee's hourly rate of pay (6 Holidays: New Year's Day; Memorial Day; Independence Day; Labor Day; Thanksgiving Day; Christmas Day).

If one of the Major Holidays identified in paragraph B falls on the employee's scheduled day off, such employee will receive eight (8) hours pay at the employee's straight time rate of pay.

The employer will assign holidays equitably to the extent possible.

Holiday pay will be paid according to the day the holiday traditionally falls and not the day observed by the Employer.

Non Salary Benefits:

Education Reimbursement and Professional Fees

Each, full time CRNA will receive reimbursement up to four thousand dollars (\$4,000) for educational expenses for programs that provide approved AANA CEU credit and professional fee reimbursement pertaining to only the year in which the reimbursement is requested (i.e., Medical/Dental Staff annual dues/reappointment fee, Register Nurse license renewal fee, ACLS certification, and AANA membership fees) per year, non-cumulative. Any CPC examination fee shall be eligible for reimbursement. Education expenses must be pre-approved by the Senior Vice President of Surgical and Outpatient Services or his designee. Requests for reimbursement for payment of annual dues, reappointment fees, or AANA fees must be submitted on the requisite forms with supporting documentation. Any expenses that provide for a material benefit to a CRNA that would qualify as a gift of public funds under the law will not be reimbursed. This amount shall be prorated for Regular Part Time CRNAs.

ECMC will provide funds for the educational/dues/fees expense account. Employees will follow the established procedures for submission of supportive details and documentation for prior approval and post-course payment. Any request for travel must be pre-approved and in accordance with the ECMCC Policy on Travel. Travel will be limited to travel within the continental United States.

After submission of the required documentation, reimbursement will be within thirty (30) days.

Health and Dental Benefits

Health and dental insurance benefits shall be provided to CRNAs as agreed to under the terms of Article 9 of the Collective Bargaining Agreement (CBA).

PAID TIME OFF (PTO)

Effective January 1, 2009, CRNAs will convert to a Benefit Time: Paid Time Off – (A Bank for Compensatory, Vacation, Short Term Sick Time and Personal Leave).

Up to four (4) CRNAs will be allowed off on PTO at one time, except for the time period between the first full week of July and the last full week of August. During the period between the first full week of July and the last full week of August, up to three (3) CRNAs will be allowed off at one time.

CRNAs shall be allowed to select two (2) full weeks of PTO by seniority during the initial PTO selection pass. After the initial pass, CRNAs will be able to select accrued weeks of PTO as of date of utilization, by seniority until all full PTO weeks are selected.

Effective upon ratification of the agreement, PTO for each CRNA shall be accrued based on years of experience as a CRNA as follows:

<u>Years OF</u> <u>EXPERIENCE AS</u> <u>CRNA</u>	PTO Accrual FT	Total Days FT	Max Hours Annual Accrual's
0 calendar months of full-time CRNA experience through the completion of 60 calendar months of full-time CRNA experience	10.15	33	264
completion of 60 calendar months of full-time CRNA experience through the completion of 120 calendar months of full-time CRNA experience	11.69	38	304
After completion of 120 calendar months of full-time CRNA experience	12.62	41	328

<u>Years OF</u>	PTO	Total	Max Hours
<u>EXPERIENCE AS</u>	Accrual	Days	Annual
<u>CRNA</u>	RPT	RPT	Accrual's
0 calendar months of full-time CRNA experience through the completion of 60 calendar months of full-time CRNA	5.08	16.5	132

experience	Annual St. Million (n) principal (n) registration and recommendation of the second	general and and benefit and Windowski to an arrange of Pri	and at all and a set to the second
completion of 60 calendar months of full-time CRNA experience through the completion of 120 calendar months of full-time CRNA experience	5.85	19	152
After completion of 120 calendar months of full-time CRNA experience	6.31	20.5	164

PTO is available for use subject to the approval of the supervisor. All employees are responsible for maintaining their PTO bank.

The maximum PTO bank balance is one and one half (1.5) times the annual accrual. If the PTO maximum is reached, no additional PTO will accrue until the balance is reduced by using PTO and/or utilizing the PTO sell feature, a copy of which is attached hereto as attachment "A." One time per year, employees may elect to "sell" up to eighty (80) hours of vacation leave or PTO.

CRNAs will no longer have a compensatory time option as provided under any terms of the CBA.

CRNAs will no longer have personal leave days as provided in Article 11.03 of the CBA. CRNAs will not be eligible to participate in clinical ladders.

Sick leave accruals as provided in Article 11.07 of the CBA will cease effective January 1, 2009. Effective February 13, 2020, all permanent full-time CRNAs shall be provided three (3) sick days. Effective January 1, 2020, all permanent full-time CRNAs shall earn sick leave credits at the annual rate of five (5) sick days. All permanent regular part-time CRNAs shall earn 50% of the sick leave credits of the permanent full-time CRNAs.

Reasons for Granting Sick Leave

Sick leave with pay shall be granted to ECMCC to a CRNA, when incapacitated or unable to perform the duties of his/her position by reason of:

A. Sickness or injury;

B. Serious illness in the CNRA's immediate family, requiring care and attendance of the CRNA. Immediate family shall include parent, spouse, brother, sister, children or grandparents; or other blood relative who is an actual member of the CRNA's household. Certification of affidavit, issued by the attending physician certified to the necessity for the attendance of the CRNA may be required by the department head if there is reason to suspect abuse of this Section. All requests for sick leave for serious family illness must be requested in writing and such request, including a doctor's affidavit if required, shall be filed with the CRNA's department head or designee and forwarded to payroll; sick leave for this purpose shall be granted only with the approval of the appointing authority;

- C. Quarantine regulations;
- D. Emergency medical or dental visits; and
- E. Maternity.

Abuse of Sick Leave Benefits

Abuse of sick leave privileges shall be grounds for disciplinary action. In addition, where an employee's absences are such that the Employer has reasonable grounds to believe that an abuse of sick leave may exist, such employee, where practicable, will be notified of such suspected abuse and thereafter may be required, regardless of the duration of the absence, to submit a satisfactory doctor's certificate or affidavit indicating the specific nature of the disability and its duration to his/her department head or designee and forwarded to payroll before such abuse may be charged against the employee's accumulated sick leave balance. The Association will work cooperatively with the Employer to reduce and prevent abuses of sick leave.

Reporting Time

- A. In case of absences for the day shift, the time for reporting absences shall not be less than two (2) hours prior to the actual start time for the shift. In the case of reporting absences for the night (7:00 p.m. to 7:00 a.m.) the time for reporting absences shall not be less than three (3) hours prior to the actual start time of the shift. If cancelling a voluntary overtime shift on the day shift, a CRNA must call in a minimum of two (2) hours prior to the start of the voluntary shift. If cancelling a voluntary overtime shift on the night shift, a CRNA must call in a minimum of three (3) hours prior to the start of the voluntary shift. Failure to call-in by these time periods prior to the start of the voluntary shift will be counted as an unapproved occurrence. In case of failure to report within the stated time limits, unless for reasons satisfactory to the employee's department head or his/her designee, the absence shall not be deductible from sick leave and shall be considered as time off without pay.
- B. A certificate or affidavit, showing incapacity and inability of the employee to perform his/her duties issued by the attending physician, shall be filed with the department head or designee and forwarded to payroll in case of absence of more than three (3) consecutive workdays. The Employer may check further on any illness regardless of certificate or affidavit.
- C. If an employee fails to submit proof of illness to his/her department head or designee when required to do so, the absence shall not be deductible from sick leave and shall be considered as time off without pay.
- D. If the proof submitted, in the judgement of the Employer, does not justify the employee's absence, such absence shall not be deducted from sick leave and shall be considered time off without pay.

Resignation or Retirement

Sick time taken after notice of resignation or retirement shall require a doctor's certificate or an equivalent amount of vacation time will be deducted. CRNAs will continue to be eligible for long term sick leave as provided in Article 11.07 of the CBA.

All current employees' accrued benefit time (i.e., vacation time, compensatory time, personal leave, and sick leave) will be frozen effective January 1, 2009. This time can be used for future needs. Such accrued leave will be used as PTO, but shall be scheduled in advance subject to the approval of the

supervisor.

A total of sixteen (16) hours of each individual employee's accrued PTO may be used for emergency purposes in each calendar year. Use of emergency PTO will not be subject to accrual of occurrences under the Time and Attendance Policy.

In order for the department head and/or the immediate supervisor to arrange for adequate work coverage, applications for emergency PTO must be filed by an employee on a prescribed form with the head of the department or immediate supervisor at three (3) working days in advance. Every effort will be made by the Employer to find a replacement before denial of such request. If, after every effort has been made, including offering the requested shift to all other CRNAs as overtime, extra time, and/or offering the shift to a Per Diem CRNA, the request would place the CRNA staffing level below minimum staffing, the request may be denied. All requests must receive the approval of the employee's immediate supervisor or department head and shall not be granted in less than one (1) hour units. In cases of personal emergency, when the above time limits have expired, the Employer will make an effort to find a replacement before invoking staffing needs as a basis for denial.

ECMC form number 621014 will be used to document all PTO usage.

ECMCC

PTO/Vacation Sell Request

_									
O	١,	Δ	•	١,	E	Ω	٠,	v	
v	γ	•	1	¥	Ľ	♥		¥	٠

Effective October _____, 2008, CRNA employees may request to sell up to forty (40.0) hours of accrued PTO and/or Vacation (Paid Time Off) one (1) time per year. The purpose of this sell provision is in addition to the expectation that employees manage their leave time banks by taking an appropriate amount of PTO/Vacation time off annually for their personal well being and the good of the organization.

Process:

- > To be eligible, current combined PTO & Vacation balance must be greater than eighty (80) hours for consideration of request.
- Complete & submit the Request form to Human Resources by October. Note: In addition to the payout, employees' may elect to defer part of the payout proceeds into the 457b Deferred Compensation Plan with AIG/VALIC.
- Request is subject to the Senior Vice President and the Vice President of Human Resources review and approval.
- Upon approval, payment will be processed on the first regularly scheduled pay date in November.

Dept:
of hours (Maximum- 40.0 hours)
Vacation bank PTO Bank
or% from PTO/Vacation sell proceeds
Date
Vice President – Human Resources

APPENDIX "L"

MEMORANDUM OFAGREEMENT BETWEEN THE NEW YORK STATE NURSES ASSOCIATION And ERIE COUNTY MEDICAL CENTER CORPORATION

(Per Diem Registered Nurse Agreement)

This Memorandum of Agreement modifies the collective bargaining agreement between the parties dated January 1, 2023— December 31, 2027. Except as modified herein, the collective bargaining agreement remains in full force and effect.

- The title Registered Nurse ECMC (Per Diem) will be utilized for use in supplementing the staff and insuring an adequate and efficient means of supplementing staff in providing professional services.
- 2. Persons appointed to these positions will be employees of ECMC and members of the NYSNA bargaining unit.
- 3. Per Diem registered nurses will not be assigned "on-call" responsibilities unless by mutual agreement between the Employer and the Employee. Per diem registered nurses will not be offered "on-call" until and unless the on-call shift(s) have first been offered to all full-time, regular part-time and part-time nurses on the unit.
- 4. Per Diem RNs will have no guaranteed hours of work and when scheduled will be subject to being cancelled up until one (1) hour before the beginning of their scheduled shift. A Per Diem may be scheduled to work the hours of an employee on a leave of absence for up to three (3) months. This time may be extended by mutual agreement to six (6) months. Per Diem employees will not be utilized in such a manner as to continually avoid filling full-time, regular part-time or part-time positions.
- 5. It is understood that these employees will have the right to refuse to work if called. However, if contacted, an employee must be available to work at least one (1) weekend every two (2) months and one (1) holiday offered. Furthermore, ECMC will not be required to contact persons in any pre-setorder. Employees who fail to meet the availability requirements will receive a written notice (non-disciplinary) for the first occurrence; the second such occurrence the employee shall be considered to have voluntarily resigned.
- 6. Per Diem registered nurses will receive an orientation and will have their competencies validated.
- 7. Effective June 22, 2022 the rate will be increased to fifty (\$50.00) dollars per hour.
- 8. Per Diem registered nurses will have no benefits set forth in the Collective Bargaining Agreement between the Erie County Medical Center and New York State Nurses Association except as specifically set forth as follows:

- a. Seniority -- one-half (1/2) month for each month during which at least one (1) shift is worked.
- b. Job posting.
- c. Payment at the rate of his/her regular straight time pay plus one and one-half (1 ½) times his/her straight time hourly rate for every hour actually worked on the holidays of New Year's Day, Thanksgiving, and Christmas.
- d. Access to the grievance procedure. The just cause standard applies as set forth in Section 14.01 of the Collective Bargaining Agreement.
- e. Per Diem registered nurses may be assigned as Charge Nurse in accordance with Section 7.06 provided they are qualified and have previous charge nurse experience.
- f. Per Diem registered nurses may be assigned as preceptor and receive the preceptor differential (Section 10.03) provided they have attended the preceptor educational training provided and can work the hours of the orientee they are assigned.
- g. A one-half (1/2) hour paid lunch break.
- h. Shift Differential in accordance with Section 10.01.
- 9. If an incumbent FT/RPT/PT employee changes his/her status to Per Diem said employee shall receive payment for any and all accrued benefit time (Holiday, Vacation, Compensatory, etc.), except personal and sick time at the time of the status change. However, accrued sick time shall remain in the bank in accordance with Section 11.07 <u>Sick Leave</u>. Said employee may not utilize any sick time in his/her bank while the employee remains a Per Diem employee. Such Per Diem employees may utilize accrued sick time if the employee returns to FT/RPT status within ten (10) years of his/her switch to Per Diem status.

APPENDIX "M"

MEMORANDUM OF AGREEMENT BETWEEN
THE
NEW YORK STATE NURSES ASSOCIATION AND
COUNTY OF ERIE
And
ERIE COUNTY MEDICAL CENTER CORPORATION

(Agency Registered Nurse Program)

This Memorandum of Agreement modifies the collective bargaining agreement between the parties dated January 1, 2023 - December 31, 2027. Except as modified herein, the collective bargaining agreement remains in full force and effect.

Every effort shall be made by the Employer to adequately staff nursing units before resorting to the use of Agency nurses.

Agency nurses shall have sufficient Hospital/Nursing Home orientation to the unit to which they are assigned to meet minimum standards/requirements of that unit.

The use of Agency nurses shall in no way disadvantage bargaining unit nurses including, but not limited to, resulting in Hospital/Nursing Home-employed nurses working extra weekends or holidays that the Hospital/Nursing Home nurses to lose the opportunity to work extra weekends, holidays and/or overtime should they desire to do so. Regular Hospital and Nursing Home staff may sign up for extra shifts as per current practice. Nursing managers will post a needs list for vacant shifts two (2) weeks before the date the schedule is posted. Regular Hospital and Nursing Home staff will be able to sign up for vacancies for the first week. However, if no regular employee volunteers for an open shift or shifts during this one week time frame, remaining open shifts may be assigned to an agency registered nurse. Once an agency registered nurse is scheduled to work such shift or shifts the agency nurse will not be displaced from the shift or shifts by any regular employee seeking to work the same shift or shifts.

The use of Agency nurses shall not in any way displace a Hospital/Nursing Home nurse or result in a decrease in hours of Hospital/Nursing Home nurses who wish to work, nor shall the use of Agency nurses result in a Hospital/Nursing Home nurse being floated out of his/her regularly assigned unit while an Agency nurse is assigned to work on the covered employee's assigned unit.

The parties will review utilization and discuss the Agency Registered Nurse Program in the Labor Management Committee.

APPENDIX "N"

MEMORANDUM OF AGREEMENT

BETWEEN THE

NEW YORK STATE NURSES ASSOCIATION

AND

ERIE COUNTY MEDICAL CENTER CORPORATION

Effective August 30, 2019

(Registered Nurse On-Call Coverage - SANE)

This Memorandum of Agreement modifies the collective bargaining agreement between the parties dated January 1, 2023 – December 31, 2027. Except as modified herein, the collective bargaining agreement remains in full force and effect.

- 1. The following constitutes an agreement between the New York State Nurses Association ("NYSNA") and the Erie County Medical Center Corporation ("ECMCC") covering compensation for "on-call Sexual Assault Nurse Examiners (SANE) nurses."
- 2. Employees with SANE credentialing, with a signed certificate of completion signed by the SANE Coordinator, who are assigned to be on-call for the care of victims alleging sexual assault shall receive compensation as follows:
 - a. There will be an on-call schedule for coverage, as needed, to cover the care needs of victims alleging sexual assault. On-call hours Monday-Friday will be 3:00 p.m. to 7:00 a.m. and other times as determined by the Manager/designee with notification to NYSNA. Saturday-Sunday on-call hours will be Saturday 7:00 a.m. to Sunday 7:00 a.m. and Sunday 7:00 a.m. to Monday 7:00 a.m. Holiday on-call hours will be 7:00 a.m. on the day of the holiday, until 7:00 a.m. the following day. These on-call hours shall be compensated at the rate of one (1) hour of straight time pay for every three (3) hours of on-call and for each hour so assigned. The employee will not be required to provide care to other Emergency Department (ED) patients that are not alleged victims of sexual assault. In addition to the on-call pay, if an employee in this rotation schedule is called in, he/she shall receive a minimum of four (4) hours call in pay and all such hours will be paid at double time (2x) regardless of absences prior to the call-in.

After being called in and working a minimum of two (2) hours within eight hours prior to the employee's scheduled shift the employee may receive rest pay and accumulated leave time, by utilizing the option below:

i. Receive up to four (4) hours of call-in rest pay and can elect to use up to four (4) hours of accumulated leave time to ensure that the employee has eight (8) hours off in between call activation end and the scheduled shift.

- ii. In times where the call-in period overlaps the scheduled shift, work one half (½) the hours of the scheduled shift after the call in activation work period ends and thereafter be released from work and receive up to one half (½) the duration of their scheduled shift in call-in rest pay.
- 3. The on-call schedule will be made out by the Nursing Care Coordinator (Emergency Department)/designee. Prior to the final posting of the schedule, if an insufficient number of employees volunteer to provide coverage, a nurse will be assigned. All efforts will be made to avoid scheduling on-call assignments prior to the start of the employee's regularly scheduled shift.

29

- 4. All SANE credentialed registered nurses assigned to the ED will be required to take call. This Agreement does not preclude other SANE credentialed medical staff who are contracted through ECMCC from taking call. The SANE on-call will be distributed equitably; therefore, every SANE credentialed registered nurse will have an on-call assignment before beginning the cycle again. Employees may volunteer for more on- call assignments, if they so desire.
- 5. Any schedule changes between peers can be made via the Nursing Care Coordinator (Emergency Department). Any unforeseen occurrence, i.e., sudden illness, will be dealt with via the Nursing Office.
- 6. The on-call nurse will be called by the ED Unit Manager or Charge Nurse/designee. Employees will be issued pagers and will report to the ED within sixty (60) minutes of receiving the page. If no response is made to a page within fifteen (15) minutes, the Nursing Care Coordinator (Emergency Department) or Charge Nurse/designee will call the employee on the employee's preferred and designated phone line. The employee is required to provide the Nursing Office with a current telephone number.
- 7. Except in emergency situations, the Hospital will give employees a minimum of two (2) weeks' notice should it be necessary to change the on-call schedule.
- 8. The SANE nurse must submit appropriate overtime forms to the Nursing Care Coordinator (Emergency Department) to insure prompt payment.
- 9. On-call assignments will be posted in the Nursing Office and ED with the work schedule.
- 10. It is the assigned employee's responsibility to furnish an on-call coverage substitute should the assigned employee be unable to fulfill his/her commitment. This change in assignment will be communicated to the Nursing Office at the earliest possible time. Failure to respond to on-call will result in forfeiture of on-call compensation and may result in disciplinary action.

APPENDIX "0"

MEMORANDUM OF AGREEMENT BETWEEN THE NEW YORK STATENURSES ASSOCIATION AND ERIE COUNTY MEDICAL CENTER CORPORATION

(Adjunct Educator Pay)

This Memorandum of Agreement modifies the collective bargaining agreement between the parties dated January 1, 2023 – December 31, 2027. Except as modified herein, the collective bargaining agreement remains in full force and effect.

Adjunct Educator Pay

The following constitutes an agreement between the New York State Nurses Association (NYSNA) and Erie County Medical Center Corporation (ECMCC) concerning the Adjunct Educator Program. The Adjunct Educator program is a partnership with colleges and universities and the Erie County Medical Center Corporation to enhance student nurse education by providing didactic and/or clinical support and staff Registered Nurse preceptors within the facility.

- 1. In order to participate in ECMCC's Adjunct Educator Program for the DEU program, an RN must have successfully completed ECMCC's preceptor training program. In addition, he/she will have completed the orientation provided by the college or university and the expectations of that particular school of nursing regarding programming and accountabilities. Attendance at any required in-service pertaining to this program shall be on paid work time.
- 2. ECMCC will determine the number of participants in the DEU program dependent upon the number of students and colleges and/or universities participating. Each staff registered nurse participating in the Adjunct Educator Program will be assigned a maximum of two (2) students.
- ECMCC will select from among RN volunteers for the DEU and Capstone program. Selection for the program will be based upon competency, communication skills, good standing and shift-specific need.
- 4. Staff RNs assigned as adjunct educators will be paid five dollars (\$5.00) per hour ("adjunct educator pay") for all hours worked related to the provision of adjunct education, training and activities, as assigned by the Unit Manager.
- 5. Staff RNs assigned students under the Capstone program will be paid three dollars (\$3.00) per hour ("Capstone educator pay") for all hours worked related to the provision of adjunct education, training and activities as assigned by the Unit Manager.
- 6. RNs who do not complete the education, training and activities required by the program will not receive adjunct educator pay. Staff RNs serving as adjunct educators will complete student evaluations and confer as needed with the school of nursing faculty regarding the student nurse's progress.
- 7. The program will be unit based. (DEU only)
- 8. Once selected, participation in the program will rotate by seniority within the participating unit(s).

programs.	-		*
programs.			
	άλ		

ECMCC will provide clearly defined roles and expectations for both the DEU and Capstone

9.

APPENDIX "P"

MEMORANDUM OF AGREEMENT BETWEEN THE NEW YORKSTATE NURSES ASSOCIATION AND COUNTY OF ERIE And ERIE COUNTY MEDICAL CENTER CORPORATION

(NYSNA Vacation Accruals)

This Memorandum of Agreement modifies the collective bargaining agreement between the parties dated January 1, 2023 - December 31, 2027. Except as modified herein, the collective bargaining agreement remains in full force and effect.

The employer, ECMCC, and union, NYSNA, reserve the right to review employees who come forward claiming they have not been able to schedule their vacation time and thus will begin to lose accruals. In these cases the parties agree that should the vacation time have been denied at no fault of the employee (timely requests during the appropriate periods), the parties may extend the time in which the employee may use up to forty (40) hours of time for up to ninety (90) days from the date the maximum accrual is reached or allow a buyback of the appropriate amount of hours, up to forty (40), at the employer's discretion. Employees to whom this applies would be credited back for any missed accruals during the extension period. This agreement does not apply to employees who are out on medical leave of absence.

APPENDIX "Q"

MEMORANDUM OF AGREEMENT BETWEEN THE NEW YORK STATE NURSES ASSOCIATION AND ERIE COUNTY MEDICAL CENTER CORPORATION

Clinical Ladder Program

ECMCC will establish a clinical ladder program, additional details of the program will be supplied in the policy on clinical ladder. Some elements include:

The Clinical Ladder Pathways includes clinical, educational, and leadership pathways.

For levels I and II, the pathway is defined as clinical. Tracks for the transition to advanced clinical, educational or leadership are developed for levels III, IV, and V.

Core requirements for Clinical Ladder Pathway are:

- Registered Nurse working in a bargaining unit position.
- 2. Formal written request to enter/advance into clinical ladder pathway.
- 3. Review Committee approval.

All RN staff that meet the core requirements may apply to the clinical ladder program. RN candidates must have successfully completed the initial six (6) month probationary period and have two years clinical nursing experience.

An RN may change levels once every two (2) years. Two-thousand eighty (2080) hours is equivalent to one year of nursing experience.

Eligible RNs may not have any disciplinary action at the time of the initial appointment to applicable clinical ladder rung. Review of any future disciplines will be reviewed by the Committee at the time of reappointment/application to the next rung.

A Review Committee comprised of six (6) members shall review the application requirements and application process with each application round. Applications will be accepted a minimum of two times per year. The same committee shall meet at least twice per round to review all applications. A quorum of each group of representatives (three each from the bargaining unit and clinical administration) must be present at all meetings. The committee will be jointly chaired by a bargaining unit representative and clinical administration representative.

The six (6) members will be:

2 Representatives from clinical administration

1HR Representative

3 NYSNA members

Additionally:

- 1 Sr. VP of Nursing or designee, in case of tie-breaker to be alternated with the NYSNA LBU Chairperson or designee
- 1 NYSNA Representative Alternate
- 1 ECMCC Representative Alternate

The financial incentive for those who participate in the clinical ladder are the following annual amounts (to be paid by pay period):

Level III \$2100 Level IV \$2500 Level V \$3000

Additional information is provided in the policy for clinical ladder and on the ECMCC intranet.

The parties agree to meet to discuss the Clinical Ladder provisions for Nurse Practitioner titles within thirty (30) days of ratification [July 17, 2023] of this Agreement.

APPENDIX "R"

Last Name	First Name
Adams	Gloria
Ball	Diane
Kuebler	Nancy
Lakso	Madonna
Oddo	Donna
Roland	Lynnette
Zoltak	Lisa

Eligibility is determined by Article 9.01 <u>Health Insurance</u> #8.

APPENDIX "S"

MEMORANDUM OF AGREEMENT
BETWEEN THE
NEW YORK NURSES ASSOCIATION
AND
ERIE COUNTY MEDICAL CENTER CORPORATION
Effective August 30, 2019

Nurse Staffing Committee

This Memorandum of Agreement modifies the collective bargaining agreement between the parties dated January 1, 2023 - December 31, 2027. Except as modified herein, the collective bargaining agreement remains in full force and effect.

- 1. ECMCC agrees to commit to post and fill forty-five (45) permanent bargaining unit positions (up to a cost of four and a half million dollars [\$4,500,000.00]) in addition to current existing and posted positions on August 30, 2019. This fund must cover all costs of wages and benefits associated with the additional positions. This allocation will not include staffing for units that are expanding or changing their intended use.
- A Nurse Staffing Committee shall be created composed of seven (7) NYSNA member employees and NYSNA staff representation and seven (7) ECMCC employees selected by the Chief Executive Officer of ECMCC.
- 3. Within ten (10) calendar days after August 30, 2019 the Nurse Staffing Committee shall meet to discuss unit designation and shift of additional positions funded pursuant to provision 1 of this Agreement. The Nurse Staffing Committee shall continue to meet and will have sixty (60) calendar days from August 30, 2019 to determine unit designation and shift placement of the additional positions. This timeframe may be extended by mutual agreement of ECMCC and NYSNA in writing.
- 4. If the Nurse Staffing Committee jointly agrees upon unit designation and shift placement of any additional positions funded pursuant to provision 1 of this Agreement, the position will be posted within ten (10) calendar days of the joint unit designation and shift placement decision and filled in accordance with Article 4.07 Job Posting of the collective bargaining agreement between the parties.
- 5. If the Nurse Staffing Committee cannot jointly agree upon unit designation and shift placement of these additional positions by the 60th day following August 30, 2019 and no written extension agreement has been executed, then the seven (7) NYSNA members and NYSNA staff shall provide a single report detailing the recommended unit designations and shift placements and the seven (7) member ECMCC employees appointed by the Chief Executive Officer shall provide a single report detailing the recommended unit designations and shift placements to the Senior Vice President of Nursing or successor position. The Senior Vice President of Nursing or successor position shall review and take into account both reports to determine the unit designation and shift placement of these additional positions. The final decision will not be arbitrary and capricious. The final decision will be limited to the unit and shift of the new positions and not to nullify the agreement to hire additional RNs.
- 6. The Nurse Staffing Committee shall meet within ten (10) calendar days after August 30, 2019 to discuss minimum numbers of Registered Nurses per shift per staffing unit in all staffing units of ECMCC. The Nurse Staffing Committee shall continue to meet and will have sixty (60) calendar days from August 30, 2019 to establish minimum number of Registered Nurses

per shift per staffing unit in all staffing units of ECMCC. Nursing Supervisors, Unit Managers, and Team Leaders shall be counted as Registered Nurses for the purposes of the minimum number of Registered Nurses per shift per staffing unit only if they have been assigned to provide direct patient care. If the Nurse Staffing Committee cannot jointly agree upon minimum number of Registered Nurses per shift per staffing unit within sixty (60) calendar days from August 30, 2019 and no written extension agreement has been executed, then the seven (7) NYSNA members and NYSNA staff shall provide reports detailing the recommended minimum number of Registered Nurses per shift per staffing unit to the applicable Senior Vice President overseeing the staffing unit. The applicable Senior Vice President shall review and take into account both reports to determine the minimum number of Registered Nurses per shift per staffing unit for all staffing units they oversee. The decision by the applicable Senior Vice President will not be arbitrary and capricious. establishment of the minimum number of Registered Nurses per shift per staffing unit. ECMCC shall in good faith expeditiously post and move to fill any Registered Nurse positions created by establishment of the minimum number of Registered Nurses per shift per staffing unit.

- 7. Upon establishment of initial minimum number of Registered Nurses per shift per staffing unit, the Nurse Staffing Committee shall meet quarterly to discuss amendment to any minimum number of Registered Nurses per shift per staffing unit. If the Nurse Staffing Committee cannot jointly agree upon amendments to the minimum number of Registered Nurses per shift per staffing unit within thirty (30) calendar days of the initial meeting and no written extension agreement has been executed, then the seven (7) NYSNA members and NYSNA staff member shall provide reports detailing the recommended amendments to the minimum number of Registered Nurses per shift per staffing unit to the applicable Senior Vice President overseeing the staffing unit. The applicable Senior Vice President shall review and take into account both reports to determine any amendments to the minimum number of Registered Nurses per shift per staffing unit for all staffing units they oversee. The Senior Vice Presidents decision will not be arbitrary and capricious.
- 8. If the number of Registered Nurses on duty on a particular shift and staffing unit fall below the established minimum number of Registered Nurses, then ECMCC shall float employees, offer the open shift to all other qualified employees as overtime, offer extra regular time for an RPT employee, offer the shift to a Per Diem employee (if available on unit), or seek usage of agency nurses in accordance with Appendix M. If these measures fail in meeting the staffing needs of the units the parties agree to meet to discuss expansion of the applicable float pool for this purpose.

The Nurse Staffing Committee shall meet monthly to discuss vacancies, progress on filling positions, recruitment and retention strategies and initiatives, modifications to existing recruitment protocol, and best practice.

- 9. The recruitment of Registered Nurses, Nurse Practitioners, and Certified Registered Nurse Anesthetists will be a primary focus for the Human Resources Department. Recruiters will regularly attend recruitment events. Two bargaining unit members appointed by the Employer— (one (1) from the Hospital and one (1) from Long Term Care) will attend the recruitment events with the Recruiter. The bargaining unit member appointed, if they agree to the appointment, will be paid at their regular rate of pay plus differentials and/or overtime if applicable to attend the recruitment events.
- 10. Nursing Response Team: In the event a nurse files a protest of assignment alleging unsafe staffing with their front-line supervisor, the front-line supervisor shall immediately contact their direct supervisor. If that supervisor is not able to resolve the issue to the satisfaction of

the nurse, then the Nursing Response Team will be triggered by the nurse(s) that filed the protest of assignment. The Nursing Response Team shall be triggered by contacting the Nurse Staffing Office. The Nursing Response Team shall consist of the manager of the unit, that manager's direct supervisor and the nurse that triggered the Nursing Response Team. The protest of assignment and the response to the protest of assignment shall be reviewed at that month's Labor Management meeting. All efforts will be made to timely respond to staffing issues and convene the Nursing Response Team. A regular pattern of protest of assignments alleging unsafe staffing in a particular staffing unit will be referred to the Nurse Staffing Committee at their next scheduled quarterly meeting for discussion.

APPENDIX "T"

STAFFING GRIDS

Unit	Ratio: New York State Nurses Association (NYSNA)	Ratio: Civil Service Employees Association, Inc. (CSEA)	Ratio: American Federation of State, County and Municipal Employees (AFSCME)
Medical Intensive Care Unit (MICU) North	1:2/1:1 High Acuity/1:3 Transitioning Patients (All 3 Pts must be transitioning)	1:2 (LPNs)	1:11 (HA)
Medical Intensive Care Unit (MICU) South	1:2/1:1 High Acuity/1:3 Transitioning Patients (All 3 Pts must be transitioning)	1:2 (LPNs)	1:11 (HA)
Trauma Intensive Care Unit (TICU)	1:2/1:1 High Acuity/1:3 Transitioning Patients (All 3 Pts must be transitioning)	1:2 (LPNs)	1 (CCT)
Burn	1:2/1:1 High Acuity/1:3 Transitioning Patients	1:2 (LPNs)	1 (CCT)
Surgical Services*	1:1 - In cases where there is bedside sedation require additional ancillary or nursing staff	1:3 (LPNs)	
Post-Anesthesia Care Unit (PACU)	1:2 - If assigned at least one Phase 1 Pt. (Phase 1 is defined as immediate post-op to anesthesia sign-out), 1:3 - If assigned a combination of Phase 2 and/or Extended Care Pts. No Phase 1 pts will be included. (Phase 2 is defined as anesthesia sign-out to discharge), 1:4 - If assigned all Extended Care pts (Extended Care is defined as pts awaiting transportation longer than thirty (30) minutes to go home or pts who have had procedures requiring extended observation/intervention	1:3 LPNs	

A STATE OF THE STA	and patients being held for an in-		
	patient bed)		
Emergency Department (ED)	1:1 Trauma, 1:2 Critical Care, 1:4 General	1:4 (LPNs) and 1:4 (ED Techs)	1:15 (HA)
Comprehensive Psychiatric Emergency Program (CPEP)	1:3	1:5 (LPNs) 1:4 (CMHTs)	1:20 (HA)
9z4	1:5	1:6 (LPNs) and 1:4 (CMHTs)	
9 z 3	1;5	1:6 (LPNs) and 1:4 (CMHTs)	
9z2	1:4	1:4 (LPNs)	1:18 (HA)
9 z 1	1:5	1:6 (LPNs) and 1:4 (CMHTs)	1:15 (HA)
5z4	1:5	1:6 (LPNs) and 1:4 (CMHTs)	1:18 (HA)
5z3	1:5	1:6 (LPNs) and 1:4 (CMHTs)	1:18 (HA)
5z2	1:5	1:6 (LPNs) and 1:4 (CMHTs)	1:18 (HA)
5z1	1:4	1:6 (LPNs) and 1:4 (CMHTs)	1:18 (HA)
4z4	1:5	1:6 (LPNs) and 1:4 (CMHTs)	1:16 (HA)
4z3	1:3	1:6 (LPNs) and 1:4 (CMHTs)	1:10 (HA)
4z2	1:5	1:6 (LPNs) and 1:4 (CMHTs)	1:18 (HA)
4z1	1:5	1:6 (LPNs) and 1:4 (CMHTs)	1:18 (HA)
12z3	1:3	1:4 (LPNs)	
12z2	1:4/1:3 - If assigned one or more "step-down" and/or vented pts	1:4 (LPNs)	
10z4	1:4 - In the case that one side (10z3 or 10z4) has less than 4 pts, there will be no less than 1 GDN and 1 ancillary staff assigned that side	1:4 (LPNs)	1:11 (HA)
10z3	1:4 - In the case that one side (10z3 or 10z4) has less than 4 pts, there will be no less than 1 GDN and 1 ancillary staff assigned that side	1:4 (LPNs)	1:11 (HA)
ED Observation	1:4	1:4 (LPNs)	
8z4	1:5	1:4 (LPNs)	1:20 (HA)
8z3	1:4	1:4 (LPNs)	1:20 (HA)

8z2	1:4	1:4 (LPNs)	1:19 (HA)
8z1	1:4	1:4 (LPNs)	1:20 (HA)
7z4	1:4	1:4 (LPNs)	1:20 (HA)
7z3	1:5	1:4 (LPNs)	1:15 (HA)
7z2	1;3	1:4 (LPNs)	1:20 (HA)
7z1	1:5	1:4 (LPNs)	1:20 (HA)
6z4	1:4	1:4 (LPNs)	1:11 (HA)
6z3	1:4	1:4 (LPNs)	1:11 (HA)
6z2	1:5	1:4 (LPNs)	1:16 (HA)
6z1	1:4	1:4 (LPNs)	1:18 (HA)
Terrace View	37 Direct Care RNs		
Hemodialysis Inpatient	1:2	1:3 (LPNs)	
Hemodialysis Outpatient	1:12 - With appropriate ancillary staff/1:4 without proper ancillary staff	1:3 (LPNs)	

* Main, Snyder, Cath Lab/IR procedures, VAC procedures, GI procedures

Job Titles:	
CMHT - Community Mental Health Technician	
CCT - Critical Care Technician	
ED Tech - Emergency Room Technician	
GDN - General Duty Nurse	
HA - Hospital Aide	
LPN - Licensed Practical Nurse	

APPENDIX "U"

MEMORANDUM OF AGREEMENT

BETWEEN THE

NEW YORK NURSES ASSOCIATION

AND

ERIE COUNTY MEDICAL CENTER CORPORATION

Effective July 17, 2023

Nurse Apprenticeship and Community Based Hiring Program

- 1. The Employer, at their discretion, will make apprenticeship slots available to local high school senior students or individuals who have received their GED by age twenty one (21) for a 10-hour per week, 12-week program starting January 15, 2025 at the hospital. The program will repeat each year of the contract.
- 2. The Association and the Employer will sit on an Apprenticeship program committee, to include appropriate Employer nurse education and administration staff, as well as Association representatives designated by the Association.
- 3. The committee will discuss and establish a mentor program. The mentor/coaches will receive the preceptor differential for such work.
- 4. The Apprentice applicants must be a Senior/ GED recipient or recent graduate (within the last 12 months), or have received their GED within the last 12 months from a school in Erie County.
- 5. The Association and Employer will specifically search out a diverse group of Apprentice candidates. It is a stated goal of the program to seek Apprentices that reflect the races, ethnicities, genders, religious affiliations, identities, and sexual orientations of Erie County and the people of the City of Buffalo. As long as the number of qualified Apprentice candidates permits, at least sixty percent (60%) of participants shall live in the City of Buffalo.
- 6. If the Apprentice successfully completes this program and provides proof of graduation from their High School or proof of GED, the Employer will pay up to three thousand dollars (\$3,000) toward the cost of their tuition for a Bachelors in the Science of Nursing (BSN) program, less any other available grants, tuition assistance, or scholarships made available to the Apprentice.
- 7. If the Apprentice finishes the BSN program, they will be guaranteed a NYSNA RN position at the Employer and serve in such a position for at least 36 months following their being hired in that role.
- 8. Apprentices that do not complete the BSN program will have to present their case to the Association and Employer for review and possible repayment to the Employer for tuition and fees.
- 9. Exceptions to the completion of the BSN program and service requirements will be made by the Association and Employer mutually on a case-by-case basis.

Community Based Hiring

- A. Effective January 1, 2024, the Employer and NYSNA agree to jointly conduct at least three (3) local job fairs in the surrounding neighborhoods of the hospital. NYSNA will co-present and discuss the benefits of becoming a union nurse at the Employer, other programs available at the Employer. NYSNA will have input on the best locations, dates, and times for such job fairs.
- B. The Employer and NYSNA will conduct joint nursing school visits to recruit new graduate nurses and explain the benefits of becoming a union nurse at the Employer.
- C. The Employer and NYSNA will, search out engagement with schools, places of worship and not for profit organizations, search out diverse and inclusive nurse candidates. It is a stated goal of this Joint Community Based Hiring Program to seek nurse candidates that reflect the sex, race, creed, color, religion, military status/service, gender, national origin, age, disability, pregnancy status, sexual orientation, gender identity, political status, marital status, familial status, status as a victim of domestic violence, arrest or conviction record or predisposing genetic characteristics of the community.
- D. It is agreed and understood that all employment decisions are made by ECMCC without regard to unlawful considerations.

APPENDIX "V"

MEMORANDUM OF AGREEMENT

BETWEEN THE

NEW YORK NURSES ASSOCIATION

AND

ERIE COUNTY MEDICAL CENTER CORPORATION

Effective December 2, 2022

SUBSTANCE ABUSE/ DEPENDENCY LEAVE OF ABSENCE

Eligibility: Permanent employees who satisfactorily completed their probationary period.

Leave: An eligible employee may take up to twenty-six (26) weeks of unpaid leave to use for treatment for Substance Abuse Dependency Leave.

Use of Accruals: An employee may at their discretion utilize any available sick, vacation, PL, or compensatory time off (in that order) during this leave.

Health Insurance Continuation: The Corporation will continue to pay the Employer contribution toward the employee's health insurance during this leave of absence.

Return to Work Test: Prior to returning to duty, an employee taking such leave must take and pass a return-to-work alcohol and controlled substance screening.

Disciplinary Action: The Employer may have cause to discipline employees who have violated the Corporation's policies on Employee with Impaired Function and/or Drug diversion Program, Surveillance, Reporting and Response. Any employee in violation of the aforementioned policy will meet with the Substance Abuse/ Dependency Review Committee consisting of the Senior Vice President of Nursing, the Director of Labor and Employee Relations and either the Vice President of Ambulatory Services and Population Health or the Vice President of Pharmacy Operations and a determination will be made regarding disciplinary action and necessary employment restrictions.

If, at the time an employee requests Substance Abuse/ Dependency Leave, there are known Criminal charges related to on-the-job impairment or diversion of narcotics, the employee will no longer be eligible for the Leave.

APPENDIX "W"

MEMORANDUM OF AGREEMENT

BETWEEN THE

NEW YORK NURSES ASSOCIATION

AND

ERIE COUNTY MEDICAL CENTER CORPORATION

Effective January 1, 2023

REFERRAL BONUS

Clarify the parties February 24, 2022 Impact Negotiations Agreement regarding the Referral Bonus as follows:

- 1. The Employer agrees to pay a referral bonus of up to \$1,800.00 to any bargaining unit employee who refers a successful applicant for a position in the bargaining unit represented by the Association provided that the successful applicant satisfies the requirements set forth in this Memorandum of Agreement.
- 2. The applicant for a position in the bargaining unit represented by the Association must indicate on the Employment Application that she/he was referred by a bargaining unit employee and identify the referring employee on the Employment Application.
- 3. The referring employee for a bargaining unit position must complete an Attestation on a form provided by the Employer confirming that the referring employee did refer the successful applicant to the Employer. This Attestation will be timestamped by Human Resources and a copy shall be provided to the referring employee.
- 4. The successful applicant must be hired by the Employer and must complete the Employer's nursing orientation. In the event that the successful applicant does satisfactorily complete the Employer's nursing orientation, the Employer will pay the referring employee the first installment of the referral bonus (\$1,000.00) in the first full pay period following completion of the nursing orientation.
- 5. The successful applicant must complete one full year of actual employment for the Employer. If the successful applicant is absent from work for any reason for a period in excess of seven consecutive days, the one-year period will be extended by the length of that leave. In the event that the successful applicant does satisfactorily complete one full year of actual employment for the Employer, the Employer will pay the referring employee the second installment of the referral bonus (\$800.00) in the first full pay period following completion of the successful applicant's first year of actual employment for the Employer.

MEMORANDUM OF AGREEMENT

BETWEEN THE

NEW YORK NURSES ASSOCIATION

AND

ERIE COUNTY MEDICAL CENTER CORPORATION

Effective January 1, 2023

PANDEMIC PREPARATION AND SAFETY

To ensure the safety and health of employees and patients, the Employer agrees to the following preparedness and safety measures for current and future pandemics. Pandemic preparedness and safety measures will be subject to the evolving regulatory requirements of all levels of government. The Employer will consult with health and safety and labor management committees on a regular basis during the pandemic, to obtain real-time information on unfolding issues, and will take into consideration recommendations made by the committees. In the event that a provision of this Agreement conflicts with state or federal laws and/or regulations, the law or regulation shall take precedence over the contract in any case where the state or federal law provides a greater benefit or protection to employees, patients, and/or visitors.

Pandemic conditions will be invoked when the Centers for Disease Control (CDC), New York State Department of Health (NYSDOH), and/or the World Health Organization (WHO) declare such conditions. However, all preparatory work outlined in this section will be done regardless of Pandemic conditions.

All provisions of this Section will be permanent and ongoing obligations, unless noted as while "under Epidemic/Pandemic conditions, or unless it conflicts with state or federal law and/or regulation. If such state or federal law provides a greater benefit or protection to employees, patients, and/ visitors, the state or federal law or regulation shall supersede this Agreement.

X.01- Personal Protective Equipment (PPE)

- A. A risk assessment will be conducted for emerging infections and a 60-day supply of unexpired PPE will be maintained. Regarding disposable PPE, e.g. N95 respirators, the supply will be determined using burn rates for conventional use of the equipment. An adequate supply of approved disposable PPE will be kept in stock and will be used for patient care and then discarded.
- B. Where respiratory protection is required, wearers shall be fit tested following OSHA requirements on the make, model and size of respirator which they have been given to wear. The Employer will implement and maintain policies and procedures to ensure that staff who experience difficulty getting a fit are able to obtain equipment that fits properly as outlined in the Respiratory Protection Program.
- C. Under Epidemic/Pandemic conditions, the Employer will provide inventory reports at Labor Management Meetings, including purchase and expiration dates, and information on the location and conditions of storage in the case that there are any changes.

- D. The Employer will follow CDC and NYSDOH guidelines for PPE utilization. Staff will not be required to reuse gloves and gowns. While in care of COVID positive patients or PUIs, staff will not be required to reuse N95s unless extended use practices are in place due to supply chain issues. Disposable N95s will be available. Reusable PPE will be cleaned and decontaminated.
- E. The Employer will follow the current Respiratory Protection Program and will purchase and issue elastomeric half cartridge respirators for use by all bedside staff who request them. The employee issued an elastomeric half cartridge respirator must clean and maintain that equipment. The Employer will purchase and maintain in proper working order a selection of powered air purifying respirators, including replacement hoods and hoses, for emergency use.

X.02- Environmental Controls

- A. All confirmed patients and patients under investigation (PUI) will be cohorted, if possible, based on census.
- B. All access controls and visitation will be reviewed and revised based on NYS DOH guidelines.
- C. All patients must be tested prior to admission in accordance with NYS DOH guidelines unless patient rights or the law allow for refusal.
- D. During pandemic conditions, masking for visitors will follow NYS DOH guidelines at the time of visitation.
- E. The Employer will comply with any restrictions by NYS DOH regarding elective procedures.
- F. The Employer will assess, in consultation with frontline staff and LM committees, the need for temporary and/or permanent enhancements to ventilation and air filtration to address the airborne transmission of pathogens. The Employer will be prepared to deploy in patient care areas, as well as other key areas, such as break rooms, a selection of enhancements, including: portable HEPA filtration units, increased percentage of outside air, the use of source devices such as ventilated headboards and override demand controlled ventilation in all areas. Any enhancements must comply with the NYS DOH guidelines and the Transmission Based Precautions policy.
- G. Units and areas designated for the care of confirmed and/or PUI patients will be provided with designated PPE donning and doffing areas. The Employer will follow the Transmission Based Precautions policy.
- H. The Labor Management Committee will review unit assessments of the entire facility to assess Pandemic preparedness and needs. The assessments will be conducted by the Environment of Care Committee and will include at least one (1) bargaining unit member appointed by the Association.

X.03 Employee Communication and Testing

- A. The Employer will establish a Command Center and hotline for all issues employees may have while under Pandemic conditions.
- B. The Employer will issue daily reports to the employees during Pandemic conditions. The report will include patient census numbers, PPE levels and availability, along with Command Center information, testing availability, and staff illness information.

- C. The Employer will make available universal diagnostic testing sites for employees while under Pandemic conditions. The testing will be free of charge to the employee and available during all shifts.
 - D. The Employer will implement contact tracing protocols while under Pandemic conditions.

X.04 Pandemic Leave Rights

A. The Employer will follow current time and attendance policies and NYS guidelines and will immediately take staff off the schedule who are under investigation for having contracted the pathogen, or tested positive for it, for the period of time they will be infectious.

X.05 Staffing and Floating Rights During Pandemic

Employees will not be floated to units and given a patient assignment unless they are clinically competent to independently perform the standard patient care assignment for that destination unit. The Employer will follow current staffing policies.

MEMORANDUM OF AGREEMENT

BETWEEN THE

NEW YORK NURSES ASSOCIATION

AND

ERIE COUNTY MEDICAL CENTER CORPORATION

Effective January 1, 2023

Workplace Violence Prevention

- 1. The employer will install, implement, and maintain weapons screening equipment, technology and protocols in areas including, but not limited to the main entrance, the visitor emergency department entrance as well as the ambulance emergency department entrance, the Erie County Medical Center (Hospital") main entrance, as well as the Terrace View main entrance to prevent weapons from entering the facility thereby reducing the risk of serious injury and/or catastrophic incidents.
- It is acknowledged that the Corporation has issued a Request for Proposals to vendors and has made an award to a prospective vendor in accordance with its procurement guidelines, for weapons screening equipment for the visitor emergency department entrance, the ambulance emergency department entrance, the Hospital main entrance, as well as the Terrace View main entrance.
- 3. The Corporation will make every effort to have the weapons screening technology operational no later than August 1, 2023; however, it is understood that this deadline is subject to the capability of the selected vendor to procure and install the technology by that date as well as the ability to hire and train a sufficient number of Building Guards to staff this equipment. In the event that the vendor is unable to meet this deadline for installation or the Corporation is unable to hire and train sufficient Building Guards by that date, the Corporation will notify the Association of the delay and the new anticipated installation dates. Thereafter, the Corporation will provide updates on the status of the installation on a bi-weekly basis.
- 4. The Corporation will make every effort to have the weapons screening technology installed in the Hospital main entrance and the Terrace View main entrance no later than November 1, 2023; however, it is understood that this deadline is subject to the capability of the selected vendor to procure and install the technology by that date as well as the ability to hire and train a sufficient number of Building Guards to staff this equipment. In the event that the vendor is unable to meet this deadline for installation, the Corporation will notify the Association of the delay and the new anticipated installation dates. Thereafter, the Corporation will provide updates on the status of the installation on a bi-weekly basis.
- 5. The Corporation agrees to include one (1) NYSNA Representative on its Workplace Violence Committee, which shall meet quarterly, and the topic of Threat Assessment will be on the agenda for each meeting as well as a standing agenda item in labor management meetings. This Committee will conduct a hazard assessment annually, during which time the members of the hazard assessment subcommittee shall conduct an on-site physical inspection of ECMCC's buildings, grounds, and other worksites to identify the presence of factors that may place employees at risk of workplace violence. After completion of the on-site assessment, the Committee shall convene to discuss recommendations arising from the on-site assessment regarding engineering controls, administrative or work practice controls, and personal protective equipment. The Workplace Violence Committee will evaluate any units with a higher potential for patient agitation and violence, including, but not limited to, staff-only work areas, including nurse station areas that should be secured to insure that patients or visitors cannot enter, cannot reach into the area or throw objects into the area, and make recommendations to ECMCC's Capital Committee.

- 6. Management will continue to deploy Public Safety Assistant, Hospital Police who are trained in violence prevention, de-escalation and intervention, in emergency departments, medical surgical units, behavioral health and detox units, as a means of reducing the risk of patient and visitor agitation, threats and violence.
- 7. In the emergency department and CPEP, the employer will make every effort to limit patients being held awaiting placement or discharge. In cases where this cannot be achieved temporarily, the employer will redeploy available clinical staff and security personnel patrolling the emergency departments.
- 8. The employer will deploy and maintain emergency/panic alarm systems including, but not limited to, one-touch badge panic alarms that can identify a staff member's location immediately upon activating. The Corporation agrees to solicit input and consult with the Association as it considers investments in new personal alarm systems.